UNIVERSITY BIRMINGHAM University of Birmingham Research at Birmingham

Digital Behaviour Change Interventions to Break and Form Habits

Pinder, Charlie; Vermeulen, Jo; Cowan, Benjamin ; Beale, Russell

DOI: 10.1145/3196830

License: None: All rights reserved

Document Version Peer reviewed version

Citation for published version (Harvard):

Pinder, C, Vermeulen, J, Cowan, B & Beale, R 2018, 'Digital Behaviour Change Interventions to Break and Form Habits', ACM Transactions on Computer-Human Interaction, vol. 25, no. 3, 15. https://doi.org/10.1145/3196830

Link to publication on Research at Birmingham portal

Publisher Rights Statement:

© ACM, 2018. This is the author's version of the work. It is posted here by permission of ACM for your personal use. Not for redistribution. The definitive version was published in ACM Transactions on Computer-Human Interaction, {VOL25, ISS3, June 2018} http://doi.acm.org/10.1145/3196830

General rights

Unless a licence is specified above, all rights (including copyright and moral rights) in this document are retained by the authors and/or the copyright holders. The express permission of the copyright holder must be obtained for any use of this material other than for purposes permitted by law.

•Users may freely distribute the URL that is used to identify this publication.

•Users may download and/or print one copy of the publication from the University of Birmingham research portal for the purpose of private study or non-commercial research.

•User may use extracts from the document in line with the concept of 'fair dealing' under the Copyright, Designs and Patents Act 1988 (?) •Users may not further distribute the material nor use it for the purposes of commercial gain.

Where a licence is displayed above, please note the terms and conditions of the licence govern your use of this document.

When citing, please reference the published version.

Take down policy

While the University of Birmingham exercises care and attention in making items available there are rare occasions when an item has been uploaded in error or has been deemed to be commercially or otherwise sensitive.

If you believe that this is the case for this document, please contact UBIRA@lists.bham.ac.uk providing details and we will remove access to the work immediately and investigate.

UNIVERSITY^{OF} BIRMINGHAM

Research at Birmingham

Digital Behaviour Change Interventions to Break and Form Habits

Pinder, Charlie; Vermeulen, Jo; Cowan, Benjamin ; Beale, Russell

DOI: 10.1145/3196830

Document Version Peer reviewed version

Citation for published version (Harvard):

Pinder, C, Vermeulen, J, Cowan, B & Beale, R 2018, 'Digital Behaviour Change Interventions to Break and Form Habits' ACM Transactions on Computer-Human Interaction, vol 25, no. 3, 15. DOI: 10.1145/3196830

Link to publication on Research at Birmingham portal

General rights

Unless a licence is specified above, all rights (including copyright and moral rights) in this document are retained by the authors and/or the copyright holders. The express permission of the copyright holder must be obtained for any use of this material other than for purposes permitted by law.

• Users may freely distribute the URL that is used to identify this publication.

• Users may download and/or print one copy of the publication from the University of Birmingham research portal for the purpose of private study or non-commercial research.

User may use extracts from the document in line with the concept of 'fair dealing' under the Copyright, Designs and Patents Act 1988 (?)
Users may not further distribute the material nor use it for the purposes of commercial gain.

Where a licence is displayed above, please note the terms and conditions of the licence govern your use of this document.

When citing, please reference the published version.

Take down policy

While the University of Birmingham exercises care and attention in making items available there are rare occasions when an item has been uploaded in error or has been deemed to be commercially or otherwise sensitive.

If you believe that this is the case for this document, please contact UBIRA@lists.bham.ac.uk providing details and we will remove access to the work immediately and investigate.

CHARLIE PINDER, University of Birmingham, UK JO VERMEULEN, Aarhus University, Denmark BENJAMIN R COWAN, University College Dublin, Ireland RUSSELL BEALE, University of Birmingham, UK

Digital behaviour change interventions, particularly those using pervasive computing technology, hold great promise in supporting users to change their behaviour. However, most interventions fail to take habitual behaviour into account, limiting their potential impact. This failure is partly driven by a plethora of overlapping behaviour change theories and related strategies that do not consider the role of habits. We critically review the main theories and models used in the research to analyse their application to designing effective habitual behaviour change interventions. We highlight the potential for Dual Process Theory, modern habit theory and Goal Setting Theory, which together model how users form and break habits, to drive effective digital interventions. We synthesise these theories into an explanatory framework, the Habit Alteration Model, and use it to outline the state of the art. We identify the opportunities and challenges of habit-focused interventions.

CCS Concepts: • Human-centered computing \rightarrow HCI theory, concepts and models; Ubiquitous and mobile computing design and evaluation methods; • General and reference \rightarrow Surveys and overviews;

Additional Key Words and Phrases: Digital behaviour change interventions, behaviour change technology, persuasive technology, habit breaking technology, habit forming technology

ACM Reference Format:

Charlie Pinder, Jo Vermeulen, Benjamin R Cowan, and Russell Beale. 2018. Digital Behaviour Change Interventions to Break and Form Habits. 1, 1 (June 2018), 68 pages. https://doi.org/10.1145/3196830

1 INTRODUCTION

There is clear potential for digital behaviour change interventions (DBCIs), particularly those using pervasive technology, to change behaviour. Humans tend to make poor health decisions [Keeney 2008], while pervasive computing technology offering multiple detection and intervention points becomes cheaper and more widely owned.

The World Health Organisation (WHO) 2017 has identified four key lifestyle behaviours that impact severely on health: "tobacco use, unhealthy diet, lack of physical activity, and the harmful use of alcohol". Meanwhile, technology becomes more ubiquitous. Smartphone ownership reached 80% in the UK in 2017 [Ipsos 2017], and there is rapid growth in wearables [Drake et al. 2017; Lee et al. 2015]. Smartphone health interventions are prevalent [Fiordelli et al. 2013; Klasnja and Pratt 2012; Lathia et al. 2013], and more ubicomp behaviour change interventions are emerging e.g. [Adams et al. 2015; Khot et al. 2015; Kim et al. 2016].

An obvious question is how best to design DBCIs to generate long-lasting results. The answer is not clear. This is primarily because the DBCI research area is cluttered with a large number of

Authors' addresses: Charlie Pinder, University of Birmingham, Edgbaston, Birmingham, B15 2TT, UK, charlie.pinder@gmail.com; Jo Vermeulen, Aarhus University, Åbogade 34, Aarhus N, 8200, Denmark, jo.vermeulen@cs.au.dk; Benjamin R Cowan, University College Dublin, Stillorgan Rd 4, Dublin 4, Belfield, Ireland, benjamin.cowan@ucd.ie; Russell Beale, University of Birmingham, Edgbaston, Birmingham, B15 2TT, UK, R.Beale@cs.bham.ac.uk.

© 2018 Copyright held by the owner/author(s).

This is the author's version of the work. It is posted here for your personal use. Not for redistribution. The definitive Version of Record was published in https://doi.org/10.1145/3196830

different theories, approaches and techniques, with research often failing to either evaluate longterm effectiveness or successfully identify long-term impact. Several meta-reviews have identified a general poor use of theory in DBCI research [Michie and Prestwich 2010; Wiafe and Nakata 2012], in DBCI apps [Cowan et al. 2012], and in behaviour change research in general [Michie and Johnston 2012; Sladek et al. 2006]. There are many theories and other explanatory structures, and they are generally not well applied. Crucially, several theories do not take the habitual nature of behaviours into account in their interventions, as we discuss in detail in Section 2.

We propose that a focus on changing habitual behaviours will improve the long-term efficacy of DBCIs. Our purpose is to outline the appropriate theoretical and strategic approaches to alter habitual and other automatic behaviours using technological interventions. We particularly focus on DBCIs using pervasive devices including smartphones, fitness trackers and smart home devices.

1.1 Article overview

In the first part of this article, Section 1, we establish why the study of habits is crucial for behaviour change and outline relevant domains for intervention. In Section 2, we give an overview of the main theoretical approaches in DBCI research to determine their suitability to target habits. We highlight three theories that together explain intervention points for habitual behaviours: Dual Process Theory, modern habit theory and Goal Setting Theory. In Section 3, we synthesise these theories into a framework (the Habit Alteration Model, HAM) to illustrate the main components and the potential intervention points for habit-focused DBCIs. We explore the related intervention strategies in Section 4, identifying the state of the art in the research field. We complement the strategies with a set of design principles in Section 5.

Section 6 outlines the main challenges in developing effective DBCIs to change habitual behaviour. In Section 7, we outline a future research agenda for the intervention opportunities afforded by increasingly pervasive context-aware technology. We present our conclusions in Section 8.

1.2 The importance of habits

1.2.1 What is a habit? Habits are learned impulses to perform a particular behaviour, triggered outside of conscious awareness by a particular context. Habitual behaviour is learned behaviour that is "frequently repeated, has acquired a high degree of automaticity, and is cued in stable contexts" [Orbell and Verplanken 2010]. Automaticity means habitual behaviours can be performed nonconsciously, i.e. "enacted with little conscious awareness" [Orbell and Verplanken 2010]. Note that in contrast to common usage of the word habit, we do not define it as the behaviour itself. Instead, following Gardner 2015, we define a habit as a link represented in associative memory between a certain context and a specific response. The occurrence of the context triggers a response impulse via context-response links. A habit is therefore a disposition to perform a given behaviour [Gardner 2015; Neal et al. 2006]. Habitual behaviour is the behaviour that results from this impulse.

Although habitual behaviours are triggered nonconsciously, people are not necessarily unaware of their actual behaviour. Instead, they tend to be unaware of the habit's internal mechanisms [Stanovich 2005], such as the context-response associations [Wood and Rünger 2016]. This inability to introspect a habit's underlying links makes them difficult to change: if the cause of an unwanted behaviour is not clear, then neither is the solution.

1.2.2 Habit prevalence and domains. Habits are highly prevalent and structure much of everyday life [Wood et al. 2014]. People report 43% of their behaviours as being performed without conscious thought ([Wood et al. 2002], study 2). Habitual behaviours span multiple domains: health [Gardner 2015], including eating [Robinson et al. 2013; Rothman et al. 2009; Wansink 2010], exercise behaviour [Aarts et al. 1997; Conroy et al. 2013] and physical activity [Rebar et al. 2016]; behaviour of

Year	Behaviour change	Mention habit	Substantive discussion	Behaviour domains
CHI '15	36 (7%)	10 (2%)	4 [Hollis et al. 2015; Paay et al. 2015; Stawarz et al. 2015; Sugarman and Lank 2015]	Smoking; smartphone behaviour; unwanted behaviours; electricity consumption
CHI '15	34 (6%)	17 (3%)	3 [Banovic et al. 2016; Hasselqvist et al. 2016; Sonne et al. 2016]	Childcare; sustainable transport; general routine behaviour
CHI '17	37 (6%)	20 (3%)	2 [Banovic et al. 2016; Jansen et al. 2017]	General routine behaviour; eating behaviour

Table 1. Recent CHI habit mentions in full behaviour change papers.

healthcare workers [Nilsen et al. 2012]; and environmental behaviours [Klöckner 2013]. Habits are also important in our use of technology [Bayer and Campbell 2012; Limayem et al. 2007], including participation in online communities [Wohn et al. 2012] and use of smartphones [Oulasvirta et al. 2012; van Deursen et al. 2015].

Despite a rise in interest in the habit construct in health psychology [Gardner 2015], few general behaviour change interventions currently use theory related to habit formation [Lally et al. 2008]. Likewise, few DBCI apps target habits [Stawarz et al. 2015]. Orji and Moffatt 2016 found that of 85 health domain DBCIs, only 3 targeted habits. Further, a review of 176 DBCI papers found only 11 that targeted nonconscious behaviour, and only 2 that mentioned related theory [Adams et al. 2015].

As a sample of recent DBCIs, we searched Google Scholar¹ for behaviour change CHI papers from the past 3 years. The results are shown in Table 1. They demonstrate that few recent CHI DBCIs have targeted habits. For instance, out of 20 CHI '17 behaviour change papers that mentioned habit, only 2 address this directly as a construct in the research, with the vast majority only using the term colloquially.

DBCIs do, however, tend to focus on domains where habitual behaviours can emerge. Table 1 and Table 3 show that recent DBCIs have focused on health, technology and environmental behaviours alongside general routine behaviours. More broadly, Hamari et al.'s review of 95 DBCIs 2014 found that the main target domains were health (47%) and environmental behaviours (21%).

1.2.3 A challenge and an opportunity. Changing behaviour via habits represents both a challenge and an opportunity. The challenge is to break unwanted nonconscious habitual behaviours that are resistant to change. The opportunity is to use the habit mechanism to establish wanted nonconscious habitual behaviours that are similarly resistant to change. Habit formation can enable the maintenance of wanted behaviours [Sheeran et al. 2017b], since habitual behaviours are the default behaviour when people are unable or unwilling to make effortful decisions about how to behave [Neal et al. 2013]. They are performed automatically with little cognitive effort. These properties mean that DBCIs that can successfully form 'good' habits and break 'bad' habits are likely to have long-lasting behavioural effects [Lally et al. 2011; Rothman et al. 2009; Sheeran et al.

¹Search completed 13th June 2017. Search terms "habit", ("behaviour change""behavior change"and/or "persuasive technology")

2017b; Verplanken and Wood 2006]. However, people often return to their unwanted behaviours over time [Bouton 2014]. This failure to sustain behaviour change is due to a lack of focus on automatic processes [Marteau et al. 2012] including habits. Behaviour change research tends to use deliberative interventions that rely on conscious resources: the provision of information is the most common DBCI technique [Webb et al. 2010b]. However, such interventions tend to be unsuccessful in the long term: Hillsdon et al. 2002 found evidence that over longer time periods, simply advising people to take more exercise is ineffective. Verplanken and Wood 2006 attribute such failures in part to the environment's automatic activation of habitual behaviour to the detriment of people's intentions. Reflecting this, habits are one of the key challenges for behavioural change policy [Jackson 2005]. There is therefore a clear research gap in considering habit as a key driver of long-term behaviour change. This article addresses this gap.

2 THEORIES OF BEHAVIOUR CHANGE

A critical component of addressing this gap is by understanding the key behavioural theories that contribute to our understanding of behaviour change at present and how they apply to habits. This section reviews the use of theory in behaviour change in general and in DBCIs.

2.1 Theory use in behaviour change research

The behaviour change research landscape is cluttered with many different theories, frameworks, models, techniques, strategies and patterns. Table 2 demonstrates the scale of the issue: just a few behaviour change researchers have identified tens of different behaviour change techniques, multiple ways behaviour might change, and numerous related theories and models.

Theory enables researchers to be more explicit about their assumptions, strategies and intervention targets [Rimer and Glanz 2005]. Despite—or perhaps because of—the number of competing models, there is a persistent lack of reference to theory in behaviour change research. The problem extends to DBCI research. Multiple reviews of DBCI research have found less than 50% specified a theoretical basis [Orji and Moffatt 2016; Wiafe and Nakata 2012]. Many interventions that claim to be based on theory fail to make explicit how the theory relates to the intervention or use theoretically predicted measures as evaluating criteria [Harris et al. 2011; Michie and Prestwich 2010]. Likewise, few interventions provide an in-depth understanding of the underlying mechanisms of behaviour and attitude change [Riley et al. 2011; Segerståhl et al. 2010], and few persuasive systems justify in detail their choice of behaviour change strategy, or the impact they are expected to have on their users [Foster et al. 2011].

This "theoretical gap" [Hekler et al. 2013] makes knowledge transfer between interventions difficult because it is not clear how and why a given intervention succeeds or fails [Nilsen 2015]. Under-use of theory is likely to impact the efficacy of the intervention, because important design characteristics are overlooked [Moller et al. 2017]. There is some evidence that interventions with a strong theoretical basis have a stronger association with efficacy [Taylor et al. 2012; Webb et al. 2010b], although this point is the subject of some debate [Michie and Prestwich 2010]. Ultimately, the paradigm shift to delivering behaviour change applications using technology is a key opportunity to deliver interventions based on systematic application of theory [Moller et al. 2017]. The gap in theory use reflects a lack of clarity around how to apply commonly-used theories to DBCIs. Health behaviour theories in particular have been criticised for being "woefully underspecified" [Sheeran et al. 2017b]. The inability of one single theory to address all aspects of behaviour change means researchers tend to use a 'pick-and-mix' approach to basing behaviour change strategies on theory [Bandura 1998; Hekler et al. 2013; Honka et al. 2011]. For example, the myBehavior system [Rabbi et al. 2015] incorporates elements from the Fogg Behaviour Model [Fogg 2009b], two decision theory models and Social Cognitive Theory, while Consolvo et al.'s set of design strategies 2009b

Number of ways behaviour might change identified in the literature	35 [Fogg 2009a] 26 [Abraham and Michie 2008]
Number of behaviour change techniques, strategies or patterns identified in the literature	 101 [Lockton et al. 2010] 93 [Michie et al. 2013a] 56 [Wiafe and Nakata 2012] 50 [Lockton et al. 2009] 15 [Hamari et al. 2014] 7 [Fogg 2002]
Number of theories, frameworks or models identified in the literature	83 [Michie et al. 2014b] 15 [Wiafe and Nakata 2012]

Table 2. Behaviour change patterns identified in the literature.

incorporates strands from Cognitive Dissonance Theory [Festinger 1957] and the Transtheoretical Model [Prochaska and Velicer 1997], amongst others.

In this pick-and-mix context, theory overlaps result in disagreement about which individual behaviour change strategy construct belongs to which theory [Doshi et al. 2003]. This is a particular problem for DBCI designers wishing to target habits because (a) it is not clear how the most commonly-used theories relate to habitual behaviour and (b) there is no theoretical consensus on habit mechanisms [Neal et al. 2006]. This article aims to provide clarity on the ability of the most commonly-used theories to explain habitual behaviours, and to bring together the most pertinent theories and strategies into an explanatory model that can inform the design of DBCIs to target habits.

2.2 Theory selection

We selected ten prominent theories in the literature and analysed their application to habit change: These theories either directly address habits (Behaviourism; Theory of Interpersonal Behaviour; Dual Process Theory; modern habit theory), are commonly used in behaviour change and DBCI research (Theory of Planned Behaviour; Social Cognitive Theory; Transtheoretical Model; Goal Setting Theory), directly addresses technology-mediated behaviour change (Fogg Behavior Model), or are very comprehensive in their coverage (COM-B). Table 3 shows a summary of the theories and an overview of their current use, the number of mentions in the ACM Digital Library within ACM Journals and Proceedings² (ACM search column), and citations for the key papers relating to each specific theory (Google Scholar and Scopus citations columns).

²Search terms: theory/model title and ("behaviour change" or "behavior change" or "persuasive technology")

Domains	Eco-friendly behaviours; exercise	Exercise; recycling; general behaviour change	Use of technology	Physical activity & diet; snacking
Recent Implementations	[Adams et al. 2009; Cowan et al. 2013; Foster et al. 2011; Kirman et al. 2010]	[Bexheti et al. 2015; Comber and Thieme 2013; Schneider et al. 2016; Suh and Hsieh 2016]	[Gimpel et al. 2016; Moody and Siponen 2013]	[Khan et al. 2012; Rabbi et al. 2015]
Key determinants of behaviour	External environment	Intention and Perceived Behavioural Control	Past behaviour, intention and situational conditions.	Expected behavioural outcomes, environment and personal factors including self-efficacy.
Summary	Behaviour is learned from interacting with the environment. This interaction forms stimulus-response associations.	Behaviour is rational, determined by conscious intentions and Perceived Behavioural Control (an internal assessment of their ability to perform the behaviour).	Extends reasoned-action theories by incorporating repeated past behaviour.	Behaviour is determined by an interaction between existing behaviours, the environment (including social factors), and personal cognitive, affective and biological influences.
Citation source	[Skinner 1938]	[Ajzen 1991]	[Triandis 1977]	[Bandura 2001] [Bandura 1986]
Scopus	N/A	19954	N/A	2882 N/A
Google Scholar	*10981	50089	2085	11673 264
ACM	45 ³	56	0	57
Theory/ Model	Operant conditioning: behaviourism	Theory of Planned Behaviour	Theory of Interpersonal Behaviour	Social Cognitive Theory

Pinder, C. et al

Table 3. Search results of theory/model mentions in the ACM plus citations and recent implementations

 $^{^3 34}$ Operant conditioning results + 11 behaviourism results $^4 \, \rm search \, run \, 07/06/2017$

Table 3. Search results of theory/model mentions in the ACM plus citations and recent implementations

Domains	Smoking; smartphone use	Sleep; physical activity	Sleep; physical activity & diet; work breaks; energy conservation
Recent Implementations	[Paay et al. 2015; Park and Gweon 2015; Southern et al. 2017; Wittekind et al. 2015]	[Howlett et al. 2017; Lee et al. 2017; Walsh et al. 2016]	[Cambo et al. 2017; Lee et al. 2017; Rabbi et al. 2015; Sugarman and Lank 2015]
Key determinants of behaviour	Individual stage of change; self-efficacy; decisional balance.	Capability, opportunity and motivation.	Motivation, ability and a trigger
Summary	Six stages of behaviour, with ten processes for change. Interventions can move people between stages by targeting self-efficacy and perception of advantages and disadvantages of behaviour.	An interacting system of capability, opportunity and motivation.	Behaviour is executed when 3 elements co-occur: motivation, ability and a trigger.
Citation source	[Prochaska and Velicer 1997]	[Michie et al. 2011]	[Fogg 2009b]
Scopus	2487	612	214
Google Scholar	4942	1123	948
ACM	150	02	86
Theory/ Model	Transtheoretical Model	COM-B model	Fogg Behavior Model

⁵search run 07/06/2017 ⁶plus 81 direct citations within ACM

recent implementations
plus citations and
he ACM
entions in t
y/model m
SOL
results of theor
3. Search results of theor

Domains	Eating: physical activity;	privacy	Stress: physical activity; learning
Recent Implementations	[Adams et al. 2015; Phelan et al. 2016; Pinder et al. 2015;	Wang et al. 2014]	[Ding et al. 2016; Gouveia et al. 2015; Konrad et al. 2015; Lomas et al. 2017]
Key determinants of behaviour	Interaction of two sets of cognitive processes: Type 1 processes (fast, automatic, nonconscious, associative); and Type 2 processes	(slower, deliberative, conscious).	Intentions, contextual constraints
Summary	Behaviour is determined by two distinct sets of cognitive processes: the Type 1 automatic system, formed of associative links; and	the Type 2 conscious, deliberative system.	Behaviour occurs where intentions are specified with an appropriate level of difficulty and specificity, and are accepted by users.
Citation source	[Strack and Deutsch 2004]	[Kahneman 2011]	[Locke and Latham 2002]
Scopus	1426	N/A	2073
Google Scholar	*2960	13071	5258
ACM	67		43
Theory/ Model	Dual Process Theory		Goal-setting theory

Except where noted, ACM searches run 28/04/16. Google Scholar and Scopus searches run fune 2017 *: theories that do not have one single article that defines them

 $^{^7\}mathrm{half}$ of these papers were co-authored by the first author of the present paper

Table 3. Search results of theory/model mentions in the ACM plus citations and recent implementations

Domains	
Recent Implementations	
Key determinants of behaviour	
Summary	
Citation source	
Scopus	
Google Scholar	
ACM	
Theory/ Model	N/A: items missing from Scopus

For theories with multiple sources (Behaviourism, Dual Process Theory, Social Cognitive Theory), we selected one or two relevant sources as a citation reference point. Since mere mentions and citations do not necessarily reflect implementations, we have augmented the results with recent applications of each theory from DBCI research (Recent Implementations column) and the domain of application (Domains column). Note that modern habit theory Section 2.7.2 is not listed in Table 3 because there is no one key paper that defines it and therefore it does not appear as a standalone search result.

This article does not aim to provide a comprehensive summary of all possible behaviour change theories and models that are available. Our aim is to consider the utility of applying the selected theories to changing habits and DBCIs, and to highlight recent research using them.

A key historical split in behaviour change theories and models is between behaviourism, which prioritise the role of the external environment in triggering behaviour, and cognitive theories, which argue that behaviour is also explained by abstract cognitive constructs such as thoughts and motivations. More recently, integrated theories and models have emerged to reconcile both standpoints since neither theory can account for all the complexities of behaviour change [Bandura 1998; Prochaska and Velicer 1997]. We consider our selected ten theories and models in each of these three categories below before addressing their overlaps and omissions.

2.3 Behaviourism

Behaviourism is a key theory in understanding habits because it focuses on the effects of the external environment on behaviour. It explicitly rejects the use of cognitive constructs to explain behaviour because they cannot be rigorously observed: only directly observable actions are considered.

2.3.1 Overview. Behaviourists see habits as stimulus-response pairs formed outside conscious decision-making [West 2006] via two mechanisms of associative learning: classical and operant conditioning. A stimulus becomes associated with a particular response via repetition. Classical conditioning is the simple pairing of stimuli with responses; operant conditioning is the pairing of a stimulus-response with a positive or negative outcome. i.e. a reward for a wanted response, and a punishment for an unwanted one. Thus, rewarding a behaviour increases stimulus-response links and makes it more likely to be repeated. With repetition over time, any contextual cues that co-occur with a behaviour can trigger it [Davis 2001]. A behaviour is considered habitual when removing the reward does not diminish the behaviour, i.e. it is resistant to extinction. For example, a smoker who initially felt a positive reward from smoking (operant conditioning) may be prompted to smoke by the sight of a cigarette packet (classical conditioning), regardless of the subsequent reward. A key determinant of the impact of operant conditioning is how the reinforcement is delivered, or the reinforcement schedule [Staddon and Cerutti 2003]. A variable reinforcement schedule (where a reward is delivered to an average time or response rate, but not always at a given time or response) is the most effective in producing behaviour that is stable and resistant to extinction, rather than using constant or random reinforcements [Bijou 1957].

2.3.2 Recent implementations and empirical evidence. Erev and Gopher 1999 argue that research suggests that a reinforcement learning model, where the probability of a certain behaviour being performed increases when it is positively reinforced, provides "an extremely good approximation of behavior in a wide set of situations" over alternative models that assume rationality. However, much of the research applying the principles of behaviourism to DBCIs is of a speculative nature. For example, Adams et al. 2009 suggest that pervasive exercise games are a good test-bed for empirical exploration of behaviourist learning principles but did not test this hypothesis.

Some researchers suggest that the use of operant conditioning and variable rewards in the technology domain underpin the use of social networks [Fogg 2009a], and the problematic use of

such social networks [Andreassen 2015] and the internet in general [Davis 2001]. However, neither claim is yet supported by empirical evidence.

There is some evidence that operant conditioning in the form of positive reinforcement may impact on unwanted habitual behaviours: a review of smoking cessation interventions during pregnancy found that strategies including the provision of incentives were the most effective [Lumley et al. 2009]. Positive reinforcements in the form of virtual rewards are common in DBCIs, but this is not a panacea for motivating behaviour, as we discuss further in Section 4.3.1. A recent implementation of variable reinforcement found some evidence that operant conditioning can change and maintain more secure behaviour, with a follow up period of 40 days, although the sample sizes were small [Villamarín-Salomón and Brustoloni 2010].

Negative reinforcement or punishment strategies are relatively rare in DBCIs [Kirman et al. 2010; Orji and Moffatt 2016]. This may be due to ethical concerns [Fogg 2002] and fear of disengaging users [Consolvo et al. 2009b]. Nevertheless, there is evidence that users are not necessarily deterred from interacting with DBCIs using aversive feedback [Foster et al. 2011]. One wearable DBCI that implements a punishment strategy is the Pavlok system, which allows users to trigger a mild electric shock (and/or beeps and vibrations) either manually or via sensors in order to punish a habitual behaviour they wish to break [Pavlok 2018], but the effectiveness of the device has yet to be demonstrated. Similarly, some researchers have implemented less painful punishment techniques, for example making interaction more tedious [Cowan et al. 2013; Cox et al. 2016; Foster et al. 2011], but none have been tested over the long term with a large user group.

2.3.3 *Theoretical issues.* A major theoretical criticism of behaviourism is its inability to explain higher-order behaviour involved in habits such as goals and conscious expectations of outcome. Behaviourism also cannot explain evidence that habitual behaviour can be triggered by cognitive constructs such as mood [Ji and Wood 2007]. Kihlstrom et al. 2007 argue that implicit learning involves some cognitive abstract representation of the knowledge, above and beyond the simple behaviourist associations.

2.4 Cognitive theory

Given the limits of behaviourism, we now turn to a key cognitive theory of behaviour change, the Theory of Planned Behaviour, which peers into the 'black box' of cognitive representations of external and internal behavioural drives.

2.4.1 Theory of Planned Behaviour.

Overview. The Theory of Planned Behaviour [Ajzen 1991] is a rational-action theory that specifies that intentions drive behaviour. A person's behaviour is determined by their conscious intention to perform that behaviour and their Perceived Behavioural Control, an internal assessment of their ability to perform the behaviour. This intention is itself determined by behavioural attitudes, perception of subjective norms relating to the behaviour and Perceived Behavioural Control [Ajzen 1991]. It is "the most extensively studied social cognition theory" [Hardeman et al. 2002].

Recent implementations and empirical evidence. Schneider et al. 2016 applied the theory to unpick motivations of 643 mobile fitness coach users, finding that attitude, subjective norm and Perceived Behavioural Control were generally good predictors of intention, although the levels varied across personality types. The authors acknowledge the theory's omission of possible nonconscious drivers of behaviour.

There is mixed evidence to support the theory from metareviews. Hardeman et al.'s review of 24 interventions 2002 found few studies actually using the theory to develop interventions, and a lack of evidence linking theory components to intervention outcomes. Webb & Sheeran's

meta-analysis 2006 indicates that the intention construct is insufficient to fully explain behaviour change, with "a medium-to-large change in intention ... lead[ing] to a small-to-medium change in behaviour". Crucially, an intention-behaviour gap persists, particularly in the presence of strong habits [Gardner et al. 2011; Webb et al. 2010b].

Theoretical issues. The Theory of Planned Behaviour is not a theory of behaviour change, and there is evidence that determinants of intention change over time [Suh and Hsieh 2016]. The theory thus has limited application to habits since they can only emerge in the presence of intentions that are enacted repeatedly in stable contexts. Sniehotta 2009 argues that the Theory of Planned Behaviour has major conceptual flaws, including no testable descriptions of how to modify intentions and therefore possibly behaviour. The theory omits context, habits and emotions as other possible determinants of behaviour [Jackson 2005; Sniehotta et al. 2014]. The inability of the model to deal with habits is particularly problematic: several studies [Triandis 1977; Webb and Sheeran 2006] have shown that interventions to alter intentions tended to impact on behaviour only where habits were not involved. Finally, assuming rationality where intentions are formed through conscious deliberative processes limits the theory's ability to explain nonconscious behaviours.

2.5 Integrated models

Integrated models try to provide more overarching models of behaviour. They address dissatisfaction with the polarised view from behaviourists, that individual behaviour is solely determined by the environment, and cognitivists, that behaviour is solely determined by internal cognitive factors [Bandura 1978].

2.5.1 Theory of Interpersonal Behaviour.

Overview. The Theory of Interpersonal Behaviour [Triandis 1977] extends reasoned-action theories by explicitly including habitual behaviour and the context. Habit (expressed as frequency of past behaviour) and behavioural intention interact with situational conditions to determine behaviour. More frequently enacted behaviour weakens the intention-behaviour relationship. Behavioural intentions themselves are the product of attitudes, social factors and affect (the experience of emotion), with affect providing a largely nonconscious input into behavioural decision making. Thus the intention construct includes both nonconscious and conscious components.

Recent implementations and empirical evidence. There have been relatively few implementations of the theory. The few DBCIs based on it tend to focus on issues of technology acceptance. Moody & Siponen 2013 used the theory to explore the use of the Internet at work for non-work purposes, finding evidence to support the model's key assumption that intention and habits both were significant in predicting the target behaviour. However, the research emphasised the social factors at play in the workplace, and it is not clear whether the results would generalise to other, less social domains. Gimpel et al. 2016 found that habit was a predictor of intention to use smartphones, although neither habits nor smartphone behaviour were measured.

Theoretical issues. Since the theory has been little used in DBCIs, it is difficult to establish its efficacy. This may be because it is not clear how to apply it to behaviour change interventions. Further, as we address in Section 6.2.3, using just *frequency of past behaviour* to approximate habits is insufficient. Danner et al. 2008 found evidence that past behaviour frequency only moderates the intention-behaviour relationship when information about context stability is also represented. Although the Theory of Interpersonal Behaviour includes "facilitating conditions", it does not directly address the role of such conditions in forming habits – i.e. context stability. It may not therefore adequately capture habitual behaviour.

2.5.2 Social Cognitive Theory.

Overview. Social Cognitive Theory [Bandura 1978, 2011] states that behaviour is determined by an interaction between existing behaviours, the environment, and personal cognitive, affective and biological influences. Social influence is a particularly important environmental factor. Social Cognitive Theory suggests that behaviour change arises from two sorts of belief: firstly that a given response will have a desired outcome, and secondly that the individual believes themselves capable of the response [Clark and Janevic 2014]. The theory predicts that desired behaviours are performed where environmental barriers are low and self-efficacy is high [Armitage and Conner 2000]. It also incorporates elements of behaviourism via the mechanism of reinforcement for learned behaviours.

Recent implementations and empirical evidence. Implementations of Social Cognitive Theory tend to emphasise the key construct of self-efficacy rather than testing the theory as a whole e.g. [Rabbi et al. 2015]. Some empirical evidence supports the interaction between self-efficacy and behaviour change: a meta-analysis of physical activity studies found 3 techniques "associated with significant increases in both self-efficacy and physical activity behaviour; 'action planning', 'reinforcing effort or progress towards behaviour' and 'provide instruction"' [Olander et al. 2013]. Nevertheless, the overall effect was small, several other techniques had non-congruent effects on self-efficacy and physical activity, and the authors found that the reporting of intervention techniques was "inadequate" [Olander et al. 2013]. In addition, the identified techniques are consistent with Goal Setting Theory [Locke and Latham 2006], and behavioural theories of reinforcement, so it is not clear what additional contributions self-efficacy might provide, either at the theoretical or empirical level. Overall, Armitage & Conner 2000 argue that Social Cognitive Theory-based interventions typically account for small- to medium- levels of variance in behaviour.

Theoretical issues. Social Cognitive Theory has been criticised for failing to encompass habituation [Martin et al. 2014]. The theory relies on conscious, rational processing of behaviour change intentions and outcome expectancies, which do not reflect the observed automaticity of contexts triggering habitual behaviour. The implication is that habits are hard to change because they are perceived to be hard to change. The theory suggests that self-management is the key to breaking habits [Bandura 1998], but it is not clear how individuals can deal with low levels of deliberative cognitive resources to perform self-monitoring and self-regulation. Further, although the model does include context as a behavioural determinant, the focus is on the impact of social pressures such as role models and social support. Again, this omits the phenomenon of habits ceding control of behaviour to contextual cues. Social Cognitive Theory does integrate both cognitive and behaviourist elements, yet its hybrid nature makes it difficult to ascertain its efficacy and has led to calls for it to be subject to more rigorous testing [Martin et al. 2014].

2.5.3 The Transtheoretical Model.

Overview. The Transtheoretical Model [Prochaska and Velicer 1997] or "stages of change" health behaviour model was derived from a study of 872 people attempting to give up smoking on their own [Prochaska and DiClemente 1983]. The model identifies six stages of behaviour change, ranging from precontemplation (not even considering changing behaviour) to actively modifying their behaviour and/or the environment, through to maintaining the new behaviour and possible relapse. The model also identifies a set of ten processes for change, each of which has a different suggested emphasis for any given stage [Prochaska and Velicer 1997]. Prochaska & DiClemente 1983 argue that the key drivers of movement between the stages are self-efficacy (belief in one's own ability to achieve a given goal) and decisional balance (weighing up pros and cons). This segmented model enables researchers to develop intervention strategies for each stage. *Recent implementations and empirical evidence.* The Transtheoretical Model has been used widely in health behaviour change, e.g. see a meta-review of 71 empirical studies on the Transtheoretical Model and physical activity [Marshall and Biddle 2001], and in DBCIs in general, e.g. [Lin et al. 2006]. Many implementations focus on the precontemplation stage, where participants require information about their behaviour in order to motivate change e.g. [Park and Gweon 2015; Southern et al. 2017]. Wittekind et al. 2015 used the model in an anti-smoking DBCI to measure participants' readiness to quit smoking rather than to design an intervention.

There are reasons to doubt the model's efficacy. A metareviews of smoking cessation interventions – the domain that drove the model – found no evidence for a significant effect of interventions based on the Transtheoretical Model [Jepson et al. 2006]. Aveyard et al. 2009 concluded that there was "no evidence that Transtheoretical Model-based interventions [are] effective".

Theoretical issues. The stages of change have been criticised as "arbitrary pseudo-stages" rather than genuine stages [Bandura 1998]. West 2005 lists several empirical challenges to the Transtheoretical Model, and argues that it should be discarded because it contains fundamental theoretical flaws. One key flaw in considering habits is that the Transtheoretical Model assumes that people make stable rational choices, rather than being subject to nonconscious influence such as impulses or habits.

2.5.4 COM-B model and the Behaviour Change Wheel.

Overview. The COM-B model [Michie et al. 2011] emerged from work in systematically reviewing and combining multiple behaviour change theories and frameworks relating to health behaviours. The model states that behaviour is determined by an interacting system with three essential components: capability, opportunity and motivation. Together with the Behaviour Change Wheel [Michie et al. 2014a, 2011], it provides comprehensive guidelines for behaviour change researchers to plan interventions. It essentially formalises the pick-and-mix approach.

The COM-B model addresses some gaps in rational-action models such as the Theory of Planned Behaviour by including nonconscious components like "impulsivity, habit, self-control, associative learning and emotional processing" [Michie et al. 2011]. The model includes both automatic and analytical processes in the motivation concept, which encompasses all brain processes that "energize and direct behaviour" [Michie et al. 2011], and is derived from the PRIME model [West 2006]. *Opportunity* includes all factors external to an individual that "make the behaviour possible or prompt it", while *capability* includes all factors internal to an individual that contribute to their ability to perform a behaviour [Michie et al. 2011].

Recent implementations and empirical evidence. Walsh et al. 2016 used the COM-B model and Behaviour Change Wheel to devise an app for physical activity over 5 weeks. Their intervention group, which featured feedback and information about discrepancy between current behaviour and goal, had showed a small but significant improvement over the control. Lee et al. 2017 recently employed the model in a context-aware sleep intervention. However, COM-B was combined with many other techniques and strategies (e.g. Fogg's Behaviour Model; goal setting theory; self-monitoring) and it is therefore difficult to make conclusions about the use of the model itself from their work.

Theoretical issues. One issue in using the COM-B model in DBCIs is that it is relatively new and therefore relatively untested in HCI [Cibrian et al. 2016]. The COM-B model and Behaviour Change Wheel are explicitly positioned as practical tools to design behaviour change interventions, rather than an explanatory theory, thus their specific application to habitual behaviour remains unclear. Further research is required to demonstrate how they can improve intervention efficacy [Michie

et al. 2011]. An important avenue of research is to explore how to use DBCIs to apply COM-B to dynamic contexts and individual preferences [Michie et al. 2013b].

Alongside more general psychological user models of behaviour, a number of models and methods have emerged that specifically relate user behaviour to technology. For reasons of brevity, here we outline Fogg's highly-cited research into technology-mediated behaviour change.

2.5.5 Fogg Behavior Model.

Overview. Fogg's work focuses on the concept of "captology", or technology as a persuasive force in behaviour change [Fogg 2002]. This approach establishes seven key strategies that computers may use to change behaviour: reduction, tunnelling, tailoring, suggestion, self-monitoring, surveillance and conditioning; and three different roles the computer may play in these strategies: as a tool, as media and as a social actor. The Fogg Behavior Model [Fogg 2009b] grew from this focus on technology and practical strategies. It is a general cross-domain model that proposes behaviour has three key determinants: motivation, ability and a trigger. All these elements must occur at the same time to generate a particular behaviour.

Fogg also created the Behavior Grid [Fogg 2009a], a taxonomy of 35 ways behaviour might change; and the Behavior Wizard [Fogg and Hreha 2010], which attempts to merge the previous two items; and the "Tiny Habits" model ([Fogg 2015] not yet published as a peer-reviewed model but already cited e.g. [Daskalova et al. 2017; Kuo and Horn 2017], involving changing the performance environment, breaking the required habitual behaviour into small steps and rewarding small step completion.

Recent implementations and empirical evidence. Establishing empirical evidence for the Fogg Behaviour Model is difficult because it is a process model rather than a theory. Researchers also tend to use it alongside other models, e.g. Cambo et al.'s work 2017 uses the Fogg Behavior Model with the Health Action Process Approach, which alongside a small sample size and short intervention period (1 day) makes it difficult to draw conclusions about the model. Sugarman & Lank 2015 used the model to inform a qualitative survey of methods to encourage reduction in electricity consumption, although again they incorporated elements from other theories e.g. operant conditioning.

Theoretical issues. Fogg's work is essentially a behaviour change principles approach [Noar et al. 2008]. The Fogg Behaviour Model is an attractively simple conceptualisation of behaviour with clear design implications: provide people with an appropriate trigger, motivation and ability to perform a wanted behaviour and it will occur. However, the underlying psychological mechanisms of change are less clear. Further, although Fogg sees the point of persuasive technology as "fundamentally about learning to automate behavior change" [Fogg 2009b], and his Behavior Wizard [Fogg and Hreha 2010] attempts to simplify behavioural repetition to form new habits, the Fogg Behavior Model has little to say about the automatic components of behaviour or routine behaviour [Oulasvirta et al. 2012]. Ferebee 2010 found ambiguities and a lack of guidelines in mapping existing interventions to the Behavior Grid.

2.6 Discussion of competing models

2.6.1 Theory Overlaps. Health behaviour change models contain many components, some of which are shared or overlap [Taylor et al. 2006]. For example, intentions are a key behavioural determinant in the Theory of Planned Behaviour, the Theory of Interpersonal Behaviour and Social Cognitive Theory, although the theories differ in the elements that determine that intention. The notion of *Perceived Behavioural Control* from the Theory of Planned Behaviour is similar to parts of the Theory of Interpersonal Behaviour's notion of *facilitating conditions* and Social Cognitive Theory's concept of *self-efficacy*. Common to these models is the implicit assumption that users form

intentions along rational, conscious lines. Most health behaviour theories assume that conscious attitudes and intentions, self-efficacy and social influences impact most on behaviour [Noar et al. 2008].

The Transtheoretical Model, the Fogg Behaviour Model and COM-B are *process* models, which focus on translating research into concrete behaviour change interventions rather than more theory oriented framework models [Nilsen 2015]. Framework models, by contrast, focus on explaining the determinants of specific outcomes [Nilsen 2015]. Process models may have greater utility to intervention designers than framework models, since the latter may omit specific implementation detail [Rogers 2004]. However, process models may not always highlight the underlying theory: although the COM-B model is explicitly couched in the PRIME model of motivation [West 2006], the Fogg models do not have explicit theoretical underpinnings.

There are also clear overlaps between the three-part models that address interactions between behaviour, the environment and internal cognitive factors (Fogg Behavior Model, COM-B, Theory of Interpersonal Behaviour and Social Cognitive Theory). Indeed, Lee et al. 2017 recently combined the Fogg Behavior Model and COM-B into a single approach which assumed a given behaviour occurred when "opportunity, ability, motivation, and a trigger all align".

One key common determinant of behaviour is *intention* (and related motivations), which we explore here in a little more detail.

Intentions & motivations. An intention is a decision to undertake a particular behaviour at a future point in time. An intention encompasses the person's motivation to perform the behaviour - the direction (to perform the behaviour or not) and the intensity (how much value they assign to that performance) [Sheeran 2002]. The Theory of Planned Behaviour, the Theory of Interpersonal Behaviour and Social Cognitive Theory all assume that intentions are key determinants of behaviour, and those intentions arise from the likelihood and desirability of the outcomes of a given behaviour [Deutsch and Strack 2010; Webb and Sheeran 2006]. However, it is not clear exactly how intentions drive behaviour [Bruin et al. 2012], nor how the theories see any interaction between habit and intentions.

Motivation is a value attached to a particular intention. Motivation is a key construct in the Theory of Interpersonal Behaviour, Social Cognitive Theory, the COM-B model and the Fogg Behavior Model, and is central in moving people from contemplation to active stages in the Transtheoretical Model. We agree that DBCIs require people to be consciously motivated to change their behaviour to engage in interventions at the very least at the outset. However, most models (with the key exception of COM-B) focus on conscious, rational motivations, omitting important automatic aspects of motivation including the impact of contextual cues, internal physiological states (e.g. hunger) and emotions. These automatic elements are specified in the PRIME theory of motivation [West 2006], on which COM-B is based.

Theory gaps. Despite some consensus on behavioural determinants between the theories, there are gaps in their ability to drive the design of behaviour change interventions. Not all the models are dynamic, or specify how their constructs or the relationships between them change over time. This limits their application to DBCIs that can adapt rapidly to their users and to changing inputs [Riley et al. 2011]. Some models omit the impact of the context, habit and/or emotions in determining behaviour.

There is little consensus on how to combine the overlapping constructs to change behaviour [Noar et al. 2008]. In particular, not all the theories explore how behaviour changes. For example, the Theory of Planned Behaviour predicts behaviour, rather than addressing how intentions can change over time [Suh and Hsieh 2016]. The exception is the Transtheoretical model, which explores when intentions change, but not how [Armitage and Conner 2000]. Many theories incorporate elements

of behaviourist operant conditioning, e.g. the use of positive reinforcement in the Transtheoretical Model and Social Cognitive Theory [Adams et al. 2017]. However, despite the crucial role the environment plays in behaviourism, many common theories emphasise individual/interpersonal variables rather than broader social/environmental variables [Davis et al. 2015; Taylor et al. 2006]. This is a crucial omission in their application to understanding and changing habits. Since contextual features are crucial to both triggering and forming habits, models that omit them are unlikely to capture habitual behaviour determinants adequately.

Habit itself is a key construct omitted from many behaviour theories, particularly health-related theories [Nilsen et al. 2012], despite compelling empirical support for its role as a moderator of the intention-behaviour link [Sheeran et al. 2017b; Webb and Sheeran 2006]. In general, models that assume a rational, deliberative process as a key determinant of behaviour (e.g. the Transtheoretical Model and the Theory of Planned Behaviour), are insufficient to explain habitual behaviour given the persistence of an intention-behaviour gap in the presence of strong habits [Gardner et al. 2011; Webb et al. 2010b].

Theories that include habit either mention it somewhat in passing (COM-B model) or restrict its determinants too narrowly (e.g. behaviourism's failure to incorporate cognitive constructs that operate during habit formation). Even theories that explicitly incorporate habit e.g. the Theory of Interpersonal Behaviour and Social Cognitive Theory, fail to explain how and why habits operate, which limits their practical application. Further, although COM-B, the Theory of Interpersonal Behaviour and Social Cognitive theory all include elements of nonconscious motivation, it is not clear how the conscious and nonconscious elements work together to determine behaviour.

A good candidate for filling this theoretical gap in DBCIs is Dual Process Theory. Despite Dual Process Theory being the "probably one of the most significant theoretical developments in the history of social psychology" [Gawronski and Creighton 2006], it has been little used in DBCIs [Adams et al. 2015; Orji and Moffatt 2016; Webb et al. 2010a]. The under-use of Dual Process Theory is significant because together with modern habit and goal theory, it directly addresses the intention-behaviour gap and allows us to address the research gap in understanding how to build habit-focused DBCIs. One reason for this under-use is because of a lack of a clear, practical framework to apply Dual Process Theory to DBCIs. Providing this is the motivation for the current paper.

2.7 Bridging the theory gaps

In the next section, we outline three theories that address habit formation and habit breaking: Dual Process Theory, modern habit theory and Goal Setting Theory. We argue that bringing these three theories together can bridge the research gap in understanding how to change habits using technology and therefore move to close the intention-behaviour gap. Dual Process Theory allows us to see how conscious and nonconscious forces interact to determine behaviour; modern habit theory indicates how these might combine to determine habitual behaviour; and Goal Setting Theory informs effective goal-setting strategies to help drive habit formation through behavioural repetition.

2.7.1 Dual Process Theory.

Overview. Dual Process Theory argues that behaviour emerges from two distinct sets of processes: Type 1 (broadly automatic, e.g. habits) and Type 2 (broadly conscious, e.g. behavioural intentions). Type 1 processes are nonconscious cue-driven, heuristic, impulsive, associative, contextual, automatic, parallel processes that operate at speed; while Type 2 processes are conscious goal-directed, slower, rational, considered, rule-based, abstract serial processes [Evans and Frankish 2009; Evans 2011]. This split roughly maps onto the behaviourist/cognitivist rationalist divide, with habits forming part of the Type 1 set. Not all the nonconscious, automatic behaviours triggered by Type 1 processes are habitual [Marteau et al. 2012]. People may act in line with an impulse in response to a cue or in line with nonconscious goals without the action becoming a stable, repeated behaviour (see Section 4.1.2 on Priming).

The crucial difference between behaviourism and Dual Process Theory is that habits rest on cognitive constructs, and thus may be altered using both cognitive and behavioural techniques. Dual Process Theory thus unites the behaviourist-cognitivist divide: behaviour is the outcome of an interplay between both Type 1 and Type 2 processes.

We have outlined the common assumptions of Dual Process Theory, but there is no one definitive version of Dual Process Theory [Evans 2008]. It is a family of theories emerging from multiple fields of research [Stanovich 2011], including Borland's CEOS model 2013; 2016, the Reflective-Impulsive model from social psychology [Strack and Deutsch 2014], and System 1-System 2 theory from behavioural economics [Kahneman 2011].

Recent implementations & empirical evidence. Few DBCIs currently use Dual Process Theories [Adams et al. 2015]. Orji and Moffat's review of 85 DBCIs since 2000 found zero implementations [Orji and Moffatt 2016]. However, DBCI implementation pilots are emerging. Examples include our pilot on smartphones [Pinder et al. 2017] and two studies by Adams et al. 2015 in "mindless computing". Phelan et al. 2016 recently used the theory to inform a qualitative investigation into privacy behaviour, finding that the dual process view helps to inform the "privacy paradox" where users' privacy behaviour is not consistent with their privacy concerns.

Behaviour change research as a whole has recently begun to advocate the targeting of Type 1 processes alongside Type 2 approaches [Bargh and Morsella 2010; Dolan et al. 2012; Marteau et al. 2012; Sheeran et al. 2013]. Dual Process Theories are being used increasingly in health behaviour interventions [Hofmann et al. 2008], e.g. Kremers et al. 2006 used Dual Process Theory to derive a practical framework to explore the impact of environmental factors on weight gain.

In terms of neuroscientific evidence to support the theory, action-outcome behaviour (cognitivist goal-directed behaviour) and context-response behaviour (behaviourist habits) are associated with two different sets of brain processes [Gasbarri et al. 2014; Graybiel 2008; Yin and Knowlton 2006]. Habits as context-response links are therefore somewhat distinct from their origins as action-outcome sequences [Maia 2009]. At an experimental level, Presseau et al. 2014 measured deliberative and automatic predictors for six different healthcare behaviours (e.g. providing weight advice; prescribing for diabetes) using questionnaires and found that both types predicted behaviour.

Theoretical issues. Since Dual Process theories are still little-used in DBCIs, it is difficult to establish their efficacy. Further, implementing Dual Process Theory is not trivial: there are multiple versions, and researchers are still actively developing the theory as it applies to behaviour change [Borland 2016; Wiers et al. 2013]. Nevertheless, regardless of the specific Dual Process Theory chosen, they agree on two key predictions: behaviour is an outcome of both Type 1 and Type 2 processes; and Type 1 processes (including habits) will dominate when Type 2 resources are depleted, during distraction, high cognitive load, time pressure, adverse mood and low self-control [Hofmann et al. 2008; Muraven and Baumeister 2000]. The relative importance of Type 1 and Type 2 processes as a determinant of behaviour also varies with personality [Sladek et al. 2006]. Thus the influence of Type 1 and Type 2 processes on an individual's behaviour will vary both over time and in comparison with other people.

Dual Process Theory helps us to understand the underlying mechanisms of translating cues into action. However, it does not in itself provide a practical framework of applying the theory to habit-targeting DBCIs. For this we need to examine two additional theories: modern habit theory and Goal Setting Theory.

2.7.2 Modern habit theory.

Overview. Modern habit theory also integrates both stimulus-response behaviourist theories and goal-directed cognitive reasoned-action theories, e.g. [Wit and Dickinson 2009]. We have outlined the key points and some empirical evidence for the existence of habit in Section 1.2. This section therefore addresses a recent implementation and some theoretical issues.

Recent implementations. Stawarz et al. 2015 investigated the formation of habits in-the-wild, focusing on the relatively simply behaviour of reporting what participants had for lunch. In a 4-week study, they found that automaticity (as measured by the Self-Report Behavioural Automaticity Index, outlined in Section 6.2.3) was hindered both by smartphone reminders and positive reinforcement [Stawarz et al. 2015].

Theoretical issues. Habit research is ongoing across multiple fields, with ongoing challenges in determining the exact mechanisms underlying habit formation [Tobias 2009; Yin and Knowlton 2006] and in studying its automaticity [Gasbarri et al. 2014]. Nilsen et al. 2012 suggest that there is a lack of empirical evidence supporting interventions based on habit formation theory.

2.7.3 Goal Setting Theory.

Overview. We include Goal Setting Theory [Locke and Latham 1990, 2002, 2006] because it explicitly explores how best to form goals to drive behavioural repetition when Type 2 processes predominate to support habit formation. It fills the theoretical gap in many theories lacking detail in how to specify intentions. The theory proposes that goals must be accepted by users to be effective, that feedback on goal progress is important, and that two key aspects of goal setting determines their efficacy: difficulty and specificity. Hard, specific goals are more effective than easy, vague ones. Contextual constraints are considered to be a moderator [Latham et al. 2017]. The original Goal Setting Theory focused on conscious goals, but research is moving towards incorporating nonconscious dimensions [Latham et al. 2017]. This move is partly in response to ongoing debate about the theoretical underpinnings of nonconscious goal priming, which we discuss in Section 4.1.2.

Recent implementations. DBCI researchers often employ Goal Setting Theory alongside other behaviour models and theories to augment their interventions. For example, Ding et al. 2016 used goal setting theory predictions within the design of a context-aware walking app based on the Fogg Behavior Model. They found some qualitative evidence that users liked short-term step goals rather than daily or weekly goals, but it is unclear whether the results can be generalised.

Empirical evidence. Meta-analysis indicates that specific, difficult goals improved performance compared to asking people to "do their best", with effect sizes from .42 to .80 [Locke and Latham 1990], cited in [Locke and Latham 2002]. However, there is evidence that who assigns the goal makes a difference: when the DBCI sets the goals, easier goals may be more effective. Lomas et al. 2017 examined learning game goals and found that when goals were self-selected, moderately difficult tasks were most motivating, whilst when they were externally assigned, easiest games were most motivating. Konrad et al. 2015 found evidence that adaptive, easy goals set by user's technology were more motivating than difficult goals in a month-long experiment with 65 participants.

Theoretical issues. The goal setting theory picture is complicated by a lack of consensus on the measurement of goal commitment [Hollenbeck et al. 1989] or indeed goal difficulty. When designing DBCIs with Goal Setting Theory, researchers should also take individual characteristics into account. Orji et al. 2017 found evidence that goals in line with goal setting theory are motivators for people high in extraversion and conscientiousness.

2.8 Summary

We have outlined a series of theoretical approaches to changing habits. Three theories in particular are good contenders to fill the theoretical gap in explaining habitual behaviour: Dual Process Theory, modern habit theory and Goal Setting Theory. To ease interpretation of how these three theories combine to address habits, we have brought them together in a conceptual framework to explore habit intervention points for DBCIs.

3 THE HABIT ALTERATION MODEL

3.1 Overview

The Habit Alteration Model (HAM) is a practical conceptual model that synthesises Dual Process Theory, modern habit theory and Goal Setting Theory so it can be applied more easily to DBCIs. A model's function is to provide a descriptive simplification of a phenomenon [Nilsen 2015]. The HAM provides a conceptual, theory-driven graphical simplification of how these theories suggest that external and internal factors combine to generate both habitual and non-habitual behaviour. Its purpose is to provide a practical tool to describe and assess habit-targeting DBCIs, as called for by several researchers [Aarts et al. 1997; Hollands et al. 2016]. It allows researchers to devise new interventions that do not solely rely on limited deliberative cognitive resources to change behaviour.

The HAM is shown in Figure 1⁸. Behaviour is a function of: (1) the *context* consisting of a set of *cues*; (2) Type 1 associative processes relating *cues* to behavioural *impulses*; (3) Type 2 deliberative processes generating explicit *intentions*; and (4) individual differences (e.g. impulsivity), which determine the relative impact of Type 1 and Type 2 processes on behaviour. The model is dynamic: at the *Filter* stage, *cues* flow through perception and Type 1 and Type 2 *attention filters* to create an input set. At the *Prepare* stage, Type 1 and Type 2 *memory processes* match these cues to potential responses, Type 1 *impulses* or Type 2 *intentions*. These compete to become a single *response* at the *Act* stage. Information from observed *response* and *outcomes* feed back into the model and therefore into both Type 1 and Type 2 processes. Solid lines indicate processes that run continuously; dashed lines indicate processes that may run. Note that the HAM is not intended to represent the various highly complex physical architectures that operate in the brain. Instead, we present it as a virtual conceptual architecture where the boundaries between the systems need not be rigid [Sloman 2002].

With sufficient repetition of simple behaviours in stable contexts, cycles around the *Filter-PrepareAct* stages become more automatic, and the corresponding context-response habitual behaviour links become stronger and proceed faster. People's behaviour then transfers from slower Type 2 to faster Type 1 processes, from the conscious right-hand side of Figure 1 to the nonconscious left-hand side. Habit disruption strategies aim to call on Type 2 deliberation to override automatic Type 1 processes. Although disruption can be employed to a user's advantage, e.g. in error checking or reducing technology over-use [Cox et al. 2016], disruption also makes behavioural *outcomes* less stable because its success depends in part on available cognitive resources.

Dual Process Theory predicts that any behaviour may be the result of the simultaneous influence of Type 1 and Type 2 processes [Kremers et al. 2006; Presseau et al. 2014]. *Impulses* to respond in a habitual way triggered by a given context compete with other *impulses*, and with *intentions* from Type 2 processes, to determine a *response* [Gardner 2015]. The dominant *response* is determined by the relative strength of the items on the *Potential Response stack*, and is influenced by cognitive resources and an individual's cognitive capacity and processing style [Sladek et al. 2006].

⁸Ham constructs are denoted in the text with italics.



Fig. 1. Habit Alteration Model: Context cues (F1) are filtered by both Type 1 (F2) and Type 2 (F3) attentional processes to form a set of inputs to subsequent memory processes of Type 1 (P1) and Type 2 (P3). These generate competing drivers to act (impulses and intentions) to populate the Potential Response stack (P6). These may be overridden by self-control (P4), and may face competition from intentions created by self-monitoring (P5). The resulting behavioural response (A1) and (optional) outcome feed back into the model. Solid lines indicate processes that always run; dashed lines indicate optional processes.

Next we outline in more detail how the HAM models how habits are triggered, then how they are formed.

3.2 Habit-trigger process

3.2.1 *Filter.* We start with a set of *cues* that make up a given *context.* These *cues* include external features of the environment such as physical locations and other people, and internal features such as mood or physiological drives such as hunger [Wood et al. 2014]. This broad set of all possible *cues* in a given *context* is first 'filtered' by perception processes. They are then again filtered via Type 1 *implicit attention processes* (F2) and, optionally, Type 2 *explicit attention processes* (F3). Type 1 implicit attention is influenced by mood, attitudes and stereotypes such that some *cues*

receive preferential implicit attention than others [Deutsch and Strack 2010]. Type 2 processes in the deliberative system may also use directed attention (F3) to select specific *cues* from the *context*. This top-down conscious attention has limited cognitive resources [Norman and Shallice 1986], so its ability to select *cues* is impacted by cognitive load. The end result of the filter process is a subset of *cues* as inputs to the potential response generation process, *Prepare*.

3.2.2 Prepare. The *cue* inputs are used by both Type 1 and Type 2 *memory processes* to generate behavioural schemas for action [Strack and Deutsch 2014]. Type 1 processes (P1) generate *impulses* from implicit memory while Type 2 processes (P3) generate *intentions* from explicit memory. These separate schemas compete to become enacted behaviour, via a mechanism to integrate competing, parallel inputs into a single behaviour [Bargh and Morsella 2010]. We represent this mechanism in the HAM as a *Potential Response stack* (P6).

Type 1 *memory processes* are fast, modular and parallel, so multiple *impulses* may be generated by the set of available *cues* and placed on the *Potential Response stack*. Not all automatic *impulses* that arise from Type 1 are habits. Habits are context-response *impulses* for behaviour that has been repeated in a stable context. Automatic goal *impulses* are instead goal-response links, allowing for nonconscious goal-driven behaviour when a goal acts as a *cue* [Aarts and Dijksterhuis 2000]. *Impulses* emerging from Type 1 memory processes may also be of a simple approach or avoid type [Keatley et al. 2013], for example instinctive behaviour to flinch from a loud sound. We explore the application of both goal-related and instinctive *impulses* in DBCIs in Section 4.1.2.

The Potential Response stack (P6) may also contain conscious, deliberative *intentions* arising from Type 2 deliberative *memory processes* (P3). These may arise from explicit goals via the mechanisms of *self-control* (P4) and *self-monitoring* (P5). *Intentions* that have been set using Goal Setting Theory are assumed to take priority on the stack. *Intentions* include the intention not to act, i.e. impulse stifling as part of *self-control* (P4), if an unwanted *impulse* is likely to be enacted. This ability requires that the contents of the *Potential Response stack* are to a certain extent accessible to conscious thought [Bargh and Morsella 2010]. *Self-monitoring* (P5) may also place *intentions* on the stack in line with goals in response to processing information from self-tracking.

The order of *impulses* and *intentions* on the *Potential Response stack* is determined by several factors: match with the particular *cue* [Norman and Shallice 1986]; affect towards the *cue* and/or *response* [West 2006]; and accessibility [Danner et al. 2008; Kahneman 2003]. Placing value on degree of 'match' means that behaviour enacted more often appears higher on the stack than less previously-enacted behaviour, since the match with particular *cue* will be stronger. Thus an *impulse* to perform a behaviour that has been repeated in a stable *context* will appear higher in the *Potential Response stack*: these are habits.

3.2.3 Act. A competitive winner-takes-all process determines which single behaviour is performed from the competing schemas on the *Potential Response stack* [Hofmann et al. 2009]. Any potential response (*impulse* or *intention*) that exceeds a certain *act threshold* (the pink dashed line in P6 in Figure 2) will be enacted if there are no rival potential responses [Wood et al. 2014]. Where competing potential responses cross the act threshold, arbitration using Type 2 processes is required, as detailed below.

Following a *response*, there may be a particular *outcome*, for example a corresponding change in the environment or a reward. Information on the response and outcome feed back into implicit and explicit memory processes and therefore may impact on subsequent *Act* phases [Sun et al. 2005; Wood and Neal 2007].

Act arbitration process. Figure 2 shows a zoomed-in view of different possible states of the *Potential Response stack* with respect to the *act threshold* (the pink dotted line), and the resulting

Fig. 2. Response outcomes from different Potential Response stack states: single impulses and intentions above the act threshold will be enacted, otherwise arbitration will occur. Responses may occur regardless of the presence of intentions - in state 1 and state 4, intentions may not be present (indicated by unfilled intention item). Height order indicates relative value in the stack.



behavioural response. Impulse A, impulse B and impulse C are *impulses* to perform behaviours A, B and C respectively; intention D is an *intention* to perform behaviour D. The relative value of items on the *Potential Response stack* is indicated by height order: the higher in the stack, the higher the relative value.

State 1 shows that where a single *impulse* crosses the *act threshold*, the *impulse* will be enacted regardless of *intention*: intention D may or may not be present. State 2 shows that where a single *intention* is strongly-held such that it alone crosses the *act threshold*, its target behaviour will be enacted regardless of competing, weaker *impulses*. State 3 shows that where conflicts between a Type 1 *impulses* and a Type 2 *intention* occur with similar implicit values so that both cross the *act*

threshold, Type 2 processes may be alerted to arbitrate [Wood and Neal 2007]. State 4 shows that arbitration may also be alerted to differentiate between competing *impulses*, regardless of *intention*.

Arbitration is the implicit core of many habit-breaking strategies. These strategies try to populate the *Potential Response stack* with conscious Type 2 *intentions* to compete with unwanted other potential responses in order to trigger deliberative arbitration. However, calling on Type 2 arbitration resources imposes cognitive load, which is not always available. Where arbitration cannot be performed, the most likely response is the highest-value *impulse* in the stack. Arbitration is hampered by multiple load factors including other Type 2 processes, ego depletion and time pressure and individual factors including working memory capacity and low trait self-control [Hofmann et al. 2009]. This explains why many effortful *intentions* to change behaviour fail: when Type 2 cognitive resources are low, default Type 1 *impulses* predominate [Hofmann et al. 2009].

In case 3, if cognitively costly arbitration cannot be carried out, impulse A will predominate because impulses appear more quickly in the *Potential Response stack* in excess of the *act threshold* than slower *intentions*. Where impulse A represents any Type 1 habit, generated by repeating a simple behaviour in response to stable contextual cues, habits will predominate when cognitive resources are low.

3.3 Habit formation process

How might people form habits such that the default behaviour is their desired behaviour? Repetition is key. Habit formation requires that a given response (*Act* stage) is repeated in a stable context, i.e. with a stable set of *cues* arising from the *Filter* stage. With repetition, the impulse to perform the given response emerges as highest in the *Potential Response stack* (*Prepare* stage), triggered by the stable *cues*. The *response* (*Act* stage) may then proceed without conscious attention, i.e. the intervention of Type 2 processes. In behaviourist terms, stimulus-response links have been established.

This habit formation process can be accelerated by rewarding the required response, i.e. operant conditioning, providing a rewarding *outcome* (A2). Rewards can promote the learning of context-response links [Wood et al. 2014]. A reward does not have to be explicit for a habit to form. For example, Conroy et al. 2013 found evidence for an habitual element in sedentary behaviour despite this behaviour not being explicitly rewarded or even consciously intended. The difficulties of providing rewards in the right way to form habits are addressed in Section 4.3.1.

The key task for DBCIs that seek to change habits is to foster behavioural repetition in a stable context. The traditional behaviour change intervention for habit forming is to use conscious Type 2 processes (the right-hand side of Figure 1) to drive repetition via mechanisms of self-monitoring, reminders and self-control. However, as we outline below, the HAM also allows this repetition to be targeted via nonconscious Type 1 means.

3.3.1 Role of emotion. Most behaviour change theory does not include emotion as a determinant of behaviour [McDonald 2014]. Emotions primarily impact on behaviour through indirect mechanisms [Baumeister et al. 2007], although emotion and mood impact on both Type 1 and Type 2 attention and memory processes [Bargh and Morsella 2010; Deutsch and Strack 2010].

At the *Filter* stage, emotions may act directly as cues in both Type 1 and Type 2 attention processes. Emotional state, or mood, may also shape how these attention processes work by biasing attention to both negative or positive cues [Fox et al. 2009]. At the *Prepare* stage, emotion –either conscious or nonconscious– can impact the order of items in the *Potential Response stack* [West 2006] and thus influence behaviour. Negative emotions and stress can also limit deliberative cognitive processing ability and therefore limit Type 2 interventions [Tice et al. 2001; Wood et al. 2014].

Finally, in the *Act* stage, affect can also serve as a reward, for example how 'enjoyable' or 'exciting' an exercise behaviour is [Kaushal et al. 2017].

3.3.2 Example. To illustrate the HAM model, consider the unwanted behaviour of snacking whilst watching television. In the *Filter* stage, the *context* (F1) includes the physical environment (e.g. sofa, television), the presence of the snacks themselves, and the snacker's current mood. While the snacker's conscious Type 2 attention (F3) is focused on the television programme, the snacker's nonconscious Type 1 attention (F2) is aware of the presence of the snacks via sight, feel and smell. Therefore *cues* of snack presence and mood form inputs to the *Prepare* stage. During this stage, these *cues* feed into Type 1 *memory processes* (P1) and generate an *impulse* to eat the snack, which appears on the *Potential Response stack* (P6). Assuming there are no competing *impulses* or *intentions*, and given sufficient previous repetition of the snacking behaviour in the presence of these *cues*, then the snacking *impulse* will exceed the *act threshold*. Since Type 2 cognitive resources are occupied in watching the television programme, levels of *self-monitoring* (P5) and *self-control* (P6) to stop snacking are low; the person does not place a counter-snack *intention* on the *Potential Response* stack (P6). The *response* at the *Act* phase is snacking behaviour. The *outcome* (A2) may be some intrinsic reward from the snack (*implicit feedback*).

4 STRATEGIES FOR DBCIS TO INTERVENE IN HABITUAL BEHAVIOUR

This section discusses the strategies our framework identifies to intervene in habitual behaviour based on habit literature. The HAM is used to illustrate the intervention targets. The analysis covers habit breaking and habit making strategies; some techniques are appropriate for both. For habit-forming strategies, the key question is how to move from Type 2 to Type 1 processes, from right to left in Figure 1. This is a movement from behaviour arising from slow, limited, serial, explicit systems to behaviour arising from faster, parallel implicit systems. For habit-breaking strategies, the key question is how to alter existing Type 1 processes without relying on cognitively effortful disruptive Type 2 resources. Points of intervention for habit-changing DBCIs are denoted by numbers F1-F3, P1-P6 and A1-A2 in Figure 1. The following section sets out a variety of different strategies to target these points of intervention.

4.1 Filter (Phase 1)

Removing or avoiding a specific cue that forms part of a cue-response link in an unwanted habit will mean that the undesired response is not initiated or therefore performed. This approach is challenging because the associative links in Type 1 processes are not available to introspection [Neal et al. 2012; Orbell and Verplanken 2015]. An individual is therefore unlikely to be aware of which cues are relevant to their unwanted habitual behaviours. An alternative strategy is to introduce cues that are likely to trigger required responses.

4.1.1 Alter context (F1).

Outline. The rationale for altering the context is to add or remove cues in order to affect which impulses and intentions arise in the Potential Response stack. Alterations may include changing cue properties such as ambience and size and/or placement, e.g. proximity and availability [Hollands et al. 2013]. Context alteration is suggested as particularly applicable in the unhealthy eating domain [Wansink and Chandon 2014]. With our unhealthy snacking example, a DBCI could suggest to the user to not buy the snacks in the first place, or suggest replacing them with a healthy snack whilst watching television.

Evidence. The primary implementations of context-altering DBCIs are ambient persuasive technology systems. They are designed to change behaviour and/or attitude unobtrusively by exerting

influence on people without requiring their focal attention [Ham et al. 2009]. Examples include altering a workspace to encourage people towards the stairs [Rogers et al. 2010] and augmenting a shopping trolley to influence consumer behaviour [Kalnikaite et al. 2011]. However, there is a lack of a strong evidence base for the efficacy of specific context-altering strategies [Hollands et al. 2013].

An additional context alteration strategy is the alteration of moods. There is some evidence to suggest that small mood-altering interventions can be successfully ported from psychology labs onto smartphones to alter moods [Meinlschmidt et al. 2016], although the technique has yet to be applied to DBCIs to alter habits.

Challenges. Determining which cues to alter is not trivial. Kremers et al. 2012 identified 35 broad environmental changes to promote change in food and activity behaviours, and the individual efficacy of such changes remains unclear. It is a particular challenge to detecting appropriate emotional cues for a given behaviour. We address the challenges of context detection further in Section 6.1. Large-scale ambient persuasive interventions can have high installation costs. This cost drawback has triggered research into altering "micro-environments", contexts on a smaller scale, for example product labelling or design [Hollands et al. 2013].

4.1.2 Priming (F1, P1, P6). An alternative context-alerting strategy is delivering cues via technology that users carry as part of their personal context, e.g. smartphones. Type 1 processes include associative memory links between cues and affective and behavioural responses [Strack and Deutsch 2014]. They can be activated using priming, the unobtrusive presentation of cues to activate relevant mental representations [Shalev and Bargh 2011]. Priming can increase accessibility of a goal concept [Kahneman 2003], making it more likely to be performed [Bargh et al. 2001]. Positive valence towards a concept can also increase its accessibility [Kahneman 2003; West 2006], providing opportunities for affective priming [Custers and Aarts 2007]. In the HAM, priming is providing a specific cue within the context (F1) that crosses the attention barrier to form an input to Type 1 Prepare processes (P1). These processes select a target impulse from memory. Ideally, this impulse is relatively high on the Potential Response stack (P6) and therefore likely to be enacted. If enacted and repeated in a stable context, then it will become a habitual behaviour [Wood and Neal 2007].

DBCIs may implement two forms of priming behaviour: the activation of instinctive paths to achieve certain behaviour, or the activation of learned constructs such as goals.

Instinctive paths.

Outline Several 'instinctive' context-response paths already exist within humans that may be used to prime behaviour change. Stanovich 2005 argues that these fast "genetic goals" are more easily primed than learned goals, and have the advantage of being more universal. Evidence of these instinctive paths include the influence of auditory [Spence and Shankar 2010] or other environmental cues [Wansink 2010] on eating behaviour, and the influence of apparent monitoring (displaying images of eyes) increasing compliance with honesty boxes [Bateson et al. 2006]. The latter example is particularly interesting because it implies the possibility that merely the act of appearing to monitor may be sufficient to increase compliance to some behavioural norm. With our unhealthy snacking example, a DBCI could display a prime indicating monitoring (e.g. an image of an observer), to encourage more positive snacking behaviour. This example also highlights issues of ethics, which we address further in Section 5.7 and Section 6.3.

Evidence Several pervasive systems have implemented instinctive triggers targeting difference senses, particularly in the fitness domain. Several sound-based DBCIs react to user heart rate by selecting [Nirjon et al. 2012], altering [Oliver and Flores-Mangas 2006] or auto-generating workout music [Bauer and Waldner 2013], while the Zombies, Run! [zombiesrungame 2018] app exploits the 'flight from fear' hard-wired instinct by cueing users to run faster using sounds of ravening hordes of zombies. An alternative sound-based system alters the user's walking sounds in order to change their perceived body weight and therefore change their gait [Tajadura-Jiménez et al. 2015].

One sight-based intervention, in the healthy eating domain, is the Mindless Plate. This prototype explored whether perceptions of food portion size could be altered using coloured plates, with somewhat encouraging short-term results [Adams et al. 2015]. Smell-based interventions are starting to emerge based on evidence that this approach can change behaviour over a week even when administered during a single night's sleep [Arzi et al. 2014]. Amores & Maes 2017 developed the Essence prototype necklace that uses smell as a nonconscious influence on mood and cognitive performance. An alternative 'instinctive' path implemented in DBCIs is that of social priming, e.g. [Aharony et al. 2011]: the idea that humans are predisposed to react to the cue of seeing another person perform a behaviour by responding with similar behaviour. This theory is supported to a certain extent by research in neuroscience [Kessler et al. 2006], but evidence for efficacy is mixed [Froehlich et al. 2010]. Nevertheless, new technology affords the opportunity to test large-scale social contagion in real-world settings through social networks, with some evidence that exercise is socially contagious across such structures [Aral and Nicolaides 2017].

Instinctive primes are a good candidate for research where deliberative strategies have repeatedly failed, for example in the healthy eating domain [Obrist et al. 2014; Pels et al. 2014; Wansink 2010]. Indeed, using primes within a food-consumption context to achieve changes in eating may be easier than changing people's generally poor ability to monitor their consumption [Wansink and Chandon 2014].

Challenges The key challenge in using instinctive primes is to identify the correct prime for a given behaviour. Once identified, the prime needs to be delivered in a sufficiently salient manner such that it crosses the implicit and/or explicit attention filters (F2, F3). If the primed behaviour is not repeated in a stable context, no habit will be formed. This is not just an issue of context detection, since enacting the desired behaviour is not guaranteed, given differences in individual responses and concurrent different states of the Potential Response stack (P6). Further research is required to determine how best to deliver the instinctive prime such that the related impulse appears at the top of the Potential Response stack P6, and is therefore likely to be enacted in the absence of competing intentions or arbitration (see Section 3.2.3). The technique is likely to be most successful to direct people during situations of high cognitive load (e.g. driving, working).

Nonconscious goals.

Outline If we can prime instinctive associations, to what extent can we prime learned associations, such as goal constructs, i.e. an association between a goal and the behaviour required to achieve the goal [Danner et al. 2011], to drive habit formation? This strategy may mitigate some of the challenges in identifying and delivering instinctive cues outlined above. Modern goal research indicates that goals, instead of definitively forming part of conscious deliberation in Type 2 processes, can not only be activated nonconsciously [Aarts et al. 2008; Stajkovic et al. 2006], but also operate nonconsciously [Chartrand and Bargh 1996; Förster et al. 2007; Pessiglione et al. 2007]. Priming goals results in more persistent accessibility of the related concepts than simply priming behaviour alone, at least until the goal-related behaviour is enacted [Bargh and Morsella 2010]. Primed impulses will therefore

have more value in the Potential Response stack (P6). With our unwanted snacking example, a DBCI could support the user by displaying the prime of a photo of themselves consuming an alternative, wanted snack, or even more abstract representations of goals of being thinner such as Giacommetti sculptures [Stämpfli and Brunner 2016].

Evidence Chen et al. 2014 found some evidence in a single session pilot that priming intentions increased user engagement in an exergame, while Custers & Arts 2007 showed that goal priming increased both accessibility and affective valence and impacted on effort to pursue a goal. The priming can also be *delivered* nonconsciously, e.g. subliminally. Caraban et al. 2017 applied subliminal priming in a browser plug-in by decreasing the opacity of key words and found some evidence of priming on subsequent item selection. Our own research in the area found mixed results of the immediate impact of subliminal priming across different primes (polygons, words, photos) delivered on smartphones [Pinder et al. 2017]. Evidence for the efficacy of goal priming as a behaviour change technique in general is mixed. Some evidence supports the approach [Sheeran et al. 2013, 2017b], while other evidence is more ambivalent [Wood and Neal 2007].

Challenges Goal priming requires work with users ahead of the intervention to instil the goal along Goal Setting Theory lines such that it can be primed. Priming with no pre-training implicitly assumes that participants already associate the target cue with the goal. Target cues need to be designed carefully to avoid ironic effects where instructions such as 'do not X' primes behaviour X [Earp et al. 2013]. There are some theoretical objections: Papies 2016, in line with the COM-B model, argues that goal priming procedures can only be successful where individuals also hold sufficient motivation, capability and knowledge to pursue it. We agree that successful priming depends on the appropriate selection of primes and construction of related goals [Ruijten et al. 2011] such that the goals cross the conscious /nonconscious divide in memory, P2 in Figure 1. Nonconscious goal priming experiments in psychology frequently use supraliminal tasks where the aim of the task is concealed – e.g. by tasking users with a word search where the target words relate to the goal of 'performing well' [Bargh et al. 2001]. However, word search or scrambled sentence tasks are difficult interventions for DBCIs using pervasive computing technology, particularly using small-screened or unobtrusive technology. Subliminal priming provides a possible alternative method of prompting nonconsciously. Subliminal priming can help to avoid user irritation [Ham and Midden 2010], and may be more likely to result in goal-related behaviour than conscious prompts [Glaser and Kihlstrom 2005]. We suggest further research into the use of subliminal goal priming to increase the accessibility of related behaviour, and thus increase the likelihood of the goal-related behaviour being enacted [Custers and Aarts 2007; Pinder et al. 2017].

Priming activation is distinct from habit activation. A habit is a learned context-response impulse, while priming activates multiple mental concepts in memory related to the prime [Wood et al. 2014]. To prime habit formation, DBCIs either need to be sufficiently context-aware to prime only within a stable context (to form new habits) or accept that their intervention may need to be persistent if stable-context-priming is not possible. The latter technological dependency may not lead to long-term behaviour change if the technology is abandoned [Renfree et al. 2016].

Further research is required to determine the most effective method of priming. Priming design choices include design of the prompt, duration, repetitions required [Pinder et al. 2017] and delivery mechanism, e.g. using opportunistic interventions tied to an unlock screen on a smartphone such as in [Pinder et al. 2016b] or during browsing [Caraban et al. 2017].

4.1.3 Alter cue salience (F2, F3).

Outline. The likelihood that a given cue gets through the implicit perception filter is determined by its salience. Thus, a more relapse-resistant strategy is to reduce the salience of contextual cues for unwanted responses, whilst also increasing the salience of cues for wanted responses, using Cognitive Bias Modification (CBM) techniques for attention biases [MacLeod et al. 2009], CBM-A. An attention bias is the tendency for a given cue to receive disproportionate implicit and/or explicit attention, points F2 and F3 respectively in Figure 1. Reducing this attention for unwanted cues then reduces the resulting unwanted response because the cue is less likely to become an input to Type 1 and Type 2 memory processes, and likewise the reverse with wanted cues. With our unhealthy snacking example, a DBCI could try to reduce attention bias for the snack by giving the user a serious game to pair images of their problematic snack with something they find revolting.

Evidence. Biases affecting attention can be altered by appropriate training [Hertel and Mathews 2011] and evidence from psychology labs of the potential for the technique to change behaviour is encouraging, e.g. in the healthy eating domain [Kakoschke et al. 2014]. However, there are relatively few DBCI implementations, and the ones that exist are primarily at pilot stages [Pinder et al. 2016a,b]. One randomised controlled trial porting CBM-Attention techniques onto smartphones found inconsistent results with only small effects on attention bias scores [Enock et al. 2014].

Challenges. CBM techniques face challenges in identifying the relevant cues that need increasing/decreasing in salience (see Section 6.1) and ensuring longevity of the newly-learned responses [Hertel and Mathews 2011]. The mixed empirical evidence indicates that additional research is required to determine how best to port these techniques from psychology labs to DBCIs.

4.2 Prepare: target the context-response link (Phase 2)

4.2.1 Train context-response (P1, P6).

Outline. The main technique to train context-response links is Cognitive Bias Modification for approach biases, CBM-Ap. An approach bias exists when an individual has a default action or impulse towards an unwanted cue, i.e. a Type 1 Prepare processes, P1 in Figure 1. For example, a smoker may have an approach impulse towards a cigarette. CBM techniques targeting approach biases train individuals to inhibit responses or reject these unwanted items, and to accept alternative wanted items. For example, the smoker might be trained to reject cigarettes and accept chewing gum, or with our unhealthy snack example, be trained to reject the unhealthy snack and accept a fruit alternative.

Evidence. Two CBM-Ap studies have found small but significant results following brief training with challenging participants and a long follow-up period. Wiers et al. 2011 trained alcoholics with 4x15 minutes lab training sessions. Participants used a joystick to push away images of alcoholic drinks on a desktop display, and pulled towards them images of non-alcoholic drinks. The training was sufficient to alter the intervention group's small approach bias for alcohol to a strong avoidance bias, reflected in marginally significant differences in relapse rates between intervention and control groups [Wiers et al. 2011]. Wittekind et al. 2015 trained psychiatric inpatients with a similar anti-smoking CBM-Ap over 4 sessions and found small but significant differences in self-reported nicotine consumption at 3-month follow-up.

Challenges. As with CBM-A, it is not yet clear how best to translate CBM-Ap from controlled conditions in the lab to pervasive technology in the wild. Cue identification may also be an issue, although it is easier for people to identify their unwanted approach biases than unwanted attention biases. The behavioural impacts of the two CBM-Ap studies was small, and intervention groups also received standardised Type 2 interventions (e.g. motivational interviewing). Nevertheless, evidence

of impact of the minimal Type 1 training indicates potential in using more pervasive technology to deliver larger numbers of training sessions in situ [Pinder et al. 2016b]. CBM-Ap has parallels with automating self-control, which focuses on response inhibition, as discussed in Section 4.2.5.

4.2.2 Implementation Intentions (F2, F3, P1, P3).

Outline. This approach specifically tries to bridge the gap between explicit Type 2 intentions and implicit Type 1 impulses. Implementation intentions are specific, concrete *if-then* plans that link particular *if* contexts (i.e. sets of cues) to a desired response, *then*. They aim to automate the *then* behaviour by delegating its control to the selected contextual *if* [Gollwitzer et al. 2005]. Implementation intentions are therefore special form of automated goals that can bridge the intention-behaviour gap [Webb and Sheeran 2006; Wood and Rünger 2016], and are argued to be a good strategy for habit formation apps [Stawarz et al. 2015]. Implementation intentions are conscious Type 2 processes that emulate Type 1 context-response habits. With our unhealthy snack example, a DBCI could support the user to form and rehearse an intention to counter the unwanted snacking, for example '*if* I am watching TV, *then* I will only snack on apple slices'.

The mechanism through which implementation intentions work is increased accessibility [Webb and Sheeran 2008], such that the resulting behavioural intention to perform the *then* response is highest in the Potential Response stack (P6). Through rehearsal, sufficiently concrete and relevant implementation intentions can become impulses, moving from Type 2 deliberative processes into Type 1 automatic processes [Einstein and McDaniel 2005]. Implementation intentions fit with the persuasive strategy of rehearsal found in several behaviour change models e.g. [Fogg 2002], but have a stronger theoretical grounding that fits with Dual Process Theory and modern habit theory.

Evidence. The evidence for implementation intentions is good: a meta-analysis found that they "had a positive effect of medium-to-large magnitude (d = .65) on goal attainment" [Gollwitzer and Sheeran 2006]. However, the meta-analysis did not consider whether the behavioural goals related to habitual behaviour or not, and other literature reviews note a heterogeneity in effect sizes [Hagger and Luszczynska 2014]. Prestwich et al. 2010 examined the use of both goal-based and plan-based SMS to boost implementation intentions to undertake daily brisk walking, finding that both conditions significantly increased the target behaviour compared to a control group, although the study was based on self-report.

DBCIs can support implementation intentions by providing support for intention formation and rehearsal. DBCIs can enable even inexperienced users to quickly learn if-then plans with multiple triggers or actions [Ur et al. 2014]. In-situ reminders [Tobias 2009] and rehearsal [Veling et al. 2014] can support the accessibility of implementation intentions.

Challenges. An effective DBCI using implementation intentions would ideally be able to advise users on which contextual cues are appropriately stable cues (i.e. *if* s) on which to base their *if*-*then* plans, together with appropriate reminders and rehearsal. However, difficulties of accurately monitoring context cues and behaviour, which we detail in Section 6.1, mean implementation is challenging. The evidence for habit breaking using implementation intentions is more mixed: some research suggests they are not good at controlling strong habits [Wood and Rünger 2016], while Sheeran et al. 2017b suggest that they have been successful in the smoking domain. Further research is still required into the effectiveness of technology-mediated implementation intention formation and rehearsal.

4.2.3 Provide information (P3).

Outline. The provision of information is common in both DBCIs [Pejovic and Musolesi 2014; Webb et al. 2010a] and in behaviour change interventions in general, e.g. in healthcare [Nilsen et al.

2012]. Here we define the provision of information as providing the user with data intended to alter their conscious decisional balance, e.g. to give them statistics on positive health outcomes for a given desired behaviour. The underlying "information gap hypothesis" with this approach [Cowan et al. 2013] implicitly assumes a rational choice model (e.g. the Theory of Planned Behaviour), where people will alter their conscious behavioural intentions to counter a given behaviour in the light of the information provided. With our unhealthy snacking example, a DBCI could provide the user with data on the adverse health effects of regularly eating unhealthy snacks.

Evidence. Providing information can in some circumstances change behaviour, albeit with a small impact: a meta-analysis of public information campaigns showed a weighted mean effect size of .05 on behaviour [Anker et al. 2016]. However, in applying information to the DBCI field, the evidence is mixed. Comber & Thieme 2013 used information provision as a strategy to counter habitual recycling behaviours with a just-in-time recycling-monitoring system. They found no impact of their awareness-raising on attitudes or on behaviour in qualitative feedback of 22 participants over 5 weeks. It is also difficult to evaluate its use in DBCIs because it is rare to find interventions that use the provision of information as a single tactic.

Crucially, the long-term effects of providing information on behavioural intentions are not stable. A randomised control trial (RCT) with longitudinal research (12 months) demonstrated that a strategy of advising people to do more exercise is ineffective in the long term [Hillsdon et al. 2002]. Likewise, within the eating domain, there is evidence of a 'mindless eating' gap of 15-20% of consumption, which persists within stable environments regardless of an individual's knowledge of nutritional information [Bellisle et al. 2004; Wansink and Chandon 2014].

Challenges. There are three key problems with providing information as a behaviour change strategy. Firstly, deliberative cognitive resources may not be available such that Type 2 processes can attend to, analyse and deliberate the information. Interventions often use disruption alongside information provision to demand Type 2 resources to process the information they provide [Verplanken and Wood 2006]. Secondly, people may not change their attitudes and/or behaviours in line with the information. Combor & Thieme 2013 found that although disruption (an audio signal) alerted their users to the need to attend to their behaviour, they were unsuccessful at providing the right information to change the behaviour in the desired way. Thirdly, Type 1 processes may bias the information itself as an input to Type 2 processes. For example framing effects, such as presenting the same information in either positive or negative ways, impact subsequent Type 2 judgements [Kahneman and Tversky 2000]. Nevertheless, this presence of cognitive bias in decision making can also present an opportunity. Lee et al. 2011 argue that varying the way in which information is delivered -specifically in providing the required behaviour as the default -can be successful, and this is a key tenet of choice architecture or 'nudge theory' [Thaler and Sunstein 2008]. DBCI designers therefore need to be aware that the way in which information is presented may also have an impact, beyond simple Type 2 decisional balance effects. Indeed, reframing a suggested option as a default, rather than outlining its advantages, is closer to a priming intervention than simple provision of information.

4.2.4 Just-in-Time reminders (F3, P3).

Outline. An alternative to simply advising people to change their behaviour is to leverage pervasive context-aware technology to provide just-in-time reminders to behave in a particular way [Moller et al. 2017]. To distinguish such reminders from alternatives (e.g. response priming or implementation intentions), we define just-in-time reminders here as direct, specific behavioural suggestions delivered at the expected point of enactment. Just-in-time reminders are applicable to habit formation (advising people to repeat a wanted behaviour in a stable context) and habit breaking

(advising people to refrain from an unwanted behaviour in a given context). With the unhealthy snacking example above, a user's phone could alert them to the unwanted eating behaviour and suggest an alternative.

Evidence. Although reminders can support the development of habits, they can have a diminishing effect [Tobias 2009]. They can also prompt reactance, where users try to reassert behavioural control from a perceived threat to it, (discussed in detail in Section 6.4.2), particularly where users are instructed to suppress thoughts of an unwanted behaviour [Palfai et al. 1997]. As with priming, if users are not paying full attention to the reminders, ironic effects may result [Earp et al. 2013]. For example, a DBCI that warns '*do not eat your favourite snack*' may inadvertently trigger the user to eat that snack. Further, even if the 'correct' response is performed, this may be dependent on the presence of the technology as a cue as part of the context-response link itself. This makes the new habit more fragile and susceptible to disruption [Renfree et al. 2016]. Stawarz et al. 2015 found evidence in a 4-week trial that electronic reminders increased behavioural repetition but impeded automaticity. Without automaticity, once the DBCI is removed, the behaviour is unlikely to persist.

Researchers have also made efforts to reduce the complexity of reminders to reduce their cognitive load. Ding et al. 2016 tried to identify low-disruption incidental points to deliver walking prompts using context-aware technology. Nevertheless, they found that users reported embarrassment at inappropriate suggestions.

Challenges. Behavioural repetition in a stable context is crucial to forming habits. Therefore justin-time habit-forming reminders must be context-aware. However, there are persistent technological issues with context detection (see Section 6.1), and few approaches to capturing behaviour also capture causal relationships between context and response [Banovic et al. 2016]. This limits their ability to remind in a habit-forming way.

From a theoretical perspective, Wood & Neal 2007 suggest that attracting attention to a given response may inhibit automated habit formation in favour of automatic goal pursuit. This in turn may affect the long-term sustainability of the required response; if a DBCI withdraws its reward for goal-directed behaviour, the response will cease. By contrast, a behaviour that is successfully habituated will persist as long as the set of trigger cues does not contain the intervention technology itself.

4.2.5 Train self-control (P4).

Outline. Self-control is the ability to "alter [your] own behavioral patterns so as to prevent or inhibit [the] dominant response" [Muraven and Baumeister 2000]. From a Dual Process Theory perspective, it is considered to be part of Type 2 processes [Metcalfe and Mischel 1999], although there is evidence that it can, through repetition, become automated into Type 1 processes where 'stop' behaviours are consistently mapped onto the same cue(s) [Verbruggen and Logan 2009]. Self-control could therefore provide a mechanism for people to resist habitual impulses to act in unwanted ways. Taylor et al. 2013 suggest that computer-based training to enhance self-control could take advantage of neuroplasticity to play a role in treatment for drug addiction alongside pharmacological treatment. Webb12 et al. 2010b express surprise that few interventions against addictive behaviour have used self-control strategies. With our unhealthy snacking example, a DBCI could be designed to support the user to train themselves to resist the snacks by using go/no go serious games [Lawrence et al. 2015].

Evidence. De Ridder et al. 2012 found evidence for a relatively large relationship between selfcontrol traits and habits and suggest that self-control is important in both making and breaking habits. Adriaanse et al. 2014 by contrast suggest that self-control operates mostly through breaking unwanted habits. Verbruggen & Logan 2009 used 5 studies to show that participants can be trained to automatically inhibit responses to unwanted items by practicing inhibition. Muraven 2010 argues that self-control training is generalizable, such that the mechanism can be improved by training small acts of conscious behavioural inhibition, regardless of either of domain or whether the subjects believed it would help. This strategy may therefore provide an important lever for behaviour change for circumstances when Type 2 processes are in control.

Some self-control DBCIs are starting to emerge. Cranwell et al. 2014 found that a 3-daily, 4-week training task significantly increased self-control test scores in an intervention group compared to a control group. Two sets of research have found some impact of an internet-delivered go/no-go food image task on weight loss [Lawrence et al. 2015; Veling et al. 2014]. Lawrence et al. 2015 trained participants to inhibit responses to unwanted foods with 4 sessions over 1 week and found a significant, medium sized drop in self-reported calorie intake. The study also reported that 88% of participants would continue the training if shown to be effective. The theory is that practising inhibition of impulses to eat via the training can impact on eating decisions, particularly for those who are overweight.

Nevertheless, it is not clear precisely how the self-control mechanism works, nor is the evidence for its efficacy consistent: Miles et al. 2016 found no effects on either Type 1 or Type 2 self-control following a 6-week self-control training programme.

Challenges. The key challenge is that effortful (Type 2) self-control is unlikely to be effective in the long term to change habitual behaviour because of a limit on deliberative cognitive resources [Wood and Neal 2007]. Where Type 2 self-control fails, old habits will re-emerge. The inability to introspect habits [Hagger et al. 2015] may also hamper attempts to limit unwanted behaviour through self-control where people are unaware of their habit cues. A strong association between affective state and self-control capacity [Economides et al. 2015; Tice et al. 2001] may also hamper behavioural persistence. However, repeated practice of self-control can itself become automatic and therefore part of Type 1 processes, similar to CBM-Ap training [Fishbach and Shah 2006]. As with automating other behaviour, repeating the same self-control or impulse-resisting control in stable contexts should ease the transition of self-control into automatic processes, moving from Type 2 to Type 1 memory processes. The new 'response' of self-inhibition must be repeated sufficiently such that this becomes the default response rather than the old, unwanted behaviour.

Effortful self-control has limited capacity [Baumeister 2002] and may even have ironic effects [Wegner 1994] prior to becoming automatic. Therefore, as with other Type 2 processes, it is unlikely to operate under conditions of high cognitive load, and may even have the opposite effect to that intended.

4.3 Act: target the habit response (Phase 3)

Targeting the response aspect of the context-response link may involve: the use of self-monitoring to reveal previously unknown response patterns; or operant conditioning on the response outcome (A2).

4.3.1 Revalue outcome (A2).

Outline. The key strategy to revalue outcomes is the providing rewards for 'correct' behaviour [Gouveia et al. 2015] or punishments for 'incorrect' behaviour [Kirman et al. 2010]. From a behaviourist standpoint, this is operant conditioning, the most common application of which is contingency management –to provide a positive outcome for a given behaviour to strengthen the context-response link [Maia 2009]. Rewards are not a necessary part of habit formation, for example

when the target behaviour is intrinsically rewarding [Lally et al. 2008], but they may accelerate its development into automaticity.

From a cognitivist standpoint, rewards increase the expected value of a given behaviour. It is the subject of some debate whether rewards operate on a nonconscious Type 1 behaviourist level or a conscious Type 2 cognitivist level [Capa et al. 2011; Neighbors et al. 2013]. Using the HAM model in Figure 1 to illustrate, rewards boost the position of the matching *impulse* (nonconscious Type 1) or *intention* (conscious type 2) on the *Potential Response* stack (P6). Virtual rewards are commonly used in DBCIs [Hamari et al. 2014; Orji and Moffatt 2016], although punishment strategies are much rarer [Kirman et al. 2010], as we outlined in Section 2.3.2. Such rewards are mostly designed to attract attention, which means they may be overlooked in conditions of high cognitive load. With our unhealthy snacking example, the user could reward themselves for consuming healthy snacks as an alternative, for example by transferring a small amount of money into a virtual jar for each healthy snack consumed, and/or punish themselves for consuming unhealthy snacks by giving a small amount of money away.

Evidence. Contingency management has shown a positive medium mean effect size in traditional behaviour change treatments [Prendergast et al. 2006]. However, in terms of virtual rewards deployed in DBCIs, evidence for efficacy is mixed, perhaps because they target Type 2 scarce resources for conscious processing of the reward. One short-term (10-day) study found no effect [Zuckerman and Gal-Oz 2014]; while qualitative research both supports [Fritz et al. 2014] and does not support [Munson and Consolvo 2012] the strategy. Adams et al. 2017 found that in a relatively long term intervention (4 months), small, immediate rewards had a greater impact on number of steps than larger, delayed rewards. Evidence for the long-term efficacy of *financial* rewards is also mixed: Volpp et al. 2008 found that although financial incentives achieved weight loss in the short term (3 months), weight loss was not sustained in the longer term (7 months) and the long-term efficacy is unclear. Stawarz et al. 2015 identified a key challenge in delivering rewards to drive the generation of automatic behaviour: positive reinforcement over 4 weeks hindered automaticity, possibly due to reactance. Zedelius et al. 2011 also found evidence that consciously-processed rewards impaired task performance.

Challenges. Challenges to implementing an effective reward strategy include issues of monitoring, designing rewards and reward schedules, and theoretical issues. Firstly, accurately monitoring context and behaviour deliver rewards smoothly (so that the user assigns credit from the reward to the correct action [Maia 2009]) is not trivial, as we discuss in Section 6.1. Crucially, the desired behaviour may be extinguished by inaccuracy: both when a given action no longer attracts the previous reward, or when a reward is received despite the appropriate action not occurring [Yin and Knowlton 2006]. Secondly, it is not clear how to apply results from psychology labs to designing rewards and reward schedules for DBCIs. From behaviourism, rewards delivered at certain intervals (interval schedules) promote context-response habit links, while rewards delivered on a given number of responses (ratio schedules) tend to promote action-outcome goal-directed links [Gasbarri et al. 2014; Yin and Knowlton 2006]. Continuous or very frequent rewards can support the acquisition of new behaviours, but develop behaviour that is easily extinguished [Villamarín-Salomón and Brustoloni 2010]. As noted above, consciously-processed rewards may not have the intended effect since deliberative cognitive resources are limited.

Finally, on a theoretical level, a potential problem with punishment or removal of rewards to change behaviour (what behaviourists call '*extinction*') is that the underlying associations are not altered: instead participants may simply learn to inhibit the unwanted behaviour in particular contexts [Bouton 2014; Redish et al. 2007].

A common mechanism related to rewards and behaviour in DBCIs is gamification. Gamification extends beyond virtual rewards into engagement. Given that behavioural repetition is crucial to habit formation, engaging users in habit-formation apps is crucial. Nevertheless, many health apps have employed gamification elements without consideration of the underlying behavioural theory [Lister et al. 2014]. Orji et al. 2017 provide evidence that a variety of gamification strategies including "competition, simulation, self-monitoring and feedback, goal setting and suggestion, customization, reward, social comparison, cooperation, punishment and personalization" have different perceived persuasive impacts on different personality types. Further work is required to identify the persuasive impact on behaviour of each gamification strategy. The tendency for DBCI trackers to employ multiple gamification strategies makes it difficult to determine why they succeed or fail. The Basis tracker tried to instil simple habitual behaviours by rewarding repetition [mybasis.com 2015], although it also used a gamification strategy to 'unlock' additional behaviour tracking, which may provoke reactance in users who wish to have these locked behaviours tracked at the start. Activity trackers also employ goal-setting techniques, virtual rewards such as progress badges, and enable social sharing of data [Fitbit 2018; Jawbone 2018]. Again, this pick-and-mix strategy approach does not incorporate evidence that different personality types respond differently to different motivational elements [Jia et al. 2016].

Applying virtual rewards and other gamification strategies is also not a cure-all. There is no definitive evidence that either financial or virtual rewards can survive the jump from psychology labs to real-world DBCIs. DBCI designers face an awkward set of choices: what response to reward and how to detect it; what form of reward(s) to use; on what reinforcement schedule; and whether to target conscious or nonconscious processes. Since context-detection is not a solved problem (see Section 6.1), designers also need to consider the adverse impact of wrongly reinforcing behaviour due to technical failures –whether in the wrong context, at the wrong time according to a reinforcement schedule, with the wrong reward, or targeting the wrong Type 1 or Type 2 process.

4.3.2 Self-monitoring (A1, P5).

Outline. Self-monitoring involves using information from self-tracking to form alternative intentions to act [Snyder 1974]. Self-tracking, the capture and presentation of information about an individual's behaviour, often has a role in revealing information that may be previously unknown to the user [Thaler and Sunstein 2008], such as the number of steps taken each day. It can be helpful to reveal the consequences of automatic Type 1 processes to Type 2 behavioural override mechanisms [Hermsen et al. 2016]. The "self-monitoring and feedback" approach is common in DBCIs [Orji et al. 2017]. Domains include energy usage [Brynjarsdottir et al. 2012], water usage [Kuznetsov and Paulos 2010; Laschke et al. 2011] and activity tracking [Fitbit 2018; Moov 2018], the latter because it is relatively easy to monitor [Ananthanarayan and Siek 2012]. Many activity trackers implement data analysis and reminder engines over and above simple data presentation e.g. Jawbone's Smart Coach [Jawbone 2018] and Moov's Fitness Coach [Moov 2018]. With our snacking example, the user could record the amount of unhealthy snacks that they eat in front of the television using a simple self-monitoring app to inform different behavioural decisions.

Evidence. One meta-analysis of 138 interventions found evidence that self-monitoring led to small-to-medium changes in health goal attainment [Harkin et al. 2016]. Self-monitoring weight, i.e. tracking the consequences of undesired eating behaviour, can be an effective long-term strategy in maintaining weight loss for more than 1 year [Butryn et al. 2007; Wing et al. 2006]. Butryn et al. 2007 hypothesize that frequent weighing increases a person's vigilance over their diet, but also note that reversing small weight gains (made apparent by frequent weighing) may be easier than reversing larger weight gains. Thus, self-monitoring weight enables the activation of smaller

adjustments to diet to intervene early. Kelley et al. 2017 recently provided qualitative evidence that self-tracking can have a role in revealing unhealthy or unexpected behaviour and thus provide motivation for change. Hollis et al. 2015 found some evidence in a 3-week study that simple emotion tracking can augment self-reflection, particularly for reflecting on reasons for failing to stop bad habits.

However, self-monitoring is not a panacea for behaviour change [Epstein et al. 2016; Stawarz et al. 2015]. Evidence shows that use of activity trackers tends to tail off in the longer term. A survey found that more than 50% of US activity owners no longer use their device, with 1/3 stopping using it within 6 months [Ledger and McCaffrey 2014], although the number of participants and survey procedure is unclear. Another study found that even when 17 participants were given \$1,000 to purchase smart devices to pursue their goal, almost 80% of devices were abandoned within the first two months [Lazar et al. 2015]. Even when trackers are not abandoned, there is evidence to question their long-term efficacy: a large-scale randomised controlled trial found that using a tracking device alongside self-monitoring of diet and activity resulted in less weight loss compared to a self-monitoring group alone [Jakicic et al. 2016]. Users report abandoning DBCI trackers because of a lack of accuracy, lack of understanding of the mechanics of tracking and difficulties in assessing accuracy [Yang et al. 2015].

Challenges. The mechanism linking self-monitoring with behaviour change is unclear. For example, self-weighing may function as an explicit input to Type 2 processes, affecting deliberative food choices. Alternatively, self-weighing may prime Type 1 processes, triggering nonconscious restraint [Brunner and Siegrist 2012; Pacanowski et al. 2015]. Self-monitoring may be an effective behaviour change technique in the eating domain since the remedy (eat less) is relatively simple; the technique may be less applicable to more complex behaviours and solutions. Further, the design of self-monitoring systems is crucial: the results of self-tracking should be designed so that it requires only limited deliberative cognitive resources to see and interpret results e.g. as implemented in "glanceable "versions [Consolvo et al. 2008a], or minimising effort by using aggregated wellbeing scores [Lin et al. 2012; Meyer et al. 2014].

Self-monitoring is rarely implemented as a stand-alone strategy in DBCIs, which makes evaluating its efficacy more difficult. It is often combined with goal setting, goal tracking and goal feedback mechanisms [Consolvo et al. 2008a; Fitbit 2018]. Several DBCIs employ self-tracking techniques that require users to track their own behaviour e.g. several habit-specific apps [Way of Life 2018; Xavier 2018], even when the behaviour is one that can be automatically tracked like activity [Chini et al. 2012]. Dual Process Theory and habit theory suggests that self-report of behaviour is unlikely to be accurate, a prediction borne out by empirical data in the activity domain showing substantial differences between self-report and actual sedentary behaviour [Clark et al. 2009; Colbert and Schoeller 2011].

Awareness of the outcomes of Type 1 processes does not necessarily translate into corrective action. A survey of self-trackers found that those who did not also track triggers and context did not have enough information to improve their tracking measures [Choe et al. 2014]. We therefore argue that a key underexploited use for self-tracking in behaviour change is the use of pervasive technology to track trigger cues for unwanted behaviour, rather than simply the behaviour itself.

4.4 Summary

Which strategies, then, hold the most promise to take advantage of this opportunity and create effective DBCIs that can change behaviours? Although the HAM bridges the theoretical gap in understanding how both Type 1 and Type 2 processes can impact on behaviour, there is a corresponding empirical gap [Hofmann et al. 2008; Sheeran et al. 2017b]. Specifically, there is a

lack of evidence relating the efficacy of each strategy to alter habits. Table 4 shows an overview of the identified strategies, together with the processes they target, which component of habits they relate to (illustrated with reference to the HAM Figure 1), a summary of the challenges and recent implementations. The challenges are: correct context detection; mixed or missing evidence for a strategy's efficacy in relation to habit changes; difficulties in translating strategies from psychology labs to in-the-wild DBCIs; and interventions that rely on limited deliberative cognitive resources. Context detection issues and empirical gaps affect most strategies.

Note that although habitual context-response links arise from a Type 1 process, strategies that target them do not necessarily need to be Type 1 or nonconscious themselves. Many of the strategies have both a Type 1 and a Type 2 component. However, given that cognitive resources are limited, we suggest the use of strategies where Type 2 processes are targeted via preparatory training rather than via just-in-time interruptions that are delivered when the user may not have resources to attend to the DBCI. Designers and DBCI research should focus on (a) interventions that aim to automate any conscious Type 2 processes into nonconscious Type 1 processes; (b) strategies where meta-analysis shows evidence of impact on behaviour either through digital or non-digital behaviour change intervention. Based on these criteria, we suggest that implementation intentions and automation of self-control are good candidate strategies for future habit targeting DBCIs.

A caveat to employing implementation intentions is the ongoing challenges in context detection. From Table 4, it is clear that all the suggested strategies, except for providing information, may be hampered by this issue. This may explain why information-providing DBCIs are common. Nevertheless, where cues can be identified, for example in the healthy eating domain, DBCIs should exploit interventions that: do not tax deliberative cognitive resources: alter the context where the costs are not prohibitive; alter the context salience through CBM-Attention techniques; use CBM-Approach and priming. Priming and CBM techniques try to directly target Type 1 processes (Priming targets F1 & P1; CBM-Attention targets F2; CBM-Approach targets P1 in Figure 1). CBM techniques may be used to simultaneously try to lessen the impact of a cue linked to an unwanted response in favour of trying to increase the availability and/or liking of a cue linked to a wanted response. These techniques, by focusing on automating the required response, reduce the likelihood of reactance, and can support habit formation where the automated behaviour occurs in a stable context.

Strategies that target only Type 2 processes in a just-in-time manner (targeting P3 in Figure 1) are unlikely to succeed in changing behaviour with a large habitual element both because they require sparse cognitive resources and because of the challenges in delivering a just-in-time system. Nevertheless, the use of self-control training (targeting Prepare - Type 2) is a candidate for future research as a strategy for both habit-making and habit-breaking because of evidence that it may itself become automatic through appropriate practice, resulting in it emanating from Prepare - Type 1 processes instead, making it more persistent. We need more research into appropriate revalue outcome strategies to form and break habits given the issues we have identified in providing variable rewards in attributable ways.

						I	ں ار	halleng	s	
								ээ		
		HAI	M phase	and tai	'get		ис	иәрілә б	s	
	Fil	ter	Pref	are	Acti	no	oitoətəp t	uoiti Juissim	аңтың әл	
Strategy	Type 1	Type 2	Type 1	Type 2	Type 1	Type 2	хәџиоЭ	∕ b9xiM Transla	itingo)	Recent implementations
Alter context		•					•			[Kalmikaite et al. 2011; Rogers et al. 2010]
Instinctive primes							•	•		[Bateson et al. 2006; ?]
Goal priming							•	•		[Chen et al. 2014; Magaraggia et al. 2014; Pinder et al. 2017]
Alter cue salience							•	•		[Dennis and O'Toole 2014; Enock et al. 2014; Pinder et al. 2016b]
Train context-response							•	•		[Heitmann et al. 2017; Pinder et al. 2016b; Rabinovitz and Nagar 2015; Wittekind et al. 2015]
Implementation intentions							•			[Brevers et al. 2017; Cullen et al. 2016; Pinder et al. 2016c; Veling et al. 2014]
Provide information								•	•	[Comber and Thieme 2013]
Just-in-time reminders							•	•	•	[Ding et al. 2016; Klasnja et al. 2015]
Self-control							•	•	•	[Blackburne et al. 2016; Cranwell et al. 2014; Lawrence et al. 2015; Miles et al. 2016]
Self-monitoring							•	•	•	[Hollis et al. 2015; Kelley et al. 2017; Moov 2018]
Revalue outcome							•	•		[Adams et al. 2017; Fitbit 2018; Khot et al. 2015]
						Dire	ct targe	t	Indir	ect target

^aNote that self-control also has a Type 1 component that is less affected by cognitive limits than its Type 2 counterpart.

Table 4. Strategies, HAM phase and component, and Type 1 / Type 2 targets

:38

If the target behaviour is not repeated in a stable context, then there may be a role for just-in-time interventions to 'remind' people of their behavioural intentions at opportune moments, if this can be done without attracting reactance. DBCIs can also support users to form intentions in line with the Theory of Planned Behaviour, augmented with Goal Setting Theory, because of the strong empirical evidence for its efficacy in Type 2 situations. These interventions should be accompanied by priming and self-control strategies as outlined above. However, habitual behaviour should not be expected to result because of the absence of a stable context. The COM-B model and the Behaviour Change Wheel can also be used to ensure wide consideration of other non-habitual strategy options.

We address the issue of context detection in depth in Section 6, alongside a set of general difficulties in designing and evaluating DBCIs. First, in the next section, we bring together a set of design principles in applying dual process interventions for habits.

5 DESIGN PRINCIPLES

In applying the HAM to DBCIs targeting habits, we have identified several common design principles. These principles clarify the lessons learned and provide a starting point for intervention design. They also point to further challenges, which we expand on in Section 6. Some of these principles, such as understanding target behaviours, context, tailoring and ethics, are common to all DBCIs. However, there are specific issues within these principles that habit-targeting DBCIs need to consider.

5.1 Understand and simplify target behaviour and context

As with all DBCIs, the starting point is to clearly identify the target response [Michie et al. 2014a]. For DBCIs seeking to form habits, there is a particular advantage to simplifying the target behaviour as much as possible. The simpler the target response, the faster automaticity can occur [Lally and Gardner 2013; Wood and Neal 2007]. Since we define habits as behavioural impulses triggered by contextual cues, understanding the context is crucial. These cues may include those internal to the user, e.g. mood or physiological states like hunger, as well as more obvious external cues such as location and time. Habit-forming DBCIs should select the smallest possible set of salient cues to form a 'stable context' as a habit trigger, since simpler context causation models support faster habit formation than complex ones [FitzGerald et al. 2014]. DBCIs should avoid re-using contextual cues already in use in context-response links, because cues associated with multiple responses are more likely to cause response conflict, triggering arbitration and conscious Type 2 resources, and thus hinder habit formation [Wood and Neal 2007]. Habit-breaking DBCIs need to isolate the particular set of context cues that trigger an unwanted response. For example, when considering an intervention to stop snacking, the DBCI designer needs to assess when, where and why the unwanted behaviour occurs. If they determine that the snacking occurs at home under conditions of being distracted by the television, the DBCI can isolate specific contextual cues related to the television to provide the basis of an intervention, for example within an implementation intention. It is important to note that this level of rich context-detection is not trivial: Section 6.1 discusses the specific challenges such detection.

Context cues may also include responses themselves via chaining. Chaining is the linking of a new behaviour to an existing habit, which can support habit formation [Judah et al. 2012]. Within the HAM, successful chaining means an existing behavioural response acts as a cue input to trigger the new required response. For example, to promote healthy snacking in the presence of habitual TV watching, the user should be encouraged to make healthy snacks available next to the TV watching area.

5.2 Type 1 / Type 2 tailoring

It is a general DBCI principle that interventions should adapt to individual users [Ijsselsteijn et al. 2006; Orji et al. 2017; Ranfelt et al. 2009]. The HAM requires a specific form of tailoring because individuals vary in relative influence of Type 1 and Type 2 behaviours [Sladek et al. 2006]. For example, one individual may be more impulsive or susceptible to temptations such as the presence of consciously unwanted snacks than others. In these circumstances, a DBCI may need to intervene earlier in the unwanted behaviour process, e.g. by altering the context to try to prevent a user from buying unwanted snacks in the first place to remove the tempting cue. Individual users at different points in the dynamic process of habit formation or breaking may also require different sorts of intervention. For example, in the early stages of habit formation via implementation intentions, a user may need a higher level of support via intention rehearsal and reminders. These reminders should decrease over time as behavioural automaticity emerges to avoid technological dependence [Renfree et al. 2016] and possible reactance, as we discuss in principles Section 5.4 and Section 5.6 below.

5.3 Design for Type 1 and Type 2 congruence

Since the HAM shows that behaviour may result from simultaneous influences of Type 1 and Type 2 processes, the most effective habit-focused interventions are likely to be those that arise from influencing Type 1 and Type 2 processes in congruence. We have argued above that a likely failure of many interventions is the focus on Type 2 processes, undermined by incongruent Type 1 default processes. We caution against a corresponding myopic focus on Type 1 processes only without ensuring that the behaviour change participant is at least also consciously motivated to change their behaviour in a given direction. For example, priming can only be successful where a person already has relevant cognitive constructs motivated towards the given behaviour [Strahan et al. 2002]. If, in our unhealthy snacking example, the person has no interest or motivation in changing their snacking to a healthier alternative, priming them to switch snacks will not be successful. Congruent DBCIs that focus on Type 1 and 2 processes in unison would provide support for user behaviour change regardless of levels of user attention, deliberative resources and individual differences in Type 1 / Type 2 dominance.

5.4 Design for persistence

Habitual behaviour change is a long-term process. Automatic behaviour only emerges over time, so habit-focused DBCIs need to be viable over the longer term. One approach to establishing more persistent interventions is opportunistic training or incidental interaction [Ding et al. 2016; Dix 2002], fitting into and/or appropriating existing user behaviours. This design fits well with both COM-B and Fogg Behavior Model opportunity constructs. DBCIs should also follow the principle of synergy not substitution [Sellen and Whittaker 2010]. The aim should be to develop interventions that complement, and indeed augment [Rogers 2006], how our brains work to leverage the habit system, rather than leaving users dependent on their machines to substitute for unwanted brain processes. In particular, using just-in-time reminders based on technology could introduce a dependence on that technology, if it becomes part of the context-response link. The response is then more fragile, and will not occur if the technology context is abandoned [Renfree et al. 2016]. Technology abandonment is a known issue in the fitness tracker domain [Goodyear et al. 2017; Yang et al. 2015]. DBCIs instead should use training paradigms such as cognitive bias modification to alter default reactions to cues, create dependence on stable contextual cues, for example via Implementation Intentions, or co-opt existing behaviours such as unlock gestures as training methods to deliver cognitive bias modification or priming interventions [Pinder et al. 2016b, 2017].

For example, a smartphone DBCI might present a serious game on every unlock that requires a user to repeatedly pair an image of a television with an image of the healthy snack the user wishes to consume while watching television. Note that the requirement to tailor to Type 1/ Type 2 differences over time may also require the DBCI to determine when to self-destruct once it is no longer necessary to the performance of the target behaviour. We explore the evaluation implications of a longer time frame in Section 6.2.1.

5.5 Design for multiple points of intervention

DBCIs should look beyond the main technology platform of the DBCI to determine whether other interventions (both digital and analogue) can be included, given evidence that multiple component interventions (e.g. apps plus face-to-face counselling) can outperform apps alone [Schoeppe et al. 2016]. For example, a DBCI could instruct a user to print photos to act as persistent primes to consume a particular healthy snack in a particular location, in addition to supporting them to form and rehearse implementation intentions to consume that snack in the given location.

5.6 Design for reactance

Interventions that attempt to change a user's behaviour threaten a user's autonomy [Roubroeks et al. 2011], and may provoke reactance. Reactance occurs when users respond to a perceived restriction in behavioural freedom by trying to regain that freedom [Brehm 2009]. It may cause a user to disengage from a DBCI entirely or they may take steps to limit perceived infringements of freedom, e.g. by disabling notifications. Reactance may be triggered by the content of a persuasive message [Roubroeks et al. 2011] or the timing of its delivery. Inappropriate suggestions, lack of personalisation and monitoring errors may damage the credibility of the system [Ding et al. 2016; Segerståhl et al. 2010] and thus trigger reactance. Shame (e.g. from failing to achieve behavioural goals) or "excessive competition" may alienate users [Consolvo et al. 2009a].

DBCI designers should therefore ensure that just-in-time behavioural directions (i.e. those that directly threaten users' behavioural autonomy) should only be delivered when the system has confidence that the timing is appropriate. For example, to avoid issues of inappropriate timing, one strand of Ding et al.'s intervention suggested that users continue with existing walking behaviour, rather than simply directing them to start walking [Ding et al. 2016]. Where it is not possible to make such judgements of appropriateness, DBCI designers may instead choose to use a low-reactance alternative, e.g. priming or implementation intentions. For example, instead of issuing directives to start eating a particular healthy snack, a DBCI might support a user to form an implementation intention linked to snacking moments, such as '*if* I am watching TV, *then* I will eat the healthy snack'. We explore the challenge of reactance further in Section 6.4.2.

5.7 Design ethically

All DBCI designers have a moral duty to ensure that their interventions are ethical. This is a particular consideration where techniques are used to target Type 1 nonconscious processes, e.g. subliminal priming and cognitive bias modification. There is a certain amount of 'moral panic' about subliminal priming, despite evidence that the technique is only effective where the priming targets a goal the user is already motivated towards [Strahan et al. 2002]. As shown in the HAM, priming and other Type 1 targeting techniques can only activate pre-existing associations within memory.

Given these concerns, it is important for users to provide active and informed consent to their DBCI based on full disclosure of how the DBCI works. For example, a DBCI that uses subliminal priming should be clear both about the prime used and the intention with which it is presented; a cognitive bias modification intervention should be clear about the end-goal of the training. The

ultimate aim of habit-focused DBCIs is to build 'good' habits, persistent behaviours that people are not fully aware of enacting. Therefore, it is even more critical that people are able to give fully informed consent about the implications of habit-targeting DBCIs. Ultimately, users should be in control of their DBCIs, rather than the other way round [Rogers 2006]. They need to understand what the DBCI is doing and why, and have some ability to measure its impact and to remove and/or reverse its effects if required. This principle is often violated with non-digital behaviour change techniques such as tobacco packaging warning messages [Peters et al. 2013]. We address the ethical challenges further in Section 6.3.

6 CHALLENGES IN DELIVERING HABIT-FOCUSED DBCIS

This section brings together the challenges we have identified in designing and evaluating habitfocused DBCIs. The intention is to highlight areas of future research, and to assess what is possible given current technology.

6.1 Context detection

There are a number of technical challenges to implementing just-in-time context- and/or responsetargeting strategy such as "anticipatory" interventions [Pejovic and Musolesi 2015]. To detect and intervene in habitual behaviour, the technology needs to:

- a) accurately detect and/or predict a response;
- b) accurately monitor the context to determine either which cue(s) is prompting the unwanted response (for habit-breaking) or suitable candidate contextual cue(s) on which to attach a wanted response (for habit-making); and
- c) direct the user to either avoid the context for the unwanted response (habit-breaking) or alert them to the context for the wanted response (for habit-making) at the right time, in such a way that the user complies and without causing irritation; and
- d) optionally also provide rewards for desired responses and punishments for undesired responses consistently to the best schedule for the given context.

All of these steps are not solved problems. For steps (a) and (b), problems of accurate behaviour and context-detection are well known within ubicomp [Bettini et al. 2010; Rogers 2006]. The accuracy of response tracking also varies with domains: capturing smoking or eating is more difficult than using smartphone and/or activity tracker accelerometers to capture sedentary behaviour. Research is ongoing to capture non self-report data in both the smoking e.g. [Scholl et al. 2013] and eating domains e.g. MyBehavior 2.0's crowd-sourcing of calorie information of food pictures [Rabbi et al. 2015]. Yet even tracking activity via accelerometers is not trivial: proper evaluation metrics for activity recognition are not being sufficiently considered [Lukowicz et al. 2012].

A context cue may include a physical location, a particular time period, the co-presence of others or cognitive constructs such as mood [Ji and Wood 2007] or some representation of a goal. The first two are relatively easy to track. Tracking the presence of other people depends on them carrying (and having configured in a certain way) identifying technology such as Bluetooth or users accepting more invasive emerging technology such as skin-mounted RFID tags [Ziai and Batchelor 2011]. Tracking cues that are internal to the user like mood and emotions is much more difficult, despite advances in physiological computing [Fairclough 2009; Hernandez et al. 2015], the increasing use of Experience Sampling Methods [Aharony et al. 2011; Rachuri et al. 2010] and behavioural causation analysis on smartphones [Pejovic and Musolesi 2015].

It is also difficult to predetermine which cues from a given context pass through perception and implicit filters (see Figure 1) for a given individual. A target context suitable to be selected as a trigger for habit formation needs to be "sufficiently salient in daily life that it is encountered and detected

frequently and consistently" [Gardner et al. 2012b]. It is not clear what level of detection either in Type 1 or Type 2 processes must be achieved for a context to become a trigger, nor how technology might determine which individual cues within a context have been 'noticed'. Nonetheless, it is an open question the extent to which users can tolerate inaccuracy. Minor accuracy issues are likely to pass unnoticed and may still provide the user with useful information about their behaviour, context or mood.

Given these difficulties, it is perhaps not surprising that relatively few DBCIs use context analysis [Honka et al. 2011; Stawarz et al. 2015]. Implementations are starting to emerge [Lee et al. 2017; Naughton et al. 2014,?; Rabbi et al. 2015], albeit ones partially reliant on self-report. Lee et al. 2017 built a system to support users to create their own context-aware if-then plans to deal with sleep problems (e.g. to remove technology from a certain location at a certain time), while the MyBehavior system aims to provide context-aware strategic behaviour change suggestions (e.g. to continue with an existing walk) [Rabbi et al. 2015].

Our HAM framework allows internal conditions such as mood to form part of the Context or set of cues that can trigger habitual behaviour. However, few DBCIs use automated mood tracking, although (i) some trackers use other physiological markers e.g. respiration as an indicator of emotion [Spire 2018] and electrical brain activity as an indicator of calmness [Muse 2018], and (ii) self-report of emotive states is starting to appear, albeit largely as a peripheral characteristic [Hollis et al. 2015]. Perhaps the most promising avenue of research is the use of machine learning techniques on smartphones to estimate context, behaviour, and internal cognitions such as mood [Burns et al. 2011]. Nevertheless, the research is still at a preliminary stage and training of mood involves relying on self-report. As we outline in Section 6.2.3, self-report is not reliable in all circumstances.

For step (c), directing and interrupting users can trigger reactance, where users reassert their own control over their behaviour in response to perceived threats. As we discuss further in Section 6.4.2, this is a key challenge. In any case, directing user behaviour via just-in-time prompts is a difficult strategy. Firstly, the prompts may be delivered as persuasive messages via notifications on wearable technology (e.g. smartphones, smartwatches). However, research into attention shown to interruptions on smartphones shows that users may take more than 10 minutes to respond to a notification on average [Pejovic and Musolesi 2014], which implies that just-in-time interruptions on such devices are likely to fail. Secondly, regardless of prompt format, interruptions on any platform may also be counter-productive: interruptions deplete self-control resources [Freeman and Muraven 2010], and can cause disruption even if they contain important or useful information [Mehrotra et al. 2016].

To counteract the interruption problem, indirect forms of prompting could be used as a proxy for notifications, for example vibration and/or audible triggers as in SitCoach [Dantzig et al. 2012], although other research found that most users ignored vibration interruptions [Hirano et al. 2013]. Alternatively, interventions can use one of the emerging interruptibility prediction models [Kapoor and Horvitz 2008; Mehrotra et al. 2015], where interruptions are only issued where the potential cost of the user not heeding the interruption outweighs the potential cost of interrupting now and getting the timing wrong.

Future research to address 'ideal' interruptions should include analysis of how to support users to construct appropriate interruption rules, potentially from large amounts of contextual data. Several researchers exploring tailored DBCIs have found that users tend to limit their if-then rules to simple rules in small numbers, due to issues in understanding more complex sets of rules [Lee et al. 2017; Pinder et al. 2016c].

For step (d), as noted in Section 4.3.1, establishing the optimal schedules of reinforcement for rewards/punishments for a given response is also difficult.

6.2 Evaluation

Behaviour change interventions are difficult to evaluate [Intille 2004; Klasnja et al. 2011] for three main reasons: they need to be monitored in the long term, use an appropriate experiment design, and use an appropriate measure of behaviour.

6.2.1 Long-term monitoring. In the Design for Persistence design principle (Section 5.4), we noted that habits take time to change. Thus long-term evaluation is crucial for DBCIs that target habits. Lally et al.'s study 2010 of a once-a-day repeated behaviour showed that it took between 18-254 days to plateau into automaticity, with a median time of 66 days and substantial variation at an individual level. A study of exercise habitual behaviour found that to establish a new exercise habit, new gym-goers had to exercise at least four times per week for around six weeks [Kaushal and Rhodes 2015]. However, most HCI research methods do not lend themselves well to determining whether a given intervention has resulted in a *permanent* change in behaviour [Fogg and Hreha 2010; Klasnja et al. 2011], yet post-intervention monitoring is crucial to determine whether changes in habitual behaviour are stable. The gold standard is the sort of Randomised Controlled Trial used in medical research, but large-scale long-term trials are difficult to establish in HCI [Hekler et al. 2013; Klasnja et al. 2011]. Alongside this, there is the familiar issue of studies using small numbers of participants, no control groups and not reporting behaviour change results.

Empirical evidence demonstrates the difficulties: a meta-review of internet-based DBCIs rejected 26% of 549 possible studies because they did not use a measure of behaviour, and 15% because they did not use a control group [Webb et al. 2010a], while Froehlich et al. 2010 analysed 8 in-the-wild studies in the energy domain and found an average of 11 participants lasting 1-4 weeks, where no studies had a control group and only 4 reported behaviour change data. These findings are supported by Brynjarsdottir et al. 2012, who reviewed persuasive sustainability research and found almost half of the papers had no user evaluation, while field studies typically lasted 3-4 weeks with fewer than 10 participants; and Hamari et al. 2014, whose review of 95 DBCIs found that experiment length was "very short in most cases", and sample size median was 26 (where reported). On an individual study level, two highly cited trials in the health behaviour change domain, UbiFit⁹ [Consolvo et al. 2008b] and Fish'n'Steps¹⁰ [Lin et al. 2006] had small numbers of participants (12 and 19 respectively), although Fish'n'Steps was tested over 14 weeks (including pre- and postbaselines) to UbiFit's 3 weeks (no baseline).

The short-term nature of DBCI testing may also skew empirical support for theoretical constructs. Sheeran et al. provide evidence that initial experience of a new behaviour strengthens the intention-future behaviour link, while substantial experience attenuates the link due to habit formation [Sheeran et al. 2017a]. Thus short-term studies introducing participants to novel behaviours may provide evidence of a strong intention-behaviour link that does not persist over time. DBCIs that target habitual behaviour also face more specific issues of evaluation in monitoring and measuring users, which we outline below.

6.2.2 Experimental Design. There are several tensions in establishing the correct experiment design to explore habitual behaviour. The first tension, in line with evaluation issues, is in the length of study. As noted above, short term evaluations may miss key habitual changes. However, Kahneman suggests that short experiments with between-participants designs and little information about the purpose of the experiment are preferable to explore Type 1 processes [Kahneman 2003]. This is because longer designs with repeated measures "encourage the participants to search for consistent strategies to deal with the task" [Kahneman 2003].

⁹335 citations in the ACM Digital Library, 990 in Google Scholar on 22 March 2018

¹⁰247 citations in the ACM Digital Library, 723 in Google Scholar on on 22 March 2018

The second tension is that many experimental designs are inadequate to capture intra-individual variations in behaviour [Gardner 2015]. This is particularly key given evidence of individual variation in the relative impact of Type 1 vs Type 2 processes in determining behaviour [Sladek et al. 2006]. Appropriate study designs are those that: can use mixed-level modelling to capture withinperson variation [Jaeger 2008]; 'N-of-1' experimental designs, which can reveal how behaviours change over time [Klasnja et al. 2017; Lillie et al. 2011; McDonald et al. 2017]; or micro-randomized trials examining the impact of individual behaviour change components on proximal outcomes [Klasnja et al. 2017]. The latter is a variation on a sequential factorial design: participants are randomly assigned different interventions at times judged to hold potential for effective intervention [Klasnja et al. 2017]. It holds promise as a way forward in the pick-and-mix DBCI design context to judge the efficacy of different intervention elements.

6.2.3 Appropriate measure of habits.

Measuring behaviour. Directly measuring actual behaviour is the gold standard of a behaviour change DBCI, in line with a behaviourist standpoint, but is difficult to achieve in practice [Klasnja et al. 2011]. Few health game interventions are measured beyond changes in behavioural intention [Hwang and Mamykina 2017]. Even when behaviour is measured, e.g. [Cambo et al. 2017], it is important to also measure baseline pre-intervention and post-intervention behaviour to determine whether (a) behaviour has changed and (b) whether that change is sustained in the longer term as outlined above.

Self-report. Habit measurement is therefore dominated by self-report of behaviour [Gardner 2015]. The two main self-report measures for habits are Verplanken & Orbell's twelve-item Self-Report Habit Index (SRHI, 2003) and the Self-Report Behavioral Automaticity Index (SRBAI, [Gardner et al. 2012a]). The SRHI is designed to gauge habit strength and determine the role of habit without measuring behavioural frequency. The SRBAI is a subscale of the SRHI to measure automaticity more succinctly. Verplanken & Orbell 2003 found a strong correlation over four studies between the scores in the SRHI and habit frequency, and note that the use of behavioural frequency as a measure of habit is itself only a proxy for a true measure of habit strength.

Although these two measures have been used widely (e.g. SRHI [Gardner et al. 2011], SRBAI [Kaushal and Rhodes 2015]), there are concerns about their validity [Nilsen et al. 2012]. Since we define a habit as an impulse for a response triggered by a particular context, arising in Type 1 processes, self-report measures requiring Type 2 self-reflection on behaviour are insufficient. The indices measure the experience of habitual behaviour rather than the underlying processes, and a person's recollection of automaticity is unreliable as a habit measure [Hagger et al. 2015]. Further, the SRHI does not capture the impact of cues, and is limited to capturing the consequence of enacted habits [Sniehotta and Presseau 2012].

Nevertheless, self-report should not be disregarded entirely. There is some evidence that self-report can be helpful in triggering enhanced reflection around possible causes of problematic behaviour, particularly when emotions are self-reported [Hollis et al. 2015].

Self-report alternatives. Association tests, measuring response time speed and accuracy of contextresponse links, should be considered as measures of habit as alternatives to self-report [Aarts et al. 1998; Gardner 2015]. There is evidence that these techniques can provide better predictions of ongoing behaviour than those arising from self-reports [McCusker 2001]. There are a broad range of techniques of evaluating the Type 1 activity are appropriate for DBCIs [Wiers et al. 2013], each one providing an indirect measure of Type 1 activity by providing a reaction time measure to given stimuli. Three commonly used techniques, which may also be appropriate for DBCIs, are Stroop tests, Implicit Association Tests and Dot Probe Tests. Stroop tests (including emotional Stroop variants), which measure cognitive control and emotional attachment [Williams et al. 1996] are used to measure the relative control of Type 1 vs Type 2 processes in response to a particular cue: it is assumed that the more salient a cue is to Type 1 processes, the more difficult it is for Type 2 to override it, and therefore a longer reaction time. Implicit Association Tests [Greenwald et al. 1998] measure implicit evaluations of concepts and can therefore indicate where a particular action impulse might appear on the HAM's Potential Response stack (P6 in Figure 1). Dot Probe Tests [MacLeod et al. 1986] can both train and estimate automatic associations of cognitive constructs represented by words and/or images and can thus indicate implicit attentional bias at the Filter stage of the HAM (Figure 1).

However, there is ongoing debate about the validity of associative measures of Type 1 activity including habits [Hagger et al. 2015; Stacy and Wiers 2010]. Furthermore, researchers need to be careful of practice effects [Greenwald and Nosek 2001], implying the need for careful experiment design (e.g. using the latest scoring algorithms) and a control group.

Multiple measures. While it remains unclear which is the best method to measure habits, DBCIs should use multiple points of measurement. Measures should include both a measure of habitual behaviour and a measure of the habitual impulse. Until activity recognition algorithms are sufficient to capture behaviour, we suggest using the SRBAI or other validated self-report measure of the target behaviour, together with at least one associative measurement. Health psychologists are sceptical about performing associative tests in the wild [Gardner 2015; Hollands et al. 2016]. However, we suggest DBCIs have a unique advantage in being able to measure context-response associations in situ. Home-based association tests can be a better predictor of behaviour than lab-based tests [Houben and Wiers 2008]. More research is required into how best to translate lab-dependent desktop measurement techniques into DBCIs.

Not measuring behaviour. Jepson et al.'s meta-review 2006 of health behaviour change interventions identified an ongoing issue with conflation between knowledge, attitudes and behaviour. The issue also affects DBCI research. For example, a study of interactive shopping mannequins measured *perceived* length of stay at a shop window as an outcome, rather than measuring *actual* length of stay [Reitberger et al. 2009], while research into changing exercise behaviour using a heart rate monitor [Harjumaa et al. 2009] does not report the results of quantitative measures, instead reporting user attitudes and experiences. A focus on explicit attitudes as a determinant of behaviour also arises from a theoretical basis. For example, the use of the Theory of Planned behaviour implicitly accepts a link between attitudes and behaviour.

What to report. There is a lack of consensus on how to report DBCIs so that findings are generalisable. As with other behaviour change research, there is a dearth of common reporting standards [Webb et al. 2010b]. Multiple efforts are being made to deliver ontologies of behaviour change interventions to facilitate reporting (e.g. [Klasnja et al. 2017; Larsen et al. 2017; Michie et al. 2013a]. Once a consensus emerges, these should be followed. In the meantime, research must clearly report the intervention, including: theoretical basis; specific behaviour change technique(s); domain; behavioural context(s); mode of delivery (including specific technology); type of behaviour being targeted including the extent to which it is habitual; for HAM interventions, which process (Type 1 vs Type 2) is being targeted; together with transparent statistical reporting [Kay et al. 2016]. We also need more information about participants given the heterogeneity of intervention efficacy [Klasnja et al. 2017; Orji et al. 2017].

6.3 Ethics and privacy

We see much potential for DBCI technology to empower people to take control of their lives and be more proactive about their daily practices [Rogers 2006], particularly by helping them to break undesired habits and create desired habits. However, there are several ethical concerns that need to be considered in order to comply with the *design ethically* design principle (Section 5.7).

6.3.1 Who is in control? There is much potential power in enabling people to use Type 1-targeting DBCIs to alter their associative memories. However, there are also worrying opportunities for governments and corporations to covertly impose their own agendas on people via these mechanisms [Pinder 2017]. As noted in Section 5.7 above, controversy is particularly high over interventions that use subliminal priming techniques.

6.3.2 Full disclosure and safety. DBCI designers using nonconscious Type 1 methods (e.g. instinctive paths, subliminal priming) need to be explicit about how their intervention works. It is not clear how much disclosure is needed for ethical and safety reasons, and when it may start impacting the effectiveness of the intervention. For example, do we inform users that a smartphone subliminal priming intervention will flash 'activation words' on their display, or do we inform them of exactly which words are used and when, or even allow them to play back the prime in slow-motion? We argue that the benefits from an ethical perspective of informing users of how the DBCI works outweigh the drawbacks of it being less effective or ineffective. DBCIs that are only effective when people are (partially) unaware of how they work should be avoided.

A particular issue regarding disclosure is safety. Triggering fear-based flight responses as used in Zombies, Run! [zombiesrungame 2018] in inappropriate contexts (e.g. whilst crossing the road) is clearly problematic. Safety is also an issue when delivering disruptive habit interventions to activate Type 2 processes. Type 2 processes have limited capacity, so disrupting a user when they are already using Type 2 processes in another potentially dangerous task (e.g. driving) is problematic.

Who bears responsibility for the effects of tools to change behaviour is an open question: is the end-user or the system designer responsible [Verbeek 2009]? This issue is particularly apt for configurable systems: for example, what if a user with an eating disorder altered a cue-valencealtering system to devalue all foods instead of just unhealthy foods [Pinder 2017]?

6.3.3 Privacy. Alongside ethics, privacy is an issue with DBCIs, particularly where systems may be disclosing automatically-sensed information to third parties, e.g. for off-device computation. The fusion of real-world sensing with social networks also raises privacy concerns where users have limited control over how they are represented to their social network contacts [Efstratiou et al. 2012]. Again, system designers should be explicit about what data is shared and why, and allow users to opt out.

6.4 Cyberpsychology of interacting with DBCIs

There is a research gap in understanding the cyberpsychology [Norman 2008] of DBCIs that target habits. For example, how different are the reactions to interventions targeting Type 1 versus Type 2 processes? How do these reactions vary between individuals? How does the context in which the habit-targeting technology persuades (e.g. whether it is public or private) impact on efficacy [Mylonopoulou and Isomursu 2016]?

Pervasive technology gives us a great deal of power to support people to change their behaviours by targeting both Type 1 and Type processes, but we need to understand more about how these interventions are perceived. Two key considerations are agency (who is perceived to be doing the persuading) and reactance. 6.4.1 Agency. According to the "Computer As Social Actor" (CASA, [Reeves and Nass 1996]) paradigm, users tend to ascribe social characteristics to even minimal computer interfaces. If this holds true for DBCIs, then they can take advantage of extensive human-human persuasion research [Cialdini 2001], for example by leveraging social priming or contagion [Christakis and Fowler 2013]. However, it is not clear to what extent pervasive systems, particularly unobtrusive ones, may either exploit or avoid this social assumption.

6.4.2 Reactance. As noted in Section 5.6, reactance is a design issue for all DBCIs. It is an open research question to what extent and in what circumstances DBCIs targeting Type 1 processes can trigger reactance. Subliminal pervasive DBCI techniques may avoid reactance, but this has yet to be shown empirically. An unobtrusive DBCI targeting Type 1 processes may enable users to feel free from direct observation by 'official' sources and therefore reduce the likelihood of reactance. Users may therefore view unobtrusive DBCIs as potentially more acceptable and effective than externally-imposed solutions. A recent trend towards enabling DBCI participants to choose their own pick-and-mix interventions from a selection of behaviour change techniques [Lee et al. 2017; Meinlschmidt et al. 2016] may prove to be key in avoiding reactance if it makes users feel more in control of their DBCI.

6.5 Theoretical gaps in the HAM

All three of the HAM's underlying theories – Dual Process Theory, modern habit theory and Goal Setting theory –are active areas of research. Under Dual Process Theory, research is ongoing into determining which process, Type 1 or Type 2, predominates under what conditions [Sladek et al. 2006]. This contrasts with theory development in the health behaviour change domain, which has been dominated by expanding constructs rather than isolating the contexts in which those constructs have most impact on behaviour [Sheeran et al. 2017b]. As outlined in Section 2.7.2, modern habit theory needs to be subjected to more rigorous empirical testing and a more rigorous analysis of the mechanisms underlying formation and automaticity. Likewise, as noted in Section 2.7.3, Goal Setting Theory is being updated to incorporate Type 1 processes [Latham et al. 2017]. In terms of our HAM, further research is also required into effective combinations of the identified habit-changing techniques [Harjumaa et al. 2009], to establish the parameters of effectiveness of any given strategy [Michie and Johnston 2012; Peters et al. 2015], including determining the most effective technology platform on which to deliver it.

Where a behaviour targeted by a DBCI is indivisibly complex - for example some exercise behaviours that involve equipment and/or travel - several researchers have suggested that instigation habits should be distinguished from execution habits [Gardner 2015; Kaushal et al. 2017; Phillips and Gardner 2016].

Instigation habits are the generation of a behavioural impulse for a given behaviour from cues; execution habits are the carrying out of the sequence of sub-actions that make up the cued behaviour. Although our HAM accommodates this distinction (initiation occurs during the filter/prepare stage; execution during the act stage), further research is required to determine the appropriate interventions for each and their relative impacts on behaviour.

6.6 Summary

Despite our optimism that DBCIs targeting habits can achieve long-term behaviour change, several key challenges remain. Context is crucial, but highly accurate cue, prime and behaviour identification are arguably still unsolved ubicomp problems. This makes just-in-time interventions difficult to implement accurately. All DBCIs, but particularly inaccurate just-in-time interventions, risk both alienating users via reactance and inadvertently achieving ironic effects. Interventions face difficulties in selecting both an appropriate experimental design to test their efficacy, and in selecting an appropriate habit measurement mechanism.

As with any intervention to change behaviour, DBCI designers must consider the ethical implications of both what they promise and what they deliver. Ethics is also an issue where the cyberpsychology of habit-focused interventions is little-understood. Finally, although we have explicitly given our framework a solid theoretical grounding, theoretical gaps remain, particularly in establishing the boundaries of which set of processes predominates in which contexts.

7 FUTURE RESEARCH DIRECTIONS & OPPORTUNITIES

Table 5 shows the habit-altering strategies and the corresponding open research questions derived from the HAM, our overview of the strategies and our design principles. Note from Table 4 only one strategy, implementation intentions, currently has strong evidence to support its use in habit-changing interventions, and that all strategies bar the provision of information face the challenge of imprecise context detection (Section 6.1).

The empirical gap is partly due to the pick-and-mix nature of the research. Pick-and-mix is problematic where multiple strategies are used without using an experiment design that can differentiate their impact (see Section 6.2.2). However, pick-and-mix can be a good intervention where there is evidence that the individual elements are effective. In the broader behaviour change research sphere, multiple strategies tend to be more successful than single interventions [Michie et al. 2016]. The HAM suggests that the most effective approach is likely to be a combination of Type 1 and Type 2 strategies at multiple HAM levels, and research into the interplay of Type 1 and Type 2 processes as determinants of behaviour change is increasing in health psychology [Hollands et al. 2016]. Carels et al. 2014 recently employed a habit-focused weight loss intervention using multiple strategies including breaking unwanted habits, forming new healthy habits, increasing exposure to healthy cues and automatic goal priming. They found a medium to large effect of the intervention at a 6-month follow up compared to an alternative. We suggest a similar approach for DBCIs to identify multiple strategies to employ.

7.1 **Opportunities**

We see great value in continuing to research how to use context-aware pervasive technology to support highly individualised DBCIs that can deliver a mix of interventions. The focus of research should be on two elements: firstly to use DBCIs to find out which features of HAM are active at different points for different behaviours for a given individual; secondly to exploit this knowledge in DBCIs that allow that individual to pick-and-mix their own interventions.

7.1.1 Trigger hunters. Type 1 context-response associations are not easily available to introspection. Pervasive technology could play a key role in uncovering these associations to discover which contextual features (cues) act as triggers for a particular unwanted response. This may enable us to address the challenge of fluid causal influences affecting both Type 1 and Type 2 systems in the wild [Michie et al. 2013b]. Once uncovered, people can avoid, approach and/or retrain their trigger cues accordingly. We need technology with richer contextual awareness-including mood-detection-to identify both existing cue(s) that trigger unwanted behaviour and candidate cue(s) that can be used as anchors to associate with wanted behaviours within implementation intentions. Advances in context-cue detection and behaviour detection, perhaps driven by machine learning techniques [Banovic et al. 2017], will broaden our understanding of what sorts of cues act as response triggers for different types of behaviours, which can feed into our models of what sorts of cues should be used as habit triggers.

Strategy	Key research questions
General HAM application	What is the most effective combination of Type 1 and Type 2 strategies? How should DBCIs adapt as behaviour becomes more automatic? How best to use emotion as a behaviour change strategy?
Alter context	What advantages does ambient persuasive technology confer over priming on mobile device?
Priming	What is the best platform/format/schedule on which to deliver these primes? What are the instinctive primes for different contexts and differ- ent behaviours? How can we best instil and automate goals?
Alter cue salience Train context-response	What is the most effective training strategy for altering attention and approach biases via technology-mediated CBM: opportunis- tic training or session-based?
Implementation intentions	How can DBCIs best support implementation intention forma- tion? What is the most effective way for DBCIs to exploit their addi- tional rehearsal and reminder opportunities?
Provide information	What is the best method of delivering just-in-time reminders to avoid reactance and avoid overloading Type 2 cognitive limits? Who should set the goals, the DBCI or the user?
Self-control	What are the determinants of automating self-control?
Self-tracking	Which type of process does self-tracking operate on, and how can we automate it? How best to track cue triggers, including emotions?
Revalue outcome	How best to deliver rewards and reward schedules? Do virtual rewards work? Do implicit rewards work?

Table 5. Habit altering strategies and research questions

7.1.2 Self pick-and-mix. A key area of research is in enabling individuals to vary their interventions according to their preferences [Ijsselsteijn et al. 2006; Lee et al. 2017, 2011; Ranfelt et al. 2009], personality traits [Orji et al. 2017] and different digital devices. For example, DBCIs could enable people to use their own images of real-life problematic cues in CBM-Attention interventions. We foresee a crucial role for such DBCIs to help solve the fundamental variability in human behaviours, motivations and contexts [Rogers 2006]. The crucial design points for future DBCIs are that research must be theory-driven, and use experimental design and reporting standards that allow generalisability of results, as discussed in Section 6.2.2 and Section 6.2.3.

8 CONCLUSIONS

We have presented a set of strategic approaches which hold promise for designing DBCIs that alter habitual behaviour to generate sustainable behaviour change. A focus on habits is crucial to designing effective DBCIs for three main reasons. Firstly, habitual behaviour is common in everyday life in multiple domains. Secondly, we have shown that reasoned-action theories and corresponding Type 2 techniques alone are unable to achieve lasting behaviour change in the face of strong habits. Thirdly, we have identified multiple opportunities for pervasive computing technology to deliver interventions that can target both Type 1 and Type 2 processes.

We note that a lack of theory can impact on intervention efficacy and generalisability. To illustrate how three sets of theories (Dual Process Theory, modern habit theory and Goal Setting Theory) suggest habits may operate, we integrated them into a descriptive framework, the Habit Alteration Model. We outlined a set of possible interventions based on the model, and generated a set of design principles to guide these interventions. The model, in contrast to most DBCI theory and research, demonstrates how both Type 1 nonconscious and Type 2 conscious processes can contribute to stable behaviour change.

The key technical challenges for DBCIs remain accurate and fast detection of context, and the familiar problem of evaluation. Other challenges remain in addressing outstanding theoretical issues, in robust testing of the suggested strategies, in avoiding reactance and in determining the most effective combination of the techniques for any given domain and implementation technology.

In conclusion, we strongly recommend that DBCI researchers consider the role of habits in the behaviour they are targeting, and consider the implications of the HAM to design interventions that target Type 1 processes alongside Type 2 processes. We look forward to a rich era of DBCI habitual behaviour change research.

ACKNOWLEDGMENTS

The authors would like to thank the reviewers for their constructive and insightful comments. Thank you to Robert J Hendley, Dr Rowanne Fleck and Diego Garaialde for their helpful comments. Special thanks to Dr Lindsay MacDonald Vermeulen for her help with the graphical representation of the HAM.

REFERENCES

- Henk Aarts, Ruud Custers, and Hans Marien. 2008. Preparing and motivating behavior outside of awareness. *Science (New York, N.Y.)* 319, 5870 (March 2008), 1095–9203. https://doi.org/10.1126/science.1150432
- Henk Aarts and Ap Dijksterhuis. 2000. Habits as knowledge structures: automaticity in goal-directed behavior. Journal of Personality & Social Psychology 78, 1 (2000), 53. https://doi.org/10.1037/0022-3514.78.1.53
- Henk Aarts, Theo Paulussen, and Herman Schaalma. 1997. Physical exercise habit: on the conceptualization and formation of habitual health behaviours. *Health education research* 12, 3 (Sept. 1997), 363–74.
- Henk Aarts, Bas Verplanken, and Ad Knippenberg. 1998. Predicting Behavior From Actions in the Past: Repeated Decision Making or a Matter of Habit? *Journal of Applied Social Psychology* 28, 15 (Aug. 1998), 1355–1374. https://doi.org/10. 1111/j.1559-1816.1998.tb01681.x
- Charles Abraham and Susan Michie. 2008. A taxonomy of behavior change techniques used in interventions. *Health Psychology* 27, 3 (May 2008), 379–87. https://doi.org/10.1037/0278-6133.27.3.379
- Alexander T. Adams, Jean Costa, Malte F. Jung, and Tanzeem Choudhury. 2015. Mindless Computing: Designing Technologies to Subtly Influence Behavior. In *UbiComp* '15. ACM, New York, NY, USA, 719–730. https://doi.org/10.1145/2750858. 2805843
- Marc A. Adams, Jane C. Hurley, Michael Todd, Nishat Bhuiyan, Catherine L. Jarrett, Wesley J. Tucker, Kevin E. Hollingshead, and Siddhartha S. Angadi. 2017. Adaptive goal setting and financial incentives: a 2 AU 2 factorial randomized controlled trial to increase adultsâĂŹ physical activity. BMC Public Health 17, 1 (Dec. 2017). https: //doi.org/10.1186/s12889-017-4197-8
- Marc A Adams, Simon J Marshall, Lindsay Dillon, Susan Caparosa, Ernesto Ramirez, Justin Phillips, and Greg J Norman. 2009. A theory-based framework for evaluating exergames as persuasive technology. In *Persuasive '09.* ACM, New York,

NY, USA, 45:1-45:8. https://doi.org/10.1145/1541948.1542006

- Marieke A. Adriaanse, Floor M. Kroese, Marleen Gillebaart, and Denise T. D. De Ridder. 2014. Effortless inhibition: habit mediates the relation between self-control and unhealthy snack consumption. *Frontiers in Psychology* 5 (May 2014). https://doi.org/10.3389/fpsyg.2014.00444
- Nadav Aharony, Wei Pan, Cory Ip, Inas Khayal, and Alex Pentland. 2011. The social fMRI: Measuring, Understanding, and Designing Social Mechanisms in the Real World. In *Proceedings of the 13th International Conference on Ubiquitous Computing*. ACM, 445–454. http://web.media.mit.edu/nadav/misc/AharonyPhDThesisProposal.pdf
- Icek Ajzen. 1991. The theory of planned behavior. Organizational Behavior and Human Decision Processes 50, 2 (1991), 179-211.
- Judith Amores and Pattie Maes. 2017. Essence: Olfactory Interfaces for Unconscious Influence of Mood and Cognitive Performance. In CHI '17. ACM Press, 28–34. https://doi.org/10.1145/3025453.3026004
- Swamy Ananthanarayan and Katie A Siek. 2012. Persuasive wearable technology design for health and wellness. In *PervasiveHealth*. IEEE, 236–240.
- Cecilie Schou Andreassen. 2015. Online Social Network Site Addiction: A Comprehensive Review. Current Addiction Reports 2, 2 (2015), 175–184.
- Ashley E. Anker, Thomas Hugh Feeley, Bonnie McCracken, and Carolyn A. Lagoe. 2016. Measuring the Effectiveness of Mass-Mediated Health Campaigns Through Meta-Analysis. *Journal of Health Communication* 21, 4 (April 2016), 439–456. https://doi.org/10.1080/10810730.2015.1095820
- Sinan Aral and Christos Nicolaides. 2017. Exercise contagion in a global social network. *Nature Communications* 8 (April 2017), 14753. https://doi.org/10.1038/ncomms14753
- C J Armitage and M Conner. 2000. Social cognition models and health behaviour: A structured review. *Psychology and Health* 15, 2 (2000), 173–189.
- A. Arzi, Y. Holtzman, P. Samnon, N. Eshel, E. Harel, and N. Sobel. 2014. Olfactory Aversive Conditioning during Sleep Reduces Cigarette-Smoking Behavior. *Journal of Neuroscience* 34, 46 (Nov. 2014), 15382–15393. https://doi.org/10.1523/ JNEUROSCI.2291-14.2014
- Paul Aveyard, Louise Massey, Amanda Parsons, Semira Manaseki, and Carl Griffin. 2009. The effect of Transtheoretical Model based interventions on smoking cessation. *Social Science & Medicine* 68, 3 (Feb. 2009), 397–403. https://doi.org/10. 1016/j.socscimed.2008.10.036
- Albert Bandura. 1978. The self system in reciprocal determinism. *American psychologist* 33, 4 (1978), 344–358. https://doi.org/10.1037//0003-066x.33.4.344
- Albert Bandura. 1986. Social foundations of thought and action: a social cognitive theory. Prentice-Hall, Englewood Cliffs, N.J.
- Albert Bandura. 1998. Health promotion from the perspective of social cognitive theory. *Psychology and health* 13, 4 (1998), 623–649.
- Albert Bandura. 2001. Social cognitive theory: An agentic perspective. Annual review of psychology 52 (2001), 1–26.
- Albert Bandura. 2011. Social cognitive theory. Handbook of social psychological theories (2011), 349-373.
- Nikola Banovic, Tofi Buzali, Fanny Chevalier, Jennifer Mankoff, and Anind K. Dey. 2016. Modeling and Understanding Human Routine Behavior. In *Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems*. ACM, 248–260.
- Nikola Banovic, Anqi Wang, Yanfeng Jin, Christie Chang, Julian Ramos, Anind Dey, and Jennifer Mankoff. 2017. Leveraging Human Routine Models to Detect and Generate Human Behaviors. ACM Press, 6683–6694. https://doi.org/10.1145/ 3025453.3025571
- John A Bargh, Annette Lee-Chai, Kimberly Barndollar, Peter Gollwitzer, and Roman TrÄutschel. 2001. The Automated Will: Nonconscious Activation and Pursuit of Behavioral Goals. *Journal of Personality and Social Psychology* 81, 6 (2001), 1014–1027.
- John A. Bargh and Ezequiel Morsella. 2010. Unconscious behavioral guidance systems. In *Then a miracle occurs: Focusing on behavior in social psychological theory and research*, Christopher R. Agnew, Donal E. Carlston, William G. Graziano, and Janice R. Kelly (Eds.). Oxford University Press, 89–118. http://www.yale.edu/acmelab/articles/BarghMorsella_UBGS_final.pdf
- Melissa Bateson, Daniel Nettle, and Gilbert Roberts. 2006. Cues of being watched enhance cooperation in a real-world setting. *Biology Letters* 2, 3 (2006), 412–414. https://doi.org/10.1098/rsbl.2006.0509
- Christine Bauer and Florian Waldner. 2013. Reactive music: when user behavior affects sounds in real-time. In CHI '13 Extended Abstracts on Human Factors in Computing Systems. ACM, Paris, France, 739–744.
- Roy Baumeister. 2002. Yielding to Temptation: Self-Control Failure, Impulsive Purchasing, and Consumer Behavior. Journal of Consumer Research 28, 4 (2002), pp. 670–676.
- Roy F. Baumeister, Kathleen D. Vohs, C. Nathan DeWall, and Liqing Zhang. 2007. How Emotion Shapes Behavior: Feedback, Anticipation, and Reflection, Rather Than Direct Causation. *Personality and Social Psychology Review* 11, 2 (May 2007), 167–203. https://doi.org/10.1177/1088868307301033

- Joseph B. Bayer and Scott W. Campbell. 2012. Texting while driving on automatic: Considering the frequency-independent side of habit. *Computers in Human Behavior* 28, 6 (2012), 2083–2090.
- F. Bellisle, A.M. Dalix, and G. Slama. 2004. Non food-related environmental stimuli induce increased meal intake in healthy women: comparison of television viewing versus listening to a recorded story in laboratory settings. *Appetite* 43, 2 (Oct. 2004), 175–180. https://doi.org/10.1016/j.appet.2004.04.004
- Claudio Bettini, Oliver Brdiczka, Karen Henricksen, Jadwiga Indulska, Daniela Nicklas, Anand Ranganathan, and Daniele Riboni. 2010. A survey of context modelling and reasoning techniques. *Pervasive and Mobile Computing* 6, 2 (2010), 161–180.
- Agon Bexheti, Anton Fedosov, Jesper Findahl, Marc Langheinrich, and Evangelos Niforatos. 2015. Re-Live the Moment: Visualizing Run Experiences to Motivate Future Exercises. In *MobileHCI '15 Adjunct*. ACM, New York, NY, USA, 986–993. https://doi.org/10.1145/2786567.2794316
- Sidney W Bijou. 1957. Patterns of reinforcement and resistance to extinction in young children. *Child Development* (1957), 47–54.
- Tegan Blackburne, Alexandra Rodriguez, and Stuart John Johnstone. 2016. A Serious Game to Increase Healthy Food Consumption in Overweight or Obese Adults: Randomized Controlled Trial. *JMIR Serious Games* 4, 2 (2016), e10. https://doi.org/10.2196/games.5708
- Ron Borland. 2013. Understanding hard to maintain behaviour change: a dual process approach. John Wiley & Sons.
- Ron Borland. 2016. CEOS Theory: A Comprehensive Approach to Understanding Hard to Maintain Behaviour Change. Applied Psychology: Health and Well-Being (Dec. 2016). https://doi.org/10.1111/aphw.12083
- Mark E Bouton. 2014. Why behavior change is difficult to sustain. *Preventive medicine* (June 2014). https://doi.org/10.1016/j.ypmed.2014.06.010
- Jack W. Brehm. 2009. A theory of psychological reactance. In *Organization Change: A Comprehensive Reader*, W Burke, Dale G Lake, and J W Paine (Eds.). John Wiley & Sons, Ltd, San Francisco, CA, 377–390.
- Damien Brevers, Anne Rogiers, Alexis Defontaine, Guy Cheron, Anne-Marie Clarinval, Jennifer Foucart, Anne Bouchez, Véronique Bolly, Laura Tsartsafloudakis, Pénélope Jottrand, Pierre Minner, Antoine Bechara, Charles Kornreich, and Paul Verbanck. 2017. Implementation Intention for Initiating Intuitive Eating and Active Embodiment in Obese Patients Using a Smartphone Application. *Frontiers in Psychiatry* 8 (2017), 243. https://doi.org/10.3389/fpsyt.2017.00243
- Marijn Bruin, Paschal Sheeran, Gerjo Kok, Anneke Hiemstra, Jan M. Prins, Harm J. Hospers, and Gerard JP Breukelen. 2012. Self-regulatory processes mediate the intention-behavior relation for adherence and exercise behaviors. *Health Psychology* 31, 6 (2012), 695.
- Thomas A. Brunner and Michael Siegrist. 2012. Reduced food intake after exposure to subtle weight-related cues. *Appetite* 58, 3 (June 2012), 1109–1112. https://doi.org/10.1016/j.appet.2012.03.010
- Hronn Brynjarsdottir, Maria Håkansson, James Pierce, Eric Baumer, Carl DiSalvo, and Phoebe Sengers. 2012. Sustainably unpersuaded: how persuasion narrows our vision of sustainability. In *CHI '12*. ACM, 947–956.
- Michelle Nicole Burns, Mark Begale, Jennifer Duffecy, Darren Gergle, Chris J Karr, Emily Giangrande, and David C Mohr. 2011. Harnessing Context Sensing to Develop a Mobile Intervention for Depression. *Journal of Medical Internet Research* 13, 3 (Aug. 2011), e55. https://doi.org/10.2196/jmir.1838
- Meghan L. Butryn, Suzanne Phelan, James O. Hill, and Rena R. Wing. 2007. Consistent self-monitoring of weight: a key component of successful weight loss maintenance. *Obesity* 15, 12 (2007), 3091–3096. https://doi.org/10.1038/oby.2007.368
- Scott A. Cambo, Daniel Avrahami, and Matthew L. Lee. 2017. BreakSense: Combining Physiological and Location Sensing to Promote Mobility during Work-Breaks. In CHI '17. ACM Press, 3595–3607. https://doi.org/10.1145/3025453.3026021
- Rémi L. Capa, Gaëlle M. Bustin, Axel Cleeremans, and Michel Hansenne. 2011. Conscious and Unconscious Reward Cues Can Affect a Critical Component of Executive Control: (Un)conscious Updating? *Experimental Psychology* 58, 5 (July 2011), 370–375. https://doi.org/10.1027/1618-3169/a000104
- Ana Caraban, Evangelos Karapanos, Vítor Teixeira, Sean A. Munson, and Pedro Campos. 2017. On the Design of Subly: Instilling Behavior Change During Web Surfing Through Subliminal Priming. In *PERSUASIVE '17*, Peter W. Vries, Harri Oinas-Kukkonen, Liseth Siemons, Nienke Beerlage-de Jong, and Lisette Gemert-Pijnen (Eds.), Vol. 10171. Springer International Publishing, 163–174. https://doi.org/10.1007/978-3-319-55134-0_13
- Robert A. Carels, Jacob M. Burmeister, Afton M. Koball, Marissa W. Oehlhof, Nova Hinman, Michelle LeRoy, Erin Bannon, Lee Ashrafioun, Amy Storfer-Isser, and Lynn A. Darby. 2014. A randomized trial comparing two approaches to weight loss: Differences in weight loss maintenance. *Journal of health psychology* 19, 2 (2014), 296–311.
- Tanya L. Chartrand and John Bargh. 1996. Automatic activation of impression formation and memorization goals: Nonconscious goal priming reproduces effects of explicit task instructions. *Journal of Personality and Social Psychology* 71, 3 (1996), 464–478. https://doi.org/10.1037//0022-3514.71.3.464
- Frank X. Chen, Abby C. King, and Eric B. Hekler. 2014. "Healthifying" Exergames: Improving Health Outcomes Through Intentional Priming. In CHI '14 (CHI '14). ACM, New York, NY, USA, 1855–1864. https://doi.org/10.1145/2556288.2557246

- M. Chini, A. De Angeli, S. Gabrielli, R. Maimone, and V. Osmani. 2012. StandFit : Fighting Sedentary Work Life. In *Designing for Wellness and Behavior Change workshop*, NordiCHI 2012. http://www.kotiposti.net/xvalpas/dfwabc/pdf/dwbc2012_submission 7.pdf
- Eun Kyoung Choe, Nicole B Lee, Bongshin Lee, Wanda Pratt, and Julie A Kientz. 2014. Understanding quantified-selfers' practices in collecting and exploring personal data. In CHI '14. ACM, Toronto, Canada, 1143–1152.
- Nicholas A. Christakis and James H. Fowler. 2013. Social contagion theory: examining dynamic social networks and human behavior. *Statistics in medicine* 32, 4 (2013), 556–577.
- Robert B. Cialdini. 2001. Influence: Science and practice (fourth ed.). Allyn & Bacon.
- Franceli L. Cibrian, Monica Tentori, and Ana I. Martínez-García. 2016. Hunting Relics: A Persuasive Exergame to Promote Collective Exercise in Young Children. International Journal of Human-Computer Interaction 32, 3 (March 2016), 277–294. https://doi.org/10.1080/10447318.2016.1136180
- Bronwyn K. Clark, Takemi Sugiyama, Genevieve N. Healy, Jo Salmon, David W. Dunstan, and Neville Owen. 2009. Validity and reliability of measures of television viewing time and other non-occupational sedentary behaviour of adults: a review. *Obesity reviews* 10, 1 (2009), 7–16.
- N. Clark and Mary R. Janevic. 2014. Individual theories. In *The handbook for health behavior change*, K. A. Riekert, J. K. Ockene, and L. Pbert (Eds.). 3–26.
- Lisa H. Colbert and Dale A. Schoeller. 2011. Expending our physical activity (measurement) budget wisely. Journal of applied physiology 111, 2 (2011), 606–607.
- Rob Comber and Anja Thieme. 2013. Designing Beyond Habit: Opening Space for Improved Recycling and Food Waste Behaviors Through Processes of Persuasion, Social Influence and Aversive Affect. *Personal Ubiquitous Computing* 17, 6 (Aug. 2013), 1197–1210. https://doi.org/10.1007/s00779-012-0587-1
- David E Conroy, Jaclyn P Maher, Steriani Elavsky, Amanda L Hyde, and Shawna E Doerksen. 2013. Sedentary behavior as a daily process regulated by habits and intentions. *Health psychology : official journal of the Division of Health Psychology, American Psychological Association* 32, 11 (Nov. 2013), 1149–57. https://doi.org/10.1037/a0031629
- Sunny Consolvo, Predrag Klasnja, David W. McDonald, Daniel Avrahami, Jon Froehlich, Louis LeGrand, Ryan Libby, Keith Mosher, and James A. Landay. 2008a. Flowers or a robot army?: encouraging awareness & activity with personal, mobile displays. In UbiComp '08. ACM, Seoul, South Korea, 54–63.
- Sunny Consolvo, Predrag Klasnja, David W McDonald, and James A Landay. 2009a. Goal-setting considerations for persuasive technologies that encourage physical activity. In *Persuasive '09*. ACM, Claremont , CA, USA, 8:1–8:8. http: //doi.acm.org/10.1145/1541948.1541960
- Sunny Consolvo, David W McDonald, and James A Landay. 2009b. Theory-driven design strategies for technologies that support behavior change in everyday life. In *CHI '09*. ACM, Boston, USA, 405–414. https://doi.org/10.1145/1518701. 1518766
- Sunny Consolvo, David W McDonald, Tammy Toscos, Mike Y Chen, Jon Froehlich, Beverly Harrison, Predrag Klasnja, Anthony LaMarca, Louis LeGrand, Ryan Libby, and others. 2008b. Activity sensing in the wild: a field trial of ubifit garden. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*. ACM, 1797–1806.
- Benjamin R Cowan, Chris P Bowers, Russell Beale, and Charlie Pinder. 2013. The Stroppy Kettle: An Intervention to Break Energy Consumption Habits. In CHI '13 Extended Abstracts. ACM, Paris, France. https://doi.org/10.1145/2468356.2468621
- Logan T. Cowan, Sarah A. Van Wagenen, Brittany A. Brown, Riley J. Hedin, Yukiko Seino-Stephan, P. Cougar Hall, and Joshua H. West. 2012. Apps of steel: are exercise apps providing consumers with realistic expectations? A content analysis of exercise apps for presence of behavior change theory. *Health Education & Behavior* (2012).
- Anna L. Cox, Sandy J.J. Gould, Marta E. Cecchinato, Ioanna Iacovides, and Ian Renfree. 2016. Design Frictions for Mindful Interactions: The Case for Microboundaries. In CHI'16 Extended Abstracts. ACM Press, 1389–1397. https://doi.org/10. 1145/2851581.2892410
- Jo Cranwell, Steve Benford, Robert J. Houghton, Michael Golembewski, Joel E. Fischer, and Martin S. Hagger. 2014. Increasing self-regulatory energy using an Internet-based training application delivered by smartphone technology. *Cyberpsychology, Behavior, and Social Networking* 17, 3 (2014), 181–186.
- Karen W. Cullen, Yan Liu, and Debbe I. Thompson. 2016. Meal-specific dietary changes from Squires Quest! II: a serious video game intervention. *Journal of nutrition education and behavior* 48, 5 (2016), 326–330. e1.
- Ruud Custers and Henk Aarts. 2007. In search of the nonconscious sources of goal pursuit: Accessibility and positive affective valence of the goal state. *Journal of Experimental Social Psychology* 43, 2 (2007), 312–318. https://doi.org/10. 1016/j.jesp.2006.02.005
- Unna N. Danner, Henk Aarts, Esther K. Papies, and Nanne K. Vries. 2011. Paving the path for habit change: Cognitive shielding of intentions against habit intrusion. *British Journal of Health Psychology* 16, 1 (2011), 189–200.
- Unna N. Danner, Henk Aarts, and Nanne K. Vries. 2008. Habit vs. intention in the prediction of future behaviour: The role of frequency, context stability and mental accessibility of past behaviour. *British Journal of Social Psychology* 47, 2 (June 2008), 245–265. https://doi.org/10.1348/014466607X230876

- Saskia Dantzig, Gijs Geleijnse, and Aart Tijmen Halteren. 2012. Toward a persuasive mobile application to reduce sedentary behavior. *Personal and Ubiquitous Computing* (July 2012). https://doi.org/10.1007/s00779-012-0588-0
- Nediyana Daskalova, Karthik Desingh, Alexandra Papoutsaki, Diane Schulze, Han Sha, and Jeff Huang. 2017. Lessons Learned from Two Cohorts of Personal Informatics Self-Experiments. Proceedings of the ACM on Interactive, Mobile, Wearable and Ubiquitous Technologies 1, 3 (Sept. 2017), 1–22. https://doi.org/10.1145/3130911
- R.A. Davis. 2001. A cognitive-behavioral model of pathological Internet use. *Computers in Human Behavior* 17, 2 (March 2001), 187–195. https://doi.org/10.1016/S0747-5632(00)00041-8
- Rachel Davis, Rona Campbell, Zoe Hildon, Lorna Hobbs, and Susan Michie. 2015. Theories of behaviour and behaviour change across the social and behavioural sciences: a scoping review. *Health psychology review* 9, 3 (2015), 323–344.
- Tracy A. Dennis and Laura J. O'Toole. 2014. Mental Health on the Go: Effects of a Gamified Attention-Bias Modification Mobile Application in Trait-Anxious Adults. 2, 5 (2014), 576–590. https://doi.org/10.1177/2167702614522228
- Roland Deutsch and Fritz Strack. 2010. Building blocks of social behavior: reflective and impulsive processes. In *Handbook of implicit social cognition: Measurement, theory, and applications*, Bertram Gawronski and Keith Payne (Eds.). The Guilford Press, New York, NY, USA, 62–79.
- Xiang Ding, Jing Xu, Honghao Wang, Guanling Chen, Herpreet Thind, and Yuan Zhang. 2016. WalkMore: Promoting Walking with Just-in-Time Context-Aware Prompts. Bethesda, MD.
- Alan Dix. 2002. Beyond intention-pushing boundaries with incidental interaction. In Proceedings of Building Bridges: Interdisciplinary Context-Sensitive Computing, Glasgow University, Vol. 9.
- Paul Dolan, Michael Hallsworth, David Halpern, Dominic King, Robert Metcalfe, and Ivo Vlaev. 2012. Influencing behaviour: The mindspace way. *Journal of Economic Psychology* 33, 1 (2012), 264–277.
- Amol Doshi, Kevin Patrick, James F. Sallis, and Karen Calfas. 2003. Evaluation of physical activity web sites for use of behavior change theories. Annals of Behavioral Medicine 25, 2 (April 2003), 105–111. https://doi.org/10.1207/S15324796ABM2502_06
- Joel R. Drake, Ryan Cain, and Victor R. Lee. 2017. From Wearing to Wondering. In *Optimizing STEM Education With* Advanced ICTs and Simulations. IGI Global, 1–29. https://doi.org/10.4018/978-1-5225-2528-8.ch001
- B. D. Earp, B. Dill, J. L. Harris, J. M. Ackerman, and J. A. Bargh. 2013. No sign of quitting: Incidental exposure to no-smoking signs ironically boosts cigarette-approach tendencies in smokers. *Journal of Applied Social Psychology* 43, 10 (2013), 2158–2162.
- M. Economides, M. Guitart-Masip, Z. Kurth-Nelson, and R.J. Dolan. 2015. Arbitration between controlled and impulsive choices. NeuroImage 109 (April 2015), 206–216. https://doi.org/10.1016/j.neuroimage.2014.12.071
- Christos Efstratiou, Ilias Leontiadis, Marco Picone, Kiran K Rachuri, Cecilia Mascolo, and Jon Crowcroft. 2012. Sense and sensibility in a pervasive world. In *Pervasive Computing*. Springer, 406–424.
- Gilles O. Einstein and Mark A. McDaniel. 2005. Prospective memory multiple retrieval processes. *Current Directions in Psychological Science* 14, 6 (2005), 286–290.
- Philip M. Enock, Stefan G. Hofmann, and Richard J. McNally. 2014. Attention bias modification training via smartphone to reduce social anxiety: a randomized, controlled multi-session experiment. *Cognitive therapy and research* 38, 2 (2014), 200–216.
- Daniel A. Epstein, Monica Caraway, Chuck Johnston, An Ping, James Fogarty, and Sean A. Munson. 2016. Beyond Abandonment to Next Steps: Understanding and Designing for Life after Personal Informatics Tool Use. ACM Press, 1109–1113. https://doi.org/10.1145/2858036.2858045
- Ido Erev and Daniel Gopher. 1999. Attention and Performance XVII: Cognitive Regulation of Performance: Interaction of Theory and Applications. Cambridge, MA: MIT Press, 343–371.
- Jonathan Evans and Keith Frankish. 2009. In two minds: Dual processes and beyond. Oxford University Press, Oxford. http://oro.open.ac.uk/22096/
- Jonathan St BT Evans. 2011. Dual-process theories of reasoning: Contemporary issues and developmental applications. *Developmental Review* 31, 2 (2011), 86–102. https://doi.org/10.1016/j.dr.2011.07.007
- Jonathan St B T Evans. 2008. Dual-processing accounts of reasoning, judgment, and social cognition. Annu. Rev. Psychol. 59 (2008), 255–278.
- Stephen H. Fairclough. 2009. Fundamentals of physiological computing. Interacting with computers 21, 1 (2009), 133-145.
- Susan Shepherd Ferebee. 2010. Successful Persuasive Technology for Behavior Reduction: Mapping to Fogg's Gray Behavior Grid. In *Persuasive '10*. Copenhagen, Denmark, 70–81.
- Leon Festinger. 1957. A theory of cognitive dissonance. Vol. 2. Stanford University Press.

Fitbit. 2018. Fitbit homepage. (2018). http://www.fitbit.com/uk

- Maddalena Fiordelli, Nicola Diviani, and Peter J Schulz. 2013. Mapping mHealth research: a decade of evolution. *Journal of medical Internet research* 15, 5 (2013).
- Ayelet Fishbach and James Y. Shah. 2006. Self-Control in Action: Implicit Dispositions Toward Goals and Away From Temptations. *Journal of Personality and Social Psychology* 90, 5 (2006), 820–832. https://doi.org/10.1037/0022-3514.90.5.820

:55

- Thomas H. B. FitzGerald, Raymond J. Dolan, and Karl J. Friston. 2014. Model averaging, optimal inference, and habit formation. *Frontiers in Human Neuroscience* 8 (June 2014). https://doi.org/10.3389/fnhum.2014.00457
- BJ Fogg. 2015. Tiny Habits. (2015). http://tinyhabits.com/
- Brian J. Fogg. 2002. Persuasive Technology: Using Computers to Change What We Think and Do. Morgan Kaufmann Publishers, San Francisco, CA, USA.
- Brian J. Fogg. 2009a. The Behavior Grid: 35 Ways Behavior Can Change. In *Persuasive '09*. ACM, New York, NY, USA, 42:1–42:5. https://doi.org/10.1145/1541948.1542001
- Brian J. Fogg. 2009b. A behavior model for persuasive design. In *Persuasive '09*. ACM, 40. http://www.bjfogg.com/fbm_files/page4_1.pdf
- Brian J. Fogg and Jason Hreha. 2010. Behavior Wizard: A Method for Matching Target Behaviors with Solutions. In *Persuasive* '10. Copenhagen, Denmark, 117–131.
- Jens Förster, Nira Liberman, and Ronald S Friedman. 2007. Seven principles of goal activation: a systematic approach to distinguishing goal priming from priming of non-goal constructs. *Personality and social psychology review* 11, 3 (Aug. 2007), 211–33. https://doi.org/10.1177/1088868307303029
- Derek Foster, Conor Linehan, Shaun Lawson, and Ben Kirman. 2011. Power ballads: deploying aversive energy feedback in social media. In CHI'11 EA (CHI EA '11). ACM, New York, NY, USA, 2221–2226. https://doi.org/10.1145/1979742.1979944
- Elaine Fox, Anna Ridgewell, and Chris Ashwin. 2009. Looking on the bright side: biased attention and the human serotonin transporter gene. *Proceedings of the Royal Society of London B: Biological Sciences* 276, 1663 (2009), 1747–1751.
- Nicholas Freeman and Mark Muraven. 2010. DonâĂŹt interrupt me! Task interruption depletes the selfâĂŹs limited resources. *Motivation and Emotion* 34, 3 (2010), 230–241.
- Thomas Fritz, Elaine M. Huang, Gail C. Murphy, and Thomas Zimmermann. 2014. Persuasive technology in the real world: a study of long-term use of activity sensing devices for fitness. In *CHI '14*. ACM Press, New York, New York, USA, 487–496. https://doi.org/10.1145/2556288.2557383
- Jon Froehlich, Leah Findlater, and James Landay. 2010. The design of eco-feedback technology. In CHI '10. ACM, 1999-2008.
- Benjamin Gardner. 2015. A review and analysis of the use of âĂŸhabitâĂŹ in understanding, predicting and influencing health-related behaviour. *Health Psychology Review* 9, 3 (2015), 277–295.
- Benjamin Gardner, Charles Abraham, Phillippa Lally, and Gert-Jan Bruijn. 2012a. Towards parsimony in habit measurement: Testing the convergent and predictive validity of an automaticity subscale of the Self-Report Habit Index. *International Journal of Behavioral Nutrition and Physical Activity* 9, 1 (2012), 102. https://doi.org/10.1186/1479-5868-9-102
- Benjamin Gardner, Gert-Jan Bruijn, and Phillippa Lally. 2011. A systematic review and meta-analysis of applications of the Self-Report Habit Index to nutrition and physical activity behaviours. *Annals of Behavioral Medicine* 42, 2 (Oct. 2011), 174–87. https://doi.org/10.1007/s12160-011-9282-0
- Benjamin Gardner, Phillippa Lally, and Jane Wardle. 2012b. Making health habitual: the psychology of 'habit-formation' and general practice. The British Journal of General Practice 62, 605 (Dec. 2012), 664–6. https://doi.org/10.3399/bjgp12X659466
- Antonella Gasbarri, Assunta Pompili, Mark G Packard, and Carlos Tomaz. 2014. Habit learning and memory in mammals: Behavioral and neural characteristics. *Neurobiology of learning and memory* 114 (June 2014), 198–208. https://doi.org/10. 1016/j.nlm.2014.06.010
- Bertram Gawronski and Laura A Creighton. 2006. Dual-Process Theories. In *The Oxford handbook of social cognition*, D. E. Carlston (Ed.). Oxford University Press, New York, NY, USA, 1–35.
- Gregory Gimpel, Frantisek Sudzina, and Katarina Petrovcikova. 2016. Mobile ICT use in early adopter vs. late majority countries. *International Journal of Mobile Communications* 14, 6 (2016), 610. https://doi.org/10.1504/IJMC.2016.079305
- Jack Glaser and John F. Kihlstrom. 2005. Compensatory automaticity: Unconscious volition is not an oxymoron. In *The new unconscious*, Hassin, James S. Uleman, and John A Bargh (Eds.). Oxford University Press, 171–195. http://dx.doi.org/10.1093/acprof:oso/9780195307696.003.0008
- Peter M. Gollwitzer, Ute C. Bayer, and Kathleen C. McCulloch. 2005. The control of the unwanted. *The new unconscious* (2005), 485–515.
- Peter M. Gollwitzer and Paschal Sheeran. 2006. Implementation intentions and goal achievement: A meta-analysis of effects and processes. *Advances in experimental social psychology* 38 (2006), 69–119. https://doi.org/10.1016/s0065-2601(06) 38002-1
- Victoria A. Goodyear, Charlotte Kerner, and Mikael Quennerstedt. 2017. Young peopleäÄŹs uses of wearable healthy lifestyle technologies; surveillance, self-surveillance and resistance. Sport, Education and Society (2017), 1–14.
- RĂžben Gouveia, Evangelos Karapanos, and Marc Hassenzahl. 2015. How Do We Engage With Activity Trackers? A Longitudinal Study of Habito. In UbiComp âĂŹ15.
- Ann M Graybiel. 2008. Habits, rituals, and the evaluative brain. 31 (2008), 359–87. https://doi.org/10.1146/annurev.neuro. 29.051605.112851
- Anthony G. Greenwald, Debbie E. McGhee, and Jordan LK Schwartz. 1998. Measuring individual differences in implicit cognition: the implicit association test. *Journal of personality and social psychology* 74, 6 (1998), 1464.

- Anthony G. Greenwald and Brian A. Nosek. 2001. Health of the Implicit Association Test at age 3. *Experimental Psychology* 48, 2 (2001), 85–93.
- Martin S. Hagger and Aleksandra Luszczynska. 2014. Implementation intention and action planning interventions in health contexts: State of the research and proposals for the way forward. *Applied Psychology: Health and Well-Being* 6, 1 (2014), 1–47.
- Martin S. Hagger, Amanda L. Rebar, Barbara Mullan, Ottmar V. Lipp, and Nikos L.D. Chatzisarantis. 2015. The subjective experience of habit captured by self-report indexes may lead to inaccuracies in the measurement of habitual action. *Health Psychology Review* 9, 3 (Aug. 2015), 296–302. https://doi.org/10.1080/17437199.2014.959728
- Jaap Ham, Cees Midden, and Femke Beute. 2009. Can ambient persuasive technology persuade unconsciously?: using subliminal feedback to influence energy consumption ratings of household appliances. In *Persuasive '09*. ACM, New York, NY, USA, 29:1–29:6. https://doi.org/10.1145/1541948.1541988
- Jaap Ham and Cees J H Midden. 2010. Ambient Persuasive Technology Needs Little Cognitive Effort: The Differential Effects of Cognitive Load on Lighting Feedback versus Factual Feedback. In *Persuasive '10*. Copenhagen, Denmark, 132–142. https://doi.org/10.1007/978-3-642-13226-1_14
- Juho Hamari, Jonna Koivisto, and Tuomas Pakkanen. 2014. Do persuasive technologies persuade?-a review of empirical studies. In *International Conference on Persuasive Technology*. Springer, 118–136.
- Wendy Hardeman, Marie Johnston, Derek Johnston, Debbie Bonetti, Nicholas Wareham, and Ann Louise Kinmonth. 2002. Application of the Theory of Planned Behaviour in Behaviour Change Interventions: A Systematic Review. *Psychology & Health* 17, 2 (Jan. 2002), 123–158. https://doi.org/10.1080/08870440290013644a
- Marja Harjumaa, Katarina Segerståhl, and Harri Oinas-Kukkonen. 2009. Understanding persuasive software functionality in practice: a field trial of polar FT60. In *Persuasive '09.* ACM, New York, NY, USA, 2:1–2:9. http://doi.acm.org/10.1145/1541948.1541952
- Benjamin Harkin, Thomas L. Webb, Betty P. I. Chang, Andrew Prestwich, Mark Conner, Ian Kellar, Yael Benn, and Paschal Sheeran. 2016. Does monitoring goal progress promote goal attainment? A meta-analysis of the experimental evidence. *Psychological Bulletin* 142, 2 (2016), 198–229. https://doi.org/10.1037/bul0000025
- Jody Harris, Lambert Felix, Alec Miners, Elizabeth Murray, Susan Michie, Elaine L. Ferguson, Caroline Free, Karen Lock, Jane Landon, and Phil Edwards. 2011. Adaptive e-learning to improve dietary behaviour: a systematic review and cost-effectiveness analysis. *Health Technol Assessment* 15, 37 (2011). http://www.hta.ac.uk/1811
- Hanna Hasselqvist, Mia Hesselgren, and Cristian Bogdan. 2016. Challenging the Car Norm: Opportunities for ICT to Support Sustainable Transportation Practices. ACM Press, 1300–1311. https://doi.org/10.1145/2858036.2858468
- Janika Heitmann, Madelon E. van Hemel-Ruiter, Karin M. Vermeulen, Brian D. Ostafin, Colin MacLeod, Reinout W. Wiers, Laura DeFuentes-Merillas, Martine Fledderus, Wiebren Markus, and Peter J. de Jong. 2017. Internet-based attentional bias modification training as add-on to regular treatment in alcohol and cannabis dependent outpatients: a study protocol of a randomized control trial. 17, 1 (2017). https://doi.org/10.1186/s12888-017-1359-2
- Eric B. Hekler, Predrag Klasnja, Jon E. Froehlich, and Matthew P. Buman. 2013. Mind the Theoretical Gap: Interpreting, Using, and Developing Behavioral Theory in HCI Research. In *CHI '13*. ACM, New York, NY, USA, 3307–3316. https: //doi.org/10.1145/2470654.2466452
- Sander Hermsen, Jeana Frost, Reint Jan Renes, and Peter Kerkhof. 2016. Using feedback through digital technology to disrupt and change habitual behavior: A critical review of current literature. *Computers in Human Behavior* 57 (2016), 61–74. https://doi.org/10.1016/j.chb.2015.12.023
- Javier Hernandez, Daniel J. McDuff, and Rosalind W. Picard. 2015. Biophone: Physiology monitoring from peripheral smartphone motions. In *Engineering in Medicine and Biology Society (EMBC), 2015 37th Annual International Conference of the IEEE*. IEEE, 7180–7183.
- Paula T Hertel and Andrew Mathews. 2011. Cognitive Bias Modification: Past Perspectives, Current Findings, and Future Applications. Perspectives on Psychological Science 6, 6 (Oct. 2011), 521–536. https://doi.org/10.1177/1745691611421205
- Melvyn Hillsdon, Margaret Thorogood, Ian White, and Charlie Foster. 2002. Advising people to take more exercise is ineffective: a randomized controlled trial of physical activity promotion in primary care. *International Journal of Epidemiology* 31, 4 (Aug. 2002), 808–815. https://doi.org/10.1093/ije/31.4.808
- Sen H. Hirano, Robert G. Farrell, Catalina M. Danis, and Wendy A. Kellogg. 2013. WalkMinder: encouraging an active lifestyle using mobile phone interruptions. In CHI'13 Extended Abstracts. ACM, 1431–1436.
- Wilhelm Hofmann, Malte Friese, and Fritz Strack. 2009. Impulse and self-control from a dual-systems perspective. Perspectives on Psychological Science 4, 2 (2009), 162–176.
- Wilhelm Hofmann, Malte Friese, and Reinout W. Wiers. 2008. Impulsive versus reflective influences on health behavior: a theoretical framework and empirical review. *Health Psychology Review* 2, 2 (Sept. 2008), 111–137. https://doi.org/10. 1080/17437190802617668
- Gareth J. Hollands, Theresa M. Marteau, and Paul C. Fletcher. 2016. Non-conscious processes in changing health-related behaviour: a conceptual analysis and framework. *Health Psychology Review* (Feb. 2016), 1–14. https://doi.org/10.1080/

17437199.2015.1138093

- Gareth J. Hollands, Ian Shemilt, Theresa M. Marteau, Susan A. Jebb, Michael P. Kelly, Ryota Nakamura, Marc Suhrcke, and David Ogilvie. 2013. Altering micro-environments to change population health behaviour: towards an evidence base for choice architecture interventions. *BMC Public Health* 13, 1 (2013), 1–6.
- John R Hollenbeck, Charles R Williams, and Howard J Klein. 1989. An Empirical Examination of the Antecedents of Commitment to Difficult Goals. *Journal of Applied Psychology* 74, 1 (1989), 18–23. https://doi.org/10.1037/0021-9010.74.1.18
- Victoria Hollis, Artie Konrad, and Steve Whittaker. 2015. Change of Heart: Emotion Tracking to Promote Behavior Change. In CHI '15 (CHI '15). ACM, New York, NY, USA, 2643–2652. https://doi.org/10.1145/2702123.2702196
- Anita Honka, Kirsikka Kaipainen, Henri Hietala, and Niilo Saranummi. 2011. Rethinking health: ICT-enabled services to empower people to manage their health. *Biomedical Engineering, IEEE Reviews in* 4 (2011), 119–139.
- Katrijn Houben and Reinout W. Wiers. 2008. Measuring implicit alcohol associations via the Internet: Validation of Web-based implicit association tests. *Behavior Research Methods* 40, 4 (Nov. 2008), 1134–1143. https://doi.org/10.3758/BRM.40.4.1134
- Neil Howlett, Joerg Schulz, Daksha Trivedi, Nicholas Troop, and Angel Chater. 2017. A prospective study exploring the construct and predictive validity of the COM-B model for physical activity. *Journal of Health Psychology* (nov 2017). https://doi.org/10.1177/1359105317739098 arXiv:https://doi.org/10.1177/1359105317739098 PMID: 29172808.
- Maria L. Hwang and Lena Mamykina. 2017. Monster Appetite: Effects of Subversive Framing on Nutritional Choices in a Digital Game Environment. In *CHI'17*. ACM Press, 4082–4096. https://doi.org/10.1145/3025453.3026052
- Wijnand Ijsselsteijn, Yvonne Kort, Cees Midden, Berry Eggen, and Elise Den Hoven. 2006. Persuasive technology for human well-being: Setting the scene. In *Persuasive '06.* 1–5. http://www.springerlink.com/index/kr12485156689gw2.pdf
- Stephen S Intille. 2004. A New Research Challenge : Persuasive Technology to Motivate Healthy Aging. *Technology* 8, 3 (2004), 235–237.
- Ipsos. 2017. Ipsos MediaCT Tech Tracker Q4 2017. (2017). https://www.ipsos.com/sites/default/files/ct/publication/ documents/2018-01/ipsos_connect_techtracker_q4_2017.pdf
- Tim Jackson. 2005. Motivating sustainable consumption. Sustainable Development Research Network 29 (2005), 30.
- T. Florian Jaeger. 2008. Categorical data analysis: Away from ANOVAs (transformation or not) and towards logit mixed models. *Journal of memory and language* 59, 4 (2008), 434–446. https://doi.org/10.1016/j.jml.2007.11.007
- John M. Jakicic, Kelliann K. Davis, Renee J. Rogers, Wendy C. King, Marsha D. Marcus, Diane Helsel, Amy D. Rickman, Abdus S. Wahed, and Steven H. Belle. 2016. Effect of Wearable Technology Combined With a Lifestyle Intervention on Long-term Weight Loss: The IDEA Randomized Clinical Trial. JAMA 316, 11 (Sept. 2016), 1161. https://doi.org/10.1001/ jama.2016.12858
- Arne Jansen, Maarten Van Mechelen, and Karin Slegers. 2017. Personas and Behavioral Theories: A Case Study Using Self-Determination Theory to Construct Overweight Personas. ACM Press, 2127–2136. https://doi.org/10.1145/3025453. 3026003
- Jawbone. 2018. Jawbone homepage. (2018). https://jawbone.com/
- Ruth Jepson, Steve Platt, and Jennifer Cox. 2006. A review of the effectiveness of interventions, approaches and models at individual, community and population level that are aimed at changing health outcomes through changing knowledge, attitudes and behaviour. Technical Report May. Cancer Care Research Centre, University of Stirling; Alliance for Self Care Research, University of Abertay.
- Mindy F. Ji and Wendy Wood. 2007. Purchase and consumption habits: Not necessarily what you intend. *Journal of Consumer Psychology* 17, 4 (2007), 261–276.
- Yuan Jia, Bin Xu, Yamini Karanam, and Stephen Voida. 2016. Personality-targeted Gamification: A Survey Study on Personality Traits and Motivational Affordances. ACM Press, 2001–2013. https://doi.org/10.1145/2858036.2858515
- Gaby Judah, Benjamin Gardner, and Robert Aunger. 2012. Forming a flossing habit: An exploratory study of the psychological determinants of habit formation. *British Journal of Health Psychology* 18, 2 (Sept. 2012). https://doi.org/10.1111/j.2044-8287. 2012.02086.x
- D Kahneman. 2003. A perspective on judgment and choice: mapping bounded rationality. *American psychologist* 58, 9 (2003), 697.
- Daniel Kahneman. 2011. Thinking, Fast and SLow. Macmillan.
- Daniel Kahneman and Amos Tversky. 2000. Choices, values, and frames. Cambridge University Press.
- Naomi Kakoschke, Eva Kemps, and Marika Tiggemann. 2014. Attentional bias modification encourages healthy eating. Eating behaviors 15, 1 (2014), 120–124. https://doi.org/10.1016/j.eatbeh.2013.11.001
- Vaiva Kalnikaite, Yvonne Rogers, Jon Bird, Nicolas Villar, Khaled Bachour, Stephen Payne, Peter M Todd, Johannes SchÄűning, Antonio KrÄijger, and Stefan Kreitmayer. 2011. How to nudge in Situ: designing lambent devices to deliver salient information in supermarkets. In UbiComp '11. ACM, 11–20.
- Ashish Kapoor and Eric Horvitz. 2008. Experience sampling for building predictive user models: a comparative study. In *CHI '08*. ACM, 657–666.

- Navin Kaushal and Ryan E. Rhodes. 2015. Exercise habit formation in new gym members: a longitudinal study. Journal of Behavioral Medicine 38, 4 (Aug. 2015), 652–663. https://doi.org/10.1007/s10865-015-9640-7
- Navin Kaushal, Ryan E. Rhodes, John T. Meldrum, and John C. Spence. 2017. The role of habit in different phases of exercise. British Journal of Health Psychology (April 2017). https://doi.org/10.1111/bjhp.12237
- Matthew Kay, Steve Haroz, Shion Guha, and Pierre Dragicevic. 2016. Special Interest Group on Transparent Statistics in HCI. ACM Press, 1081–1084. https://doi.org/10.1145/2851581.2886442
- David Keatley, David. D. Clarke, and Martin S. Hagger. 2013. The predictive validity of implicit measures of self-determined motivation across health-related behaviours:. *British Journal of Health Psychology* 18, 1 (Feb. 2013), 2–17. https://doi.org/10.1111/j.2044-8287.2011.02063.x
- Ralph L. Keeney. 2008. Personal Decisions Are the Leading Cause of Death. Operations Research 56, 6 (Dec. 2008), 1335–1347. https://doi.org/10.1287/opre.1080.0588
- Christina Kelley, Bongshin Lee, and Lauren Wilcox. 2017. Self-tracking for Mental Wellness: Understanding Expert Perspectives and Student Experiences. In *CHI'17*. ACM, 629–641. https://doi.org/10.1145/3025453.3025750
- Klaus Kessler, Katja Biermann-Ruben, Melanie Jonas, Hartwig Roman Siebner, Tobias Bäumer, Alexander MÃijnchau, and Alfons Schnitzler. 2006. Investigating the human mirror neuron system by means of cortical synchronization during the imitation of biological movements. *NeuroImage* 33, 1 (Oct. 2006), 227–38. https://doi.org/10.1016/j.neuroimage.2006.06.014
- Danish Khan, Swamy Ananthanarayan, An Le, Christopher Schaefbauer, and Katie Siek. 2012. Designing Mobile Snack Application for Low Socioeconomic Status Families. In *Pervasive Health*. IEEE, 57–64. https://doi.org/10.4108/icst. pervasivehealth.2012.248692
- Rohit Ashok Khot, Jeewon Lee, Deepti Aggarwal, Larissa Hjorth, and Florian 'Floyd' Mueller. 2015. TastyBeats: Designing Palatable Representations of Physical Activity. In *CHI'15.* ACM, New York, NY, USA, 2933–2942. https://doi.org/10.1145/2702123.2702197
- John F Kihlstrom, Jennifer Dorfman, and Lillian Park. 2007. Implicit and explicit memory and learning. In *The Blackwell companion to consciousness*. Blackwell Publishing, Malden, MA.
- Jaejeung Kim, Joonyoung Park, and Uichin Lee. 2016. EcoMeal: A Smart Tray for Promoting Healthy Dietary Habits. In CHI '16 EA. ACM Press, 2165–2170. https://doi.org/10.1145/2851581.2892310
- Ben Kirman, Conor Linehan, Shaun Lawson, Derek Foster, and Mark Doughty. 2010. There's a monster in my kitchen: using aversive feedback to motivate behaviour change. In *CHI '10 EA*. ACM, New York, NY, USA, 2685–2694. https: //doi.org/10.1145/1753846.1753852
- Predrag Klasnja, Sunny Consolvo, and Wanda Pratt. 2011. How to Evaluate Technologies for Health Behavior Change in HCI Research. In *CHI* '11. 3063–3072. https://doi.org/10.1145/1978942.1979396
- Predrag Klasnja, Eric B. Hekler, Elizabeth V. Korinek, John Harlow, and Sonali R. Mishra. 2017. Toward Usable Evidence: Optimizing Knowledge Accumulation in HCI Research on Health Behavior Change. In *CHI '17*. ACM Press, 3071–3082. https://doi.org/10.1145/3025453.3026013
- Predrag Klasnja, Eric B. Hekler, Saul Shiffman, Audrey Boruvka, Daniel Almirall, Ambuj Tewari, and Susan A. Murphy. 2015. Microrandomized trials: An experimental design for developing just-in-time adaptive interventions. *Health Psychology* 34 (2015), 1220–1228. Issue Suppl. https://doi.org/10.1037/hea0000305
- Predrag Klasnja and Wanda Pratt. 2012. Healthcare in the pocket: mapping the space of mobile-phone health interventions. *Journal of biomedical informatics* 45, 1 (2012), 184–198.
- Christian A. Klöckner. 2013. A comprehensive model of the psychology of environmental behaviour A meta-analysis. *Global Environmental Change* 23, 5 (Oct. 2013), 1028–1038. https://doi.org/10.1016/j.gloenvcha.2013.05.014
- Artie Konrad, Victoria Bellotti, Nicole Crenshaw, Simon Tucker, Les Nelson, Honglu Du, Peter Pirolli, and Steve Whittaker. 2015. Finding the Adaptive Sweet Spot: Balancing Compliance and Achievement in Automated Stress Reduction. In *CHI*'15. ACM, New York, NY, USA, 3829–3838. https://doi.org/10.1145/2702123.2702512
- Stef PJ Kremers, Gert-Jan Bruijn, Tommy LS Visscher, Willem Mechelen, Nanne K. Vries, and Johannes Brug. 2006. Environmental influences on energy balance-related behaviors: A dual-process view. International Journal of Behavioral Nutrition and Physical Activity 3, 1 (2006), 9.
- Stef PJ Kremers, Frank F Eves, and Ross E Andersen. 2012. Environmental changes to promote physical activity and healthy dietary behavior. *Journal of environmental and public health* 2012 (2012).
- Pei-Yi (Patricia) Kuo and Michael S. Horn. 2017. Daily challenges for sustainable lifestyles: design implications from a mobile intervention study. ACM Press, 635–641. https://doi.org/10.1145/3123024.3124425
- Stacey Kuznetsov and Eric Paulos. 2010. UpStream: motivating water conservation with low-cost water flow sensing and persuasive displays. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems. ACM, 1851–1860.
- Phillippa Lally, Alison Chipperfield, and Jane Wardle. 2008. Healthy habits: efficacy of simple advice on weight control based on a habit-formation model. *International journal of obesity* 32, 4 (2008), 700–707.
- Phillippa Lally and Benjamin Gardner. 2013. Promoting habit formation. Health Psychology Review 7 (sup 1) (2013), S137–S158. https://doi.org/10.1080/17437199.2011.603640

- Phillippa Lally, Cornelia H. M. Jaarsveld, Henry W. W. Potts, and Jane Wardle. 2010. How are habits formed: Modelling habit formation in the real world. *European Journal of Social Psychology* 40, 6 (Oct. 2010), 998–1009. https://doi.org/10. 1002/ejsp.674
- Phillippa Lally, Jane Wardle, and Benjamin Gardner. 2011. Experiences of habit formation: A qualitative study. *Psychology, health & medicine* 16, 4 (2011), 484–489.
- Kai R. Larsen, Susan Michie, Eric B. Hekler, Bryan Gibson, Donna Spruijt-Metz, David Ahern, Heather Cole-Lewis, Rebecca J. Bartlett Ellis, Bradford Hesse, Richard P. Moser, and Jean Yi. 2017. Behavior change interventions: the potential of ontologies for advancing science and practice. *Journal of Behavioral Medicine* 40, 1 (Feb. 2017), 6–22. https: //doi.org/10.1007/s10865-016-9768-0
- Matthias Laschke, Marc Hassenzahl, Sarah Diefenbach, and Marius Tippkämper. 2011. With a little help from a friend: a shower calendar to save water. In CHI '11 Extended Abstracts. ACM, 633–646.
- Gary P. Latham, Jelena Brcic, and Alana Steinhauer. 2017. Toward an Integration of Goal Setting Theory and the Automaticity
- Model: Goal difficulty, effort and choice. Applied Psychology 66, 1 (Jan. 2017), 25–48. https://doi.org/10.1111/apps.12087
 Neal Lathia, Veljko Pejovic, Kiran K Rachuri, Cecilia Mascolo, Mirco Musolesi, and Peter J Rentfrow. 2013. Smartphones for large-scale behavior change interventions. *IEEE Pervasive Computing* 3 (2013), 66–73.
- Natalia S. Lawrence, Jamie O'Sullivan, David Parslow, Mahmood Javaid, Rachel C. Adams, Christopher D. Chambers, Katarina Kos, and Frederick Verbruggen. 2015. Training response inhibition to food is associated with weight loss and reduced energy intake. *Appetite* 95 (Dec. 2015), 17–28. https://doi.org/10.1016/j.appet.2015.06.009
- Amanda Lazar, Christian Koehler, Joshua Tanenbaum, and David H. Nguyen. 2015. Why We Use and Abandon Smart Devices. In UbiComp '15 Adjunct. Japan. https://doi.org/10.1145/2750858.2804288
- Dan Ledger and D. McCaffrey. 2014. Inside wearables: how the science of human behavior change offers the secret to long-term engagement. Technical Report. Endeavour Partners. http://endeavourpartners.net/assets/ Endeavour-Partners-Wearables-White-Paper-20141.pdf
- Jisoo Lee, Erin Walker, Winslow Burleson, Matthew Kay, Matthew Buman, and Eric B. Hekler. 2017. Self-Experimentation for Behavior Change: Design and Formative Evaluation of Two Approaches. In *CHI '17*. ACM Press, 6837–6849. https: //doi.org/10.1145/3025453.3026038
- MK Lee, Sara Kiesler, and Jodi Forlizzi. 2011. Mining behavioral economics to design persuasive technology for healthy choices. In *Proceedings of the SIGCHI Conference on \ldots*. 325–334. http://dl.acm.org/citation.cfm?id=1978989
- Maria R. Lee, Irena Bojanova, and Tom Suder. 2015. The New Wearable Computing Frontier. *IT Professional* 17, 5 (2015), 16–19.
- Elizabeth O Lillie, Bradley Patay, Joel Diamant, Brian Issell, Eric J Topol, and Nicholas J Schork. 2011. The n-of-1 clinical trial: the ultimate strategy for individualizing medicine? *Personalized Medicine* 8, 2 (March 2011), 161–173. https://doi.org/10.2217/pme.11.7
- Moez Limayem, Sabine Gabriele Hirt, and Christy MK Cheung. 2007. How habit limits the predictive power of intention: The case of information systems continuance. *MIS quarterly* (2007), 705–737.
- James J Lin, Lena Mamykina, Silvia Lindtner, Gregory Delajoux, and Henry B Strub. 2006. FishâĂŹnâĂŹSteps: Encouraging physical activity with an interactive computer game. In *UbiComp 2006*. Springer, 261–278.
- Mu Lin, Nicholas D. Lane, Mashfiqui Mohammod, Xiaochao Yang, Hong Lu, Giuseppe Cardone, Shahid Ali, Afsaneh Doryab, Ethan Berke, and Andrew T. Campbell. 2012. BeWell+: multi-dimensional wellbeing monitoring with community-guided user feedback and energy optimization. In *Proceedings of the conference on Wireless Health*. ACM, 10.
- Cameron Lister, Joshua H West, Ben Cannon, Tyler Sax, and David Brodegard. 2014. Just a Fad? Gamification in Health and Fitness Apps. JMIR Serious Games 2, 2 (Aug. 2014), e9. https://doi.org/10.2196/games.3413
- Edwin A. Locke and Gary P. Latham. 1990. A theory of goal setting & task performance. Prentice-Hall, Inc.
- Edwin A. Locke and Gary P. Latham. 2002. Building a practically useful theory of goal setting and task motivation: A 35-year odyssey. American Psychologist 57, 9 (2002), 705. https://doi.org/10.1037/0003-066x.57.9.705
- Edwin A. Locke and Gary P. Latham. 2006. New Directions in Goal-Setting Theory. Current Directions in Psychological Science 15, 5 (Oct. 2006), 265–268. https://doi.org/10.1111/j.1467-8721.2006.00449.x
- Dan Lockton, David Harrison, Tim Holley, and Neville A Stanton. 2009. Influencing interaction: development of the design with intent method. In *Persuasive '09*. ACM, 5:1–5:8. http://doi.acm.org/10.1145/1541948.1541956
- Dan Lockton, David Harrison, and Neville Anthony Stanton. 2010. Design with intent: 101 patterns for influencing behaviour through design. Equifine.
- J. Derek Lomas, Kenneth Koedinger, Nirmal Patel, Sharan Shodhan, Nikhil Poonwala, and Jodi L. Forlizzi. 2017. Is Difficulty Overrated?: The Effects of Choice, Novelty and Suspense on Intrinsic Motivation in Educational Games. ACM Press, 1028–1039. https://doi.org/10.1145/3025453.3025638
- Paul Lukowicz, Sandy Pentland, and Alois Ferscha. 2012. From Context awareness to socially aware Computing. Pervasive Computing, IEEE 11, 1 (2012), 32–41.

- Judith Lumley, Catherine Chamberlain, Therese Dowswell, Sandy Oliver, Laura Oakley, and Lyndsey Watson. 2009. Interventions for promoting smoking cessation during pregnancy. *Cochrane Database Syst Rev* 3 (2009).
- Colin MacLeod, Ernst H W Koster, Elaine Fox, and Others. 2009. Whither cognitive bias modification research? Commentary on the special section articles. *Journal of abnormal psychology* 118, 1 (2009), 89.
- Colin MacLeod, Andrew Mathews, and Philip Tata. 1986. Attentional bias in emotional disorders. *Journal of abnormal psychology* 95, 1 (1986), 15.
- Christian Magaraggia, James Dimmock, and Ben Jackson. 2014. Motivational priming as a strategy for maximising exercise outcomes: effects on exercise goals and engagement. 32, 9 (2014), 826-835. https://doi.org/10.1080/02640414.2013.862841
- Tiago V. Maia. 2009. Reinforcement learning, conditioning, and the brain: Successes and challenges. *Cognitive, Affective, & Behavioral Neuroscience* 9, 4 (Dec. 2009), 343–364. https://doi.org/10.3758/CABN.9.4.343
- Simon Marshall and Stuart Biddle. 2001. The transtheoretical model of behavior change: a meta-analysis of applications to physical activity and exercise. *Annals of Behavioral Medicine* 23, 4 (2001), 229–246.
- Theresa M Marteau, Gareth J Hollands, and Paul C Fletcher. 2012. Changing human behavior to prevent disease: the importance of targeting automatic processes. *Science (New York, N.Y.)* 337, 6101 (Sept. 2012), 1492–5. https://doi.org/10. 1126/science.1226918
- Cesar A. Martin, Daniel E. Rivera, William T. Riley, Eric B. Hekler, Matthew P. Buman, Marc A. Adams, and Abby C. King. 2014. A dynamical systems model of social cognitive theory. In *American Control Conference (ACC), 2014.* IEEE, 2407–2412.
- Christopher G McCusker. 2001. Cognitive biases and addiction: an evolution in theory and method. *Addiction* 96, 1 (2001), 47–56.
- Faye McDonald. 2014. Developing an Integrated Conceptual Framework of Pro-Environmental Behavior in the Workplace through Synthesis of the Current Literature. Administrative Sciences 4, 3 (Aug. 2014), 276–303. https://doi.org/10.3390/ admsci4030276
- Suzanne McDonald, Francis Quinn, Rute Vieira, Nicola O'Brien, Martin White, Derek W. Johnston, and Falko F. Sniehotta. 2017. The state of the art and future opportunities for using longitudinal n-of-1 methods in health behaviour research: a systematic literature overview. *Health Psychology Review* (April 2017), 1–17. https://doi.org/10.1080/17437199.2017. 1316672
- Abhinav Mehrotra, Veljko Pejovic, Jo Vermeulen, Robert Hendley, and Mirco Musolesi. 2016. My Phone and Me: Understanding PeopleäÄŹs Receptivity to Mobile Notifications. In 2016 ACM International Conference on Human Factors in Computing Systems.
- Abhinav Mehrotra, Jo Vermeulen, Veljko Pejovic, and Mirco Musolesi. 2015. Ask, but Don't Interrupt: The Case for Interruptibility-aware Mobile Experience Sampling. In *UbiComp âĂŹ15*. ACM, New York, NY, USA, 723–732. https://doi.org/10.1145/2800835.2804397
- Gunther Meinlschmidt, Jong-Hwan Lee, Esther Stalujanis, Angelo Belardi, Minkyung Oh, Eun Kyung Jung, Hyun-Chul Kim, Janine Alfano, Seung-Schik Yoo, and Marion Tegethoff. 2016. Smartphone-Based Psychotherapeutic Micro-Interventions to Improve Mood in a Real-World Setting. *Frontiers in Psychology* 7 (2016). https://doi.org/10.3389/fpsyg.2016.01112
- Janet Metcalfe and Walter Mischel. 1999. A hot/cool-system analysis of delay of gratification: dynamics of willpower. *Psychological review* 106, 1 (1999), 3.
- Jochen Meyer, Steven Simske, Katie A. Siek, Cathal G. Gurrin, and Hermie Hermens. 2014. Beyond quantified self: data for wellbeing. ACM Press, 95–98. https://doi.org/10.1145/2559206.2560469
- Susan Michie, Lou Atkins, and Robert West. 2014a. *The behaviour change wheel: a guide to designing interventions*. Silverback Publishing.
- Susan Michie, Rachel N. Carey, Marie Johnston, Alexander J. Rothman, Marijn Bruin, Michael P. Kelly, and Lauren E. Connell. 2016. From Theory-Inspired to Theory-Based Interventions: A Protocol for Developing and Testing a Methodology for Linking Behaviour Change Techniques to Theoretical Mechanisms of Action. *Annals of Behavioral Medicine* (July 2016). https://doi.org/10.1007/s12160-016-9816-6
- Susan Michie and Marie Johnston. 2012. Theories and techniques of behaviour change: Developing a cumulative science of behaviour change. *Health Psychology Review* 6, 1 (2012), 1–6.
- Susan Michie and Andrew Prestwich. 2010. Are interventions theory-based? Development of a theory coding scheme. *Health Psychology* 29, 1 (Jan. 2010), 1–8. https://doi.org/10.1037/a0016939
- Susan Michie, Michelle Richardson, Marie Johnston, Charles Abraham, Jill Francis, Wendy Hardeman, Martin P. Eccles, James Cane, and Caroline E. Wood. 2013a. The behavior change technique taxonomy (v1) of 93 hierarchically clustered techniques: building an international consensus for the reporting of behavior change interventions. *Annals of behavioral medicine* 46, 1 (2013), 81–95.
- Susan Michie, Maartje M Stralen, and Robert West. 2011. The behaviour change wheel: A new method for characterising and designing behaviour change interventions. *Implementation science : IS* 6, 1 (Jan. 2011), 42. https://doi.org/10.1186/ 1748-5908-6-42

- Susan Michie, Robert West, Rona Campbell, Jamie Brown, and Heather Gainforth. 2014b. *ABC of behaviour change theories*. Silverback Publishing, London.
- Susan Michie, Robert West, and Bonnie Spring. 2013b. Moving from theory to practice and back in social and health psychology. *Health Psychology* 32, 5 (2013), 581–585.
- Eleanor Miles, Paschal Sheeran, Harriet Baird, Ian Macdonald, Thomas L. Webb, and Peter R. Harris. 2016. Does self-control improve with practice? Evidence from a six-week training program. *Journal of Experimental Psychology: General* 145, 8 (2016), 1075–1091. https://doi.org/10.1037/xge0000185
- Arlen C. Moller, Gina Merchant, David E. Conroy, Robert West, Eric Hekler, Kari C. Kugler, and Susan Michie. 2017. Applying and advancing behavior change theories and techniques in the context of a digital health revolution: proposals for more effectively realizing untapped potential. *Journal of Behavioral Medicine* 40, 1 (Feb. 2017), 85–98. https: //doi.org/10.1007/s10865-016-9818-7
- Gregory D. Moody and Mikko Siponen. 2013. Using the theory of interpersonal behavior to explain non-work-related personal use of the Internet at work. *Information & Management* 50, 6 (Sept. 2013), 322–335. https://doi.org/10.1016/j.im.2013.04.005 Moov. 2018. Moov Fitness Coach: Guided Exercise & HIIT Workout. (2018). https://welcome.moov.cc/
- Sean A. Munson and Sunny Consolvo. 2012. Exploring goal-setting, rewards, self-monitoring, and sharing to motivate physical activity. In *PervasiveHealth*. IEEE, 25–32.
- Mark Muraven. 2010. Practicing self-control lowers the risk of smoking lapse. Psychology of Addictive Behaviors 24, 3 (Sept. 2010), 446–52. https://doi.org/10.1037/a0018545
- Mark Muraven and Roy F Baumeister. 2000. Self-regulation and depletion of limited resources: Does self-control resemble a muscle? *Psychological Bulletin* 126, 2 (2000), 247.
- Muse. 2018. MUSE homepage. (2018). http://www.choosemuse.com/
- mybasis.com. 2015. Basis. (2015). https://www.mybasis.com
- Vasiliki Mylonopoulou and Minna Isomursu. 2016. Context of use and timing of social comparison techniques in behavior change support: a qualitative systematic review. ACM Press, 249–259. https://doi.org/10.1145/3012709.3012734
- Felix Naughton, Neal Lathia, Barbara Brafman-Price, Cecilia Mascolo, Andrew McEwen, and Stephen Sutton. 2014. The Development of a Context Aware Smoking Cessation app (QSense). European Health Psychologist 16, S (2014), 855.
- David T. Neal, Wendy Wood, and Aimee Drolet. 2013. How do people adhere to goals when willpower is low? The profits (and pitfalls) of strong habits. *Journal of Personality and Social Psychology* 104, 6 (2013), 959.
- David T Neal, Wendy Wood, Jennifer S Labrecque, and Phillippa Lally. 2012. How do habits guide behavior? Perceived and actual triggers of habits in daily life. *Journal of Experimental Social Psychology* 48, 2 (2012), 492–498. https://doi.org/10.1016/j.jesp.2011.10.011
- David T Neal, Wendy Wood, and Jeremy M Quinn. 2006. HabitsâĂŤA repeat performance. Current Directions in Psychological Science 15, 4 (2006), 198–202. https://doi.org/10.1111/j.1467-8721.2006.00435.x
- Clayton Neighbors, Kristen P. Lindgren, Lindsey Rodriguez, Judy Tidwell, and Ivori Zvorsky. 2013. Cognitive factors in addictive processes. *Principles of addiction: Comprehensive addictive behaviors and disorders* 1 (2013), 259–269.
- Per Nilsen. 2015. Making sense of implementation theories, models and frameworks. *Implementation Science* 10, 1 (Dec. 2015). https://doi.org/10.1186/s13012-015-0242-0
- Per Nilsen, Kerstin Roback, Anders BrostrÄüm, and Per-Erik EllstrÄüm. 2012. Creatures of habit: accounting for the role of habit in implementation research on clinical behaviour change. *Implement Sci* 7, 53 (2012), 1–6. https://doi.org/10.1186/ 1748-5908-7-53
- Shahriar Nirjon, Robert F. Dickerson, Qiang Li, Philip Asare, John A. Stankovic, Dezhi Hong, Ben Zhang, Xiaofan Jiang, Guobin Shen, and Feng Zhao. 2012. Musicalheart: A hearty way of listening to music. In *SenSys '12*. ACM, 43–56.
- Seth M. Noar, Melissa Chabot, and Rick S. Zimmerman. 2008. Applying health behavior theory to multiple behavior change: Considerations and approaches. *Preventive Medicine* 46, 3 (March 2008), 275–280. https://doi.org/10.1016/j.ypmed.2007. 08.001
- Donald A. Norman and Tim Shallice. 1986. Attention to action. Springer.
- Marianna Obrist, Rob Comber, Sriram Subramanian, Betina Piqueras-Fiszman, Carlos Velasco, and Charles Spence. 2014. Temporal, affective, and embodied characteristics of taste experiences: A framework for design. In *CHI*'14. ACM, 2853–2862.
- Ellinor K. Olander, Helen Fletcher, Stefanie Williams, Lou Atkinson, Andrew Turner, and David P. French. 2013. What are the most effective techniques in changing obese individualsâĂŹ physical activity self-efficacy and behaviour: a systematic review and meta-analysis. *Int J Behav Nutr Phys Act* 10, 29 (2013), 1–15.
- Nuria Oliver and Fernando Flores-Mangas. 2006. MPTrain: a mobile, music and physiology-based personal trainer. In *MobileHCI'06.* ACM, 21–28.
- Sheina Orbell and Bas Verplanken. 2010. The automatic component of habit in health behavior: habit as cue-contingent automaticity. *Health Psychology* 29, 4 (July 2010), 374–83. https://doi.org/10.1037/a0019596
- Sheina Orbell and Bas Verplanken. 2015. The strength of habit. Health psychology review 9, 3 (2015), 311-317.

- World Health Organization. 2017. Noncommunicable Diseases Progress Monitor 2017. (2017). ISBN 978-92-4-151302-9 Licence: CC BY-NC-SA 3.0 IGO.
- R. Orji and K. Moffatt. 2016. Persuasive technology for health and wellness: State-of-the-art and emerging trends. *Health Informatics Journal* (May 2016). https://doi.org/10.1177/1460458216650979
- Rita Orji, Lennart E. Nacke, and Chrysanne Di Marco. 2017. Towards Personality-driven Persuasive Health Games and Gamified Systems. In *CHI '17*. ACM Press, 1015–1027. https://doi.org/10.1145/3025453.3025577
- Antti Oulasvirta, Tye Rattenbury, Lingyi Ma, and Eeva Raita. 2012. Habits make smartphone use more pervasive. *Personal and Ubiquitous Computing* 16, 1 (2012), 105–114. https://doi.org/10.1007/s00779-011-0412-2
- Jeni Paay, Jesper Kjeldskov, Mikael B. Skov, Lars Lichon, and Stephan Rasmussen. 2015. Understanding Individual Differences for Tailored Smoking Cessation Apps. In Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems (CHI '15). ACM, New York, NY, USA, 1699–1708. https://doi.org/10.1145/2702123.2702321
- Carly R. Pacanowski, Jeffery Sobal, David A. Levitsky, Nancy E. Sherwood, Chelsey L. Keeler, April M. Miller, Ashley R. Acosta, Natalie Hansen, Peter L. Wang, Sarah R. Guilbert, Arianne L. Paroly, Michael Commesso, and Francoise M. Vermeylen. 2015. Does Measuring Body Weight Impact Subsequent Response to Eating Behavior Questions? *Journal of the American College of Nutrition* 34, 3 (May 2015), 199–204. https://doi.org/10.1080/07315724.2014.931263
- Tibor P. Palfai, Peter M. Monti, Suzanne M. Colby, and Damaris J. Rohsenow. 1997. Effects of suppressing the urge to drink on the accessibility of alcohol outcome expectancies. *Behaviour Research and Therapy* 35, 1 (Jan. 1997), 59–65. https://doi.org/10.1016/S0005-7967(96)00079-4
- Esther K. Papies. 2016. Health goal priming as a situated intervention tool: How to benefit from nonconscious motivational routes to health behavior. *Health psychology review* just-accepted (2016), 1–35.
- Hancheol Park and Gahgene Gweon. 2015. Initiating Moderation in Problematic Smartphone Usage Patterns. In *CHI'15 Extended Abstracts*. ACM, New York, NY, USA, 1585–1590. https://doi.org/10.1145/2702613.2732886
- Pavlok. 2018. Pavlok homepage. (2018). http://pavlok.com/
- Veljko Pejovic and Mirco Musolesi. 2014. InterruptMe: designing intelligent prompting mechanisms for pervasive applications. In UbiComp '14. ACM, 897–908. https://doi.org/10.1145/2632048.2632062
- Veljko Pejovic and Mirco Musolesi. 2015. Anticipatory mobile computing: A survey of the state of the art and research challenges. ACM Computing Surveys (CSUR) 47, 3 (2015), 47.
- Trevor Pels, Christina Kao, and Saguna Goel. 2014. FatBelt: motivating behavior change through isomorphic feedback. In *UIST'14 Adjunct*. ACM, 123–124.
- Mathias Pessiglione, Liane Schmidt, Bogdan Draganski, Raffael Kalisch, Hakwan Lau, Ray J Dolan, and Chris D Frith. 2007. How the brain translates money into force: a neuroimaging study of subliminal motivation. *Science* 316, 5826 (2007), 904–906. https://doi.org/10.1126/science.1140459
- Gjalt-Jorn Ygram Peters, Marijn Bruin, and Rik Crutzen. 2015. Everything should be as simple as possible, but no simpler: towards a protocol for accumulating evidence regarding the active content of health behaviour change interventions. *Health Psychology Review* 9, 1 (2015), 1–14.
- Chanda Phelan, Cliff Lampe, and Paul Resnick. 2016. It's Creepy, But it Doesn't Bother Me. ACM Press, 5240–5251. https://doi.org/10.1145/2858036.2858381
- L. Alison Phillips and Benjamin Gardner. 2016. Habitual exercise instigation (vs. execution) predicts healthy adultsâĂŹ exercise frequency. *Health Psychology* 35, 1 (2016), 69.
- Charlie Pinder. 2017. The Anti-Influence Engine: Escaping the Diabolical Machine of Pervasive Advertising. In *CHI'17 EA*. ACM Press, Denver, USA. https://doi.org/10.1145/3027063.3052762
- Charlie Pinder, Russell Beale, and Robert J. Hendley. 2016a. Zephyr: exploring digital behaviour change interventions to treat hoarding. In *MobileHCI '16 Adjunct*. ACM Press, 875–882. https://doi.org/10.1145/2957265.2961838
- Charlie Pinder, Rowanne Fleck, Rosa Lillia Segundo Diaz, Russell Beale, and Robert J Hendley. 2016b. Accept the Banana: Exploring Incidental Cognitive Bias Modification Techniques on Smartphones. In *CHI'16 EA (CHI EA '16)*. ACM Press, San Jose, CA, USA, 2923–2931. https://doi.org/10.1145/2851581.2892453
- Charlie Pinder, Jo Vermeulen, Russell Beale, and Robert Hendley. 2015. Subliminal Priming of Nonconscious Goals on Smartphones. In *MobileHCl'15 Adjunct*. ACM, New York, NY, USA, 825–830. https://doi.org/10.1145/2786567.2793707
- Charlie Pinder, Jo Vermeulen, Benjamin R. Cowan, Russell Beale, and Robert J. Hendley. 2017. Exploring the Feasibility of Subliminal Priming on Smartphones. In *MobileHCI'17*. ACM, Vienna, Austria.
- Charlie Pinder, Jo Vermeulen, Adhi Wicaksono, Russell Beale, and Robert J. Hendley. 2016c. If this, then habit: exploring context-aware implementation intentions on smartphones. In *MobileHCI '16 Adjunct*. ACM Press, 690–697. https://doi.org/10.1145/2957265.2961837
- Michael Prendergast, Deborah Podus, John Finney, Lisa Greenwell, and John Roll. 2006. Contingency management for treatment of substance use disorders: a meta-analysis. *Addiction* 101, 11 (2006), 1546–1560.
- Justin Presseau, Marie Johnston, Tarja Heponiemi, Marko Elovainio, Jill J. Francis, Martin P. Eccles, Nick Steen, Susan Hrisos, Elaine Stamp, Jeremy M. Grimshaw, Gillian Hawthorne, and Falko F. Sniehotta. 2014. Reflective and Automatic

Processes in Health Care Professional Behaviour: a Dual Process Model Tested Across Multiple Behaviours. *Annals of Behavioral Medicine* 48, 3 (Dec. 2014), 347–358. https://doi.org/10.1007/s12160-014-9609-8

- Andrew Prestwich, Marco Perugini, and Robert Hurling. 2010. Can implementation intentions and text messages promote brisk walking? A randomized trial. *Health Psychology* 29, 1 (2010), 40. https://doi.org/10.1037/a0016993
- James O Prochaska and Carlo C DiClemente. 1983. Stages and processes of self-change of smoking: toward an integrative model of change. *Journal of consulting and clinical psychology* 51, 3 (1983), 390.
- James O. Prochaska and Wayne V. Velicer. 1997. The transtheoretical model of health behavior change. American Journal of Health Promotion 12, 1 (1997), 38–48.
- Mashfiqui Rabbi, Min Hane Aung, Mi Zhang, and Tanzeem Choudhury. 2015. MyBehavior: Automatic Personalized Health Feedback from User Behaviors and Preferences Using Smartphones. In *UbiComp âĂŹ15*. ACM, New York, NY, USA, 707–718. https://doi.org/10.1145/2750858.2805840
- Sharon Rabinovitz and Maayan Nagar. 2015. Possible End to an Endless Quest? Cognitive Bias Modification for Excessive Multiplayer Online Gamers. 18, 10 (2015), 581–587. https://doi.org/10.1089/cyber.2015.0173
- Kiran K Rachuri, Mirco Musolesi, Cecilia Mascolo, Peter J Rentfrow, Chris Longworth, and Andrius Aucinas. 2010. Emotion-Sense: a mobile phones based adaptive platform for experimental social psychology research. In *UbiComp '10 (Ubicomp '10)*. ACM, New York, NY, USA, 281–290. https://doi.org/10.1145/1864349.1864393
- Anja Meiland Ranfelt, Tony Wigram, and Peter m. 2009. Towards a handy interactive persuasive diary for teenagers with a diagnosis of autism. In *Proceedings of the 4th International Conference on Persuasive Technology*. ACM, New York, NY, USA, 3:1–3:9. http://doi.acm.org/10.1145/1541948.1541953
- Amanda L. Rebar, James A. Dimmock, Ben Jackson, Ryan E. Rhodes, Andrew Kates, Jade Starling, and Corneel Vandelanotte. 2016. A Systematic Review of the Effects of Non-Conscious Regulatory Processes in Physical Activity. *Health psychology review* just-accepted (2016), 1–86.
- David Redish, Steve Jensen, Adam Johnson, and Zeb Kurth-Nelson. 2007. Reconciling reinforcement learning models with behavioral extinction and renewal: implications for addiction, relapse, and problem gambling. *Psychological review* 114, 3 (July 2007), 784–805. https://doi.org/10.1037/0033-295X.114.3.784
- Byron Reeves and Clifford Nass. 1996. The media equation: How people respond to computers, television, and new media like real people and places. Cambridge University Press.
- Wolfgang Reitberger, Alexander Meschtscherjakov, Thomas Mirlacher, Thomas Scherndl, Hermann Huber, and Manfred Tscheligi. 2009. A persuasive interactive mannequin for shop windows. In *Persuasive '09*. ACM, 4:1–4:8. http://doi.acm. org/10.1145/1541948.1541954
- Ian Renfree, Daniel Harrison, Paul Marshall, Katarzyna Stawarz, and Anna Cox. 2016. Don't Kick the Habit: The Role of Dependency in Habit Formation Apps. In CHI'16 Extended Abstracts. ACM, 2932–2939. https://doi.org/10.1145/2851581. 2892495
- Denise T. D. Ridder, Gerty Lensvelt-Mulders, Catrin Finkenauer, F. Marijn Stok, and Roy F. Baumeister. 2012. Taking Stock of Self-Control: A Meta-Analysis of How Trait Self-Control Relates to a Wide Range of Behaviors. *Personality and Social Psychology Review* 16, 1 (Feb. 2012), 76–99. https://doi.org/10.1177/1088868311418749
- William Riley, Daniel Rivera, Audie Atienza, Wendy Nilsen, Susannah Allison, and Robin Mermelstein. 2011. Health behavior models in the age of mobile interventions: are our theories up to the task? *Translational Behavioral Medicine* 1, 1 (2011), 53–71. https://doi.org/10.1007/s13142-011-0021-7
- Barbara K. Rimer and Karen Glanz. 2005. Theory at a glance: a guide for health promotion practice. (2005).
- Eric Robinson, Paul Aveyard, Amanda Daley, Kate Jolly, Amanda Lewis, Deborah Lycett, and Suzanne Higgs. 2013. Eating attentively: a systematic review and meta-analysis of the effect of food intake memory and awareness on eating. *The American journal of clinical nutrition* (2013), ajcn. 045245.
- Yvonne Rogers. 2004. New theoretical approaches for HCI. Annual review of information science and technology 38, 1 (2004), 87–143.
- Yvonne Rogers. 2006. Moving on from WeiseràĂŹs vision of calm computing: Engaging ubicomp experiences. In UbiComp 2006. Springer, 404–421.
- Yvonne Rogers, William R. Hazlewood, Paul Marshall, Nick Dalton, and Susanna Hertrich. 2010. Ambient influence: Can twinkly lights lure and abstract representations trigger behavioral change?. In *UbiComp* '10. ACM, 261–270. https: //doi.org/10.1145/1864349.1864372
- Alexander J Rothman, Paschal Sheeran, and Wendy Wood. 2009. Reflective and automatic processes in the initiation and maintenance of dietary change. Annals of behavioral medicine : a publication of the Society of Behavioral Medicine 38 Suppl 1 (Dec. 2009), S4–17. https://doi.org/10.1007/s12160-009-9118-3
- Maaike Roubroeks, Jaap Ham, and Cees Midden. 2011. When artificial social agents try to persuade people: The role of social agency on the occurrence of psychological reactance. *International Journal of Social Robotics* 3, 2 (2011), 155–165. https://doi.org/10.1007/s12369-010-0088-1

- Peter AM Ruijten, Cees JH Midden, and Jaap Ham. 2011. Unconscious persuasion needs goal-striving: the effect of goal activation on the persuasive power of subliminal feedback. In *Persuasive '11*. http://dl.acm.org/citation.cfm?id=2467807
- Hanna Schneider, Kilian Moser, Andreas Butz, and Florian Alt. 2016. Understanding the Mechanics of Persuasive System Design: A Mixed-Method Theory-driven Analysis of Freeletics. ACM Press, 309–320. https://doi.org/10.1145/2858036. 2858290
- Stephanie Schoeppe, Stephanie Alley, Wendy Van Lippevelde, Nicola A. Bray, Susan L. Williams, Mitch J. Duncan, and Corneel Vandelanotte. 2016. Efficacy of interventions that use apps to improve diet, physical activity and sedentary behaviour: a systematic review. International Journal of Behavioral Nutrition and Physical Activity 13, 1 (Dec. 2016). https://doi.org/10.1186/s12966-016-0454-y
- Philipp M. Scholl, Nagihan KÃijcÃijkyildiz, and Kristof Van Laerhoven. 2013. When Do You Light a Fire?: Capturing Tobacco Use with Situated, Wearable Sensors. In *UbiComp '15 Adjunct*. ACM, New York, NY, USA, 1295–1304. https: //doi.org/10.1145/2494091.2497284
- Katarina Segerståhl, Tanja Kotro, and Kaisa Väänänen-Vainio-Mattila. 2010. Pitfalls in Persuasion: How Do Users Experience Persuasive Techniques in a Web Service?. In *Persuasive '10*. Copenhagen, Denmark, 211–222.
- Abigail J. Sellen and Steve Whittaker. 2010. Beyond total capture: a constructive critique of lifelogging. *Commun. ACM* 53, 5 (May 2010), 70. https://doi.org/10.1145/1735223.1735243
- Idit Shalev and John A. Bargh. 2011. Use of Priming-Based Interventions to Facilitate Psychological Health Commentary on Kazdin and Blase (2011). *Perspectives on Psychological Science* 6, 5 (2011), 488–492.
- Paschal Sheeran. 2002. IntentionâĂŤBehavior Relations: A Conceptual and Empirical Review. European Review of Social Psychology 12, 1 (Jan. 2002), 1–36. https://doi.org/10.1080/14792772143000003
- Paschal Sheeran, Gaston Godin, Mark Conner, and Marc Germain. 2017a. Paradoxical Effects of Experience: Past Behavior Both Strengthens and Weakens the Intention-Behavior Relationship. *Journal of the Association for Consumer Research* 2, 3 (April 2017), 309–318. https://doi.org/10.1086/691216
- Paschal Sheeran, Peter M. Gollwitzer, and John A. Bargh. 2013. Nonconscious processes and health. *Health Psychology* 32, 5 (2013), 460.
- Paschal Sheeran, William MP Klein, and Alexander J. Rothman. 2017b. Health behavior change: Moving from observation to intervention. *Annual Review of Psychology* 68 (2017), 573–600.
- Burrhus Frederic Skinner. 1938. The behavior of organisms: An experimental analysis. (1938).
- Ruth M Sladek, Paddy A Phillips, and Malcolm J Bond. 2006. Implementation science: a role for parallel dual processing models of reasoning? *Implementation Science* 1, 1 (Dec 2006). https://doi.org/10.1186/1748-5908-1-12
- Aaron Sloman. 2002. How many separately evolved emotional beasties live within us. *Emotions in humans and artifacts* (2002), 35–114.
- Falko F. Sniehotta. 2009. Towards a theory of intentional behaviour change: Plans, planning, and self-regulation. 14, 2 (2009), 261–273.
- Falko F. Sniehotta and Justin Presseau. 2012. The Habitual Use of the Self-report Habit Index. *Annals of Behavioral Medicine* 43, 1 (Feb. 2012), 139–140. https://doi.org/10.1007/s12160-011-9305-x
- Falko F. Sniehotta, Justin Presseau, and Vera AraÞjo-Soares. 2014. Time to retire the theory of planned behaviour. *Health Psychology Review* 8, 1 (Jan. 2014), 1–7. https://doi.org/10.1080/17437199.2013.869710
- Mark Snyder. 1974. Self-monitoring of expressive behavior. Journal of personality and social psychology 30, 4 (1974), 526.
- Tobias Sonne, Jörg Müller, Paul Marshall, Carsten Obel, and Kaj Grønbæk. 2016. Changing Family Practices with Assistive Technology: MOBERO Improves Morning and Bedtime Routines for Children with ADHD. ACM Press, 152–164. https: //doi.org/10.1145/2858036.2858157
- Caleb Southern, Yunnuo Cheng, Cheng Zhang, and Gregory D. Abowd. 2017. Understanding the Cost of Driving Trips. In *CHI'17.* ACM Press, 430–434. https://doi.org/10.1145/3025453.3025686
- Charles Spence and Maya Shankar. 2010. The influence of auditory cues on the perception of, and responses to, food and drink. *Journal of Sensory Studies* 25, 3 (2010), 406–430.
- Spire. 2018. Spire homepage. (2018). https://www.spire.io/
- Alan W. Stacy and Reinout W. Wiers. 2010. Implicit Cognition and Addiction: A Tool for Explaining Paradoxical Behavior. Annual Review of Clinical Psychology 6, 1 (March 2010), 551–575. https://doi.org/10.1146/annurev.clinpsy.121208.131444
- J. E. R. Staddon and D. T. Cerutti. 2003. Operant Conditioning. Annual Review of Psychology 54, 1 (Feb. 2003), 115–144. https://doi.org/10.1146/annurev.psych.54.101601.145124
- Alexander D Stajkovic, Edwin Locke, and Eden S Blair. 2006. A first examination of the relationships between primed subconscious goals, assigned conscious goals, and task performance. *The Journal of applied psychology* 91, 5 (Sept. 2006), 1172–80. https://doi.org/10.1037/0021-9010.91.5.1172
- Aline E. Stämpfli and Thomas A. Brunner. 2016. The art of dieting: Exposure to thin sculptures effortlessly reduces the intake of unhealthy food in motivated eaters. *Food Quality and Preference* 50 (2016), 88–93.
- Keith Stanovich. 2011. Rationality and the reflective mind. Oxford University Press.

Keith E. Stanovich. 2005. The robot's rebellion: Finding meaning in the age of Darwin. University of Chicago Press.

- Katarzyna Stawarz, Anna L. Cox, and Ann Blandford. 2015. Beyond Self-Tracking and Reminders: Designing Smartphone Apps That Support Habit Formation. In *CHI* '15. 2653–2662. https://doi.org/10.1145/2702123.2702230
- Fritz Strack and Roland Deutsch. 2004. Reflective and impulsive determinants of social behavior. Personality and social psychology review : an official journal of the Society for Personality and Social Psychology, Inc 8, 3 (Jan. 2004), 220–47. https://doi.org/10.1207/s15327957pspr0803_1
- Fritz Strack and Roland Deutsch. 2014. The reflective-impulsive model. In *Dual-process theories of the social mind*, Jeffrey W. Sherman, B. Gawronski, and Y. Trope (Eds.). The Guilford Press, New York, NY, USA, 92–104.
- Erin J. Strahan, Steven J. Spencer, and Mark P. Zanna. 2002. Subliminal priming and persuasion: Striking while the iron is hot. *Journal of Experimental Social Psychology* 38, 6 (2002), 556–568. https://doi.org/10.1016/s0022-1031(02)00502-4
- Valerie Sugarman and Edward Lank. 2015. Designing Persuasive Technology to Manage Peak Electricity Demand in Ontario Homes. In Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems. ACM, 1975–1984.
- Minhyang Mia Suh and Gary Hsieh. 2016. Designing for Future Behaviors: Understanding the Effect of Temporal Distance on Planned Behaviors. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems. ACM, 1084–1096.
- Ron Sun, Paul Slusarz, and Chris Terry. 2005. The Interaction of the Explicit and the Implicit in Skill Learning: A Dual-Process Approach. Psychological Review 112, 1 (2005), 159–192. https://doi.org/10.1037/0033-295X.112.1.159
- Ana Tajadura-Jiménez, Maria Basia, Ophelia Deroy, Merle Fairhurst, Nicolai Marquardt, and Nadia Bianchi-Berthouze. 2015. As Light As Your Footsteps: Altering Walking Sounds to Change Perceived Body Weight, Emotional State and Gait. In *CHI* '15. ACM, Seoul, South Korea, 2943–2952. https://doi.org/10.1145/2702123.2702374
- David Taylor, Michael Bury, Natasha Campling, Sarah Carter, Sara Garfied, Jenny Newbould, and Tim Rennie. 2006. A Review of the use of the Health Belief Model (HBM), the Theory of Reasoned Action (TRA), the Theory of Planned Behaviour (TPB) and the Trans-Theoretical Model (TTM) to study and predict health related behaviour change. Technical Report. National Institute for Health and Clinical Excellence, London, UK. https://www.nice.org.uk/guidance/ph6/resources/ behaviour-change-taylor-et-al-models-review2
- Natalie Taylor, Mark Conner, and Rebecca Lawton. 2012. The impact of theory on the effectiveness of worksite physical activity interventions: a meta-analysis and meta-regression. *Health Psychology Review* 6, 1 (2012), 33–73.
- Sara B Taylor, Candace R Lewis, and M Foster Olive. 2013. The neurocircuitry of illicit psychostimulant addiction: acute and chronic effects in humans. *Substance Abuse and Rehabilitation* (Feb. 2013), 29. https://doi.org/10.2147/SAR.S39684
- Richard H. Thaler and Cass R. Sunstein. 2008. Nudge: Improving decisions about health, wealth, and happiness. Yale Univ Pr.
- Dianne M. Tice, Ellen Bratslavsky, and Roy F. Baumeister. 2001. Emotional distress regulation takes precedence over impulse control: If you feel bad, do it! *Journal of Personality and Social Psychology* 80, 1 (2001), 53–67. https://doi.org/10.1037/ /0022-3514.80.1.53
- Robert Tobias. 2009. Changing behavior by memory aids: a social psychological model of prospective memory and habit development tested with dynamic field data. *Psychological review* 116, 2 (2009), 408.
- Harry Charalambos Triandis. 1977. Interpersonal behavior. Brooks/Cole Publishing Company Monterey, CA.
- Blase Ur, Elyse McManus, Melwyn Pak Yong Ho, and Michael L. Littman. 2014. Practical trigger-action programming in the smart home. In CHI '14. ACM, 803–812. https://doi.org/10.1145/2556288.2557420
- Alexander JAM van Deursen, Colin L. Bolle, Sabrina M. Hegner, and Piet AM Kommers. 2015. Modeling habitual and addictive smartphone behavior: The role of smartphone usage types, emotional intelligence, social stress, self-regulation, age, and gender. *Computers in human behavior* 45 (2015), 411–420.
- Harm Veling, Guido M Koningsbruggen, Henk Aarts, and Wolfgang Stroebe. 2014. Targeting impulsive processes of eating behavior via the internet. Effects on body weight. *Appetite* 78 (July 2014), 102–109. https://doi.org/10.1016/j.appet.2014. 03.014
- Peter-Paul Verbeek. 2009. Ambient intelligence and persuasive technology: the blurring boundaries between human and technology. *NanoEthics* 3, 3 (2009), 231–242.
- Frederick Verbruggen and Gordon D Logan. 2009. Automatic and Controlled Response Inhibition: Associative Learning in the Go/No-Go and Stop-Signal Paradigms. *Journal of Experimental Psychology* 137, 4 (2009), 649–672. https: //doi.org/10.1037/a0013170
- Bas Verplanken and Sheina Orbell. 2003. Reflections on Past Behavior: A Self-Report Index of Habit Strength. *Journal of Applied Social Psychology* 33 (2003), 1313–1330. https://doi.org/10.1111/j.1559-1816.2003.tb01951.x
- Bas Verplanken and Wendy Wood. 2006. Interventions to Break and Create Consumer Interventions Habits. *Journal of Public Policy* 25, 1 (2006), 90–103.
- Ricardo Mark Villamarín-Salomón and José Carlos Brustoloni. 2010. Using Reinforcement to Strengthen Users' Secure Behaviors. In *CHI '10*. ACM, New York, NY, USA, 363–372. https://doi.org/10.1145/1753326.1753382
- Kevin G. Volpp, Leslie K. John, Andrea B. Troxel, Laurie Norton, Jennifer Fassbender, and George Loewenstein. 2008. Financial incentive-based approaches for weight loss: a randomized trial. Jama 300, 22 (2008), 2631–2637.

- Jane C Walsh, Teresa Corbett, Michael Hogan, Jim Duggan, and Abra McNamara. 2016. An mHealth Intervention Using a Smartphone App to Increase Walking Behavior in Young Adults: A Pilot Study. *JMIR mHealth and uHealth* 4, 3 (Sept. 2016), e109. https://doi.org/10.2196/mhealth.5227
- Yang Wang, Pedro Giovanni Leon, Alessandro Acquisti, Lorrie Faith Cranor, Alain Forget, and Norman Sadeh. 2014. A Field Trial of Privacy Nudges for Facebook. In CHI'14. ACM, New York, NY, USA, 2367–2376. https://doi.org/10.1145/2556288. 2557413
- Brian Wansink. 2010. From mindless eating to mindlessly eating better. *Physiology & behavior* 100, 5 (July 2010), 454–63. https://doi.org/10.1016/j.physbeh.2010.05.003
- Brian Wansink and Pierre Chandon. 2014. Slim by design: Redirecting the accidental drivers of mindless overeating. *Journal of Consumer Psychology* 24, 3 (2014), 413–431.
- Way of Life. 2018. Way of Life The Ultimate Habit Building App. (2018). http://wayoflifeapp.com/
- Thomas L. Webb, Judith Joseph, Lucy Yardley, and Susan Michie. 2010a. Using the Internet to Promote Health Behavior Change: A Systematic Review and Meta-analysis of the Impact of Theoretical Basis, Use of Behavior Change Techniques, and Mode of Delivery on Efficacy. *Journal of Internet Medical Research* 12, 1 (2010). https://doi.org/doi:10.2196/jmir.1376
- Thomas L Webb and Paschal Sheeran. 2006. Does changing behavioral intentions engender behavior change? A metaanalysis of the experimental evidence. *Psychological bulletin* 132, 2 (March 2006), 249–68. https://doi.org/10.1037/ 0033-2909.132.2.249
- Thomas L. Webb and Paschal Sheeran. 2008. Mechanisms of implementation intention effects: the role of goal intentions, self-efficacy, and accessibility of plan components. *British Journal of Social Psychology* 47, 3 (2008), 373–395. https://doi.org/10.1348/014466607x267010
- Thomas L. Webb, Falko F Sniehotta, and Susan Michie. 2010b. Using theories of behaviour change to inform interventions for addictive behaviours. *Addiction (Abingdon, England)* 105, 11 (Nov. 2010), 1879–92. https://doi.org/10.1111/j.1360-0443. 2010.03028.x
- Daniel M Wegner. 1994. Ironic processes of mental control. Psychological review 101, 1 (1994), 34.
- Robert West. 2005. Time for a change: putting the Transtheoretical (Stages of Change) Model to rest. *Addiction* 100, 8 (Aug. 2005), 1036–9.
- Robert West. 2006. Theory of addiction. Blackwell Publishing Ltd, Oxford.
- Isaac Wiafe and Keiichi Nakata. 2012. Bibliographic Analysis of Persuasive Systems: Techniques, Methods and Domains of Application. In *Persuasive Technology*. 61–64. http://www.ep.liu.se/ecp/068/ecp12068.pdf#page=65
- Reinout W Wiers, Carolin Eberl, Mike Rinck, Eni S Becker, and Johannes Lindenmeyer. 2011. Retraining automatic action tendencies changes alcoholic patients' approach bias for alcohol and improves treatment outcome. *Psychological Science* 22, 4 (April 2011), 490–7. https://doi.org/10.1177/0956797611400615
- Reinout W. Wiers, Thomas E. Gladwin, Wilhelm Hofmann, Elske Salemink, and K. Richard Ridderinkhof. 2013. Cognitive Bias Modification and Cognitive Control Training in Addiction and Related Psychopathology: Mechanisms, Clinical Perspectives, and Ways Forward. *Clinical Psychological Science* 1, 2 (Jan. 2013), 192–212. https://doi.org/10.1177/ 2167702612466547
- J M Williams, A Mathews, and C MacLeod. 1996. The emotional Stroop task and psychopathology. *Psychological Bulletin* 120, 1 (July 1996), 3–24. https://doi.org/10.1037/0033-2909.120.1.3
- Rena R. Wing, Deborah F. Tate, Amy A. Gorin, Hollie A. Raynor, and Joseph L. Fava. 2006. A Self-Regulation Program for Maintenance of Weight Loss. New England Journal of Medicine 355, 15 (Oct. 2006), 1563–1571. https://doi.org/10.1056/ NEJMoa061883
- Sanne Wit and Anthony Dickinson. 2009. Associative theories of goal-directed behaviour: a case for animal-human translational models. *Psychological Research Psychologische Forschung* 73, 4 (July 2009), 463–476. https://doi.org/10.1007/s00426-009-0230-6
- Charlotte E. Wittekind, Ansgar Feist, Brooke C. Schneider, Steffen Moritz, and Anja Fritzsche. 2015. The approach-avoidance task as an online intervention in cigarette smoking: A pilot study. *Journal of behavior therapy and experimental psychiatry* 46 (2015), 115–120.
- Donghee Wohn, Alcides Velasquez, Tor Bjornrud, and Cliff Lampe. 2012. Habit as an explanation of participation in an online peer-production community. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*. ACM, 2905–2914.
- Wendy Wood, Jennifer S. Labrecque, Pei-Ying Lin, and Dennis Rünger. 2014. Habits in Dual Process Models. In *Dual-process theories of the social mind*, Jeffrey Sherman, Bertram Gawronski, and Yaacov Trope (Eds.). 371.
- Wendy Wood and David T Neal. 2007. A new look at habits and the habit-goal interface. *Psychological review* 114, 4 (Oct. 2007), 843–63. https://doi.org/10.1037/0033-295X.114.4.843
- Wendy Wood, Jeffrey M. Quinn, and Deborah A. Kashy. 2002. Habits in everyday life: thought, emotion, and action. Journal of personality and social psychology 83, 6 (2002), 1281.

- Wendy Wood and Dennis Rünger. 2016. Psychology of Habit. Annual Review of Psychology 67, 1 (Jan. 2016), 289–314. https://doi.org/10.1146/annurev-psych-122414-033417
- Álinson Santos Xavier. 2018. Loop Habit Tracker, an Android app for creating and maintaining good habits. (2018). https://github.com/iSoron/uhabits/
- Rayoung Yang, Eunice Shin, Mark W. Newman, and Mark S. Ackerman. 2015. When Fitness Trackers DonâĂŹt âĂŸFitâĂŹ: End -User Difficulties in the Assessment of Personal Tracking Device Accuracy. In UbiComp âĂŹ15. https://doi.org/10. 1145/2750858.2804269
- Henry H. Yin and Barbara J. Knowlton. 2006. The role of the basal ganglia in habit formation. *Nature Reviews Neuroscience* 7, 6 (2006), 464–476.
- Claire M. Zedelius, Harm Veling, and Henk Aarts. 2011. Boosting or choking ? How conscious and unconscious reward processing modulate the active maintenance of goal-relevant information. *Consciousness and Cognition* 20, 2 (June 2011), 355–362. https://doi.org/10.1016/j.concog.2010.05.001
- Mohamed Ali Ziai and John C. Batchelor. 2011. Temporary on-skin passive UHF RFID transfer tag. Antennas and Propagation, IEEE Transactions on 59, 10 (2011), 3565–3571.
- zombiesrungame. 2018. Zombies, Run! (2018). https://www.zombiesrungame.com
- Oren Zuckerman and Ayelet Gal-Oz. 2014. Deconstructing gamification: evaluating the effectiveness of continuous measurement, virtual rewards, and social comparison for promoting physical activity. *Personal and Ubiquitous Computing* 18, 7 (2014), 1705–1719.